

# Role of Impact Assessment in Sustainable Development

Edited by Paolo Biancone and Silvana Secinaro Printed Edition of the Special Issue Published in *Sustainability* 



www.mdpi.com/journal/sustainability

# **Role of Impact Assessment in Sustainable Development**

# **Role of Impact Assessment in Sustainable Development**

Editors

Paolo Biancone Silvana Secinaro

MDPI • Basel • Beijing • Wuhan • Barcelona • Belgrade • Manchester • Tokyo • Cluj • Tianjin



*Editors* Paolo Biancone Management University of Turin Turin Italy

Silvana Secinaro Management University of Turin Turin Italy

*Editorial Office* MDPI St. Alban-Anlage 66 4052 Basel, Switzerland

This is a reprint of articles from the Special Issue published online in the open access journal *Sustainability* (ISSN 2071-1050) (available at: www.mdpi.com/journal/sustainability/special\_issues/ role\_impact\_assessment).

For citation purposes, cite each article independently as indicated on the article page online and as indicated below:

LastName, A.A.; LastName, B.B.; LastName, C.C. Article Title. *Journal Name* Year, *Volume Number*, Page Range.

ISBN 978-3-0365-3526-5 (Hbk) ISBN 978-3-0365-3525-8 (PDF)

© 2022 by the authors. Articles in this book are Open Access and distributed under the Creative Commons Attribution (CC BY) license, which allows users to download, copy and build upon published articles, as long as the author and publisher are properly credited, which ensures maximum dissemination and a wider impact of our publications.

The book as a whole is distributed by MDPI under the terms and conditions of the Creative Commons license CC BY-NC-ND.

### Contents

About the Editors
Preface to "Role of Impact Assessment in Sustainable Development"
<b>Luigi Corvo, Lavinia Pastore, Arianna Manti and Daniel Iannaci</b> Mapping Social Impact Assessment Models: A Literature Overview for a Future Research Agenda
Reprinted from: <i>Sustainability</i> <b>2021</b> , <i>13</i> , 4750, doi:10.3390/su13094750
Adam KozieńThe Principle of Sustainable Development as the Basis for Weighing the Public Interest andIndividual Interest in the Scope of the Cultural Heritage Protection Law in the European UnionReprinted from: Sustainability 2021, 13, 3985, doi:10.3390/su1307398517
Laura Pellegrini, Mirko Locatelli, Silvia Meschini, Giulia Pattini, Elena Seghezzi and Lavinia Chiara Tagliabue et al
Information Modelling Management and Green Public Procurement for Waste Management and Environmental Renovation of Brownfields Reprinted from: Sustainability 2021, 13, 8585, doi:10.3390/su13158585
Peter Ho, Bin Md Saman Nor-Hisham and Heng Zhao   Limits of the Environmental Impact Assessment (EIA) in Malaysia: Dam Politics, Rent-Seeking, and Conflict   Reprinted from: Sustainability 2020, 12, 10467, doi:10.3390/su122410467
<b>Paolo Esposito, Valerio Brescia, Chiara Fantauzzi and Rocco Frondizi</b> Understanding Social Impact and Value Creation in Hybrid Organizations: The Case of Italian Civil Service
Reprinted from: <i>Sustainability</i> <b>2021</b> , <i>13</i> , 4058, doi:10.3390/su13074058
Maurizio Cisi and Francesca Alice Centrone The Human Capital for Value Creation and Social Impact: The Interpretation of the IR's HC Definition
Reprinted from: <i>Sustainability</i> <b>2021</b> , <i>13</i> , 6989, doi:10.3390/su13136989
<b>Silvana Secinaro, Davide Calandra, Denisa Petricean and Federico Chmet</b> Social Finance and Banking Research as a Driver for Sustainable Development: A Bibliometric Analysis
Reprinted from: <i>Sustainability</i> <b>2020</b> , <i>13</i> , 330, doi:10.3390/su13010330 <b>129</b>

### About the Editors

#### **Paolo Biancone**

Paolo Biancone is a Full Professor in Business Administration at the University of Turin. He is a chartered accountant, a freelance journalist, and the author of numerous publications on accounting, entrepreneurship, and Islamic finance. He is Editor-in-Chief of the *European Journal of Islamic Finance* and the *European Journal of Social Impact and Circular Economy*. He is the scientific coordinator of the University of Turin team of the European project Avangard on electric cars. He has promoted research in new technologies, business models, and the Sustainable Development Goals.

#### Silvana Secinaro

Silvana Secinaro is an Associate Professor at the Department of Management at the University of Turin, chartered accountant, a freelance journalist, and the author of numerous publications on public and private accounting. She has a Degree in Accounting from Bocconi University and a PhD in Business and Management at the University of Turin. A Component of National Commission of Italian Ministry of Finances on IPSAS and Member of the editorial staff of *Press*, the national magazine of the Chartered Experts Accounting, October 2014. She has participated in review projects funded by the European Community on Financial Reporting with a three-year mandate 2018–2020. She is part of the Avangard European research group on electric cars.

# Preface to "Role of Impact Assessment in Sustainable Development"

At a time of extreme difficulty for companies and society, a wide range of studies and guidelines invoke impact assessments as a driver for more sustainable development and effective resource planning.

The issue of sustainable development is determined not only by the effective use of resources and the application of appropriate strategies, but also by considering social aspects. Thus, impact assessments have been established in companies considering multiple criteria of different natures (i.e., economic, environmental, and social) and the transparency and engagement of various stakeholders, such as organizations, government, and communities.

Several studies have highlighted the role of environmental and strategic impact assessments (EIAs and SEAs, respectively) as facilitators of the evidence of climate change impacts within organizations. In the same way, social impact assessments (SIAs) are a means of resilience in responding to changes in the needs of external stakeholders seeking to replace the contributions of public authorities, and therefore identifying opportunities that enhance benefits for local communities. The literature focuses extensively on both lines of research; however, few studies have focused on how hybrid organizations, for example, can be involved in these processes.

Social impacts and environmental impacts have reflections on economic aspects. They are closely related to the complex reality in which hybrid organizations provide their services by focusing on both social impact and profit aspects by involving different sectors (i.e., public/private/third sectors in terms of governmental/profit/non-profit). Involvement of all of these factors could play a key role in promoting sustainable development through specific social, institutional, and technological innovations and practices.

This book discusses the role of environmental and social impact assessments in companies, such as profit and non-profit realities as a pathway for sustainable development and the role of hybrid organizations in the context of growth and change. The book includes theoretical analyses, case studies, and literature reviews on impact assessment using broad perspectives. Finally, the links between social assessment and social finance are addressed.

We like to imagine that this book might be helpful primarily to managers and representatives of the third sector. In addition, it will be useful for researchers to explore and continue analysis in this area of research.

The editors are grateful to all the authors who contributed to the volume, and to the reviewers who made it possible to increase the final scientific quality.

#### Paolo Biancone and Silvana Secinaro Editors





Luigi Corvo<sup>1</sup>, Lavinia Pastore<sup>1,\*</sup>, Arianna Manti<sup>1</sup> and Daniel Iannaci<sup>2</sup>

- <sup>1</sup> School of Economics, Department of Management and Law, University of Rome "Tor Vergata",
- Via Columbia 2, 00133 Rome, Italy; luigi.corvo@uniroma2.it (L.C.); arianna.manti@uniroma2.it (A.M.)
- <sup>2</sup> Department of Management, University of Turin, Corso Unione Sovietica 218 bis, 10134 Turin, Italy; daniel.iannaci@unito.it
- \* Correspondence: pastore@economia.uniroma2.it

**Abstract:** The social impact assessment (SIA) process is widely utilised and is receiving increasing interest from both scholars and practitioners. A systematic approach was applied in this study to search for articles about SIA models. In the first step, we analysed six main SIA model mappings between 2004 and 2015. In the second step, 98 models were identified. The main findings include the definition of emerging paths for the future research agenda on this topic. Compared with previous SIA mappings, we identified 22 additional models that are related to the sustainability discourse. The meaning of sustainability is defined both by the emergence of new systems in finance that require specific metrics and in relation to the global agenda towards sustainable development. It is interesting to notice how social impact models, sustainability indicators (under the global framework of sustainable development goals (SDGs)) and new financial scores (such as environmental, social and governance (ESGs)) are converging into a common discourse, even if divergence is still present, and further research is needed to unlock the relationships among them.

**Keywords:** social impact; social impact assessment; social impact models; sustainability; third sector; SIA models; comprehensive review

#### 1. Introduction

Over the last few years, attention has focused on the study of different models of measurement and evaluation of value that cannot be directly recognised through conventional financial metrics [1]. This growing attention has led to changes in organisations that operate in the third sector and in hybrid companies, which, at various levels, are increasingly trying to find credible solutions to demonstrate the social value generated through processes of measurement and social impact assessment (SIA).

Issues in SIA are receiving more attention from a whole range of social enterprises in what Young et al. [2] defined as the "Social Enterprise Zoo", each with a different approach and purpose.

At the same time, governments on regional and national scales have launched programmes to strengthen their ability to assess the social value generated by the evolution of the traditional welfare state. They are in the difficult position of having to respond to a growing demand for public goods and services with fewer resources available for public financing yet, at the same time, are being faced with challenges concerning the management of public debt, as dictated by spending constraints and increasing social need from an aging population and increased immigration [3,4].

When this study discusses social impact, it refers to Clark's definition [5]: "Social impact is the portion of the total outcome that occurred as a direct result of the intervention, net of that portion that would have occurred equally without the intervention".

The scope of social impact is the construction and transmission of a set of information capable of expanding and deepening the knowledge of the value generated to better

**Citation:** Corvo, L.; Pastore, L.; Manti, A.; Iannaci, D. Mapping Social Impact Assessment Models: A Literature Overview for a Future Research Agenda. *Sustainability* **2021**, *13*, 4750. https://doi.org/10.3390/ su13094750

Academic Editor: Antonio Miguel Martínez-Graña

Received: 12 March 2021 Accepted: 21 April 2021 Published: 23 April 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).



guide decision-making processes at different levels is precisely the treated topic of social impact [6–8].

A central issue is the debate about "who" is affected by the impact discussion. It is necessary to start from the assumption that the impact generated can be positive or negative, and this can be precisely the most immediate answer, because it is able to influence the well-being of people, the community and the community [9].

For this reason, each stakeholder, with different levels of priority, is interested in the issue. Some scholars argue that the primary stakeholder in impact evaluation is the PA [10]. We find impact as an element of relationship between different organizational configurations that populate a redrawn geography of value, where distinctions and perimeters are no longer marked by mere formal elements, but by different visions, intentions, and values. To address the question of who is interested in SIA, Klemelä [11] examined the role of legitimation means from the perspective of different stakeholders.

The study does not address the issue of impact for specific stakeholders, such as the public, private or non-profit sector. However, through this insight, the impact generated will directly and indirectly benefit the whole community.

When dealing with the topic of social impact, there are three key aspects to keep in mind: intentionality, measurability and additionality; the authors focus on the second mentioned. Assessing social impact is still not easy today, largely because of the difficulties in identifying qualitative and quantitative metrics to demonstrate the extent to which social impact is generated.

In many of the existing contributions, the effort needed to align theory and practice is noticeable. These studies show that in academic discourse, there are many assumptions that practitioners are not able to make for various reasons. For Smith and Stevens [5], this is particularly true for SIA models, where the developed metrics range from highly qualitative, self-developed input measures to more sophisticated quantitative output and impact measures.

However, the SIA process is widely utilised, and it is gaining increasing interest from both scholars and practitioners. In this study, we analysed SIA models in order to understand their main features and the emerging paths that may determine the future research agenda on this topic. We particularly noticed that the last academic work that tried to summarise the scientific production on SIA is a paper from six years ago [6], and since this time, the number of publications on this topic has continually increased, thus suggesting a need for a new systematisation of the literature.

The remainder of the paper is as follows. Section 2 presents the theoretical background of the study. Section 3 describes the method for conducting the literature review. Section 4 focuses on the analysis of the six main SIA model mappings that emerged from the literature. Section 5 presents an update of SIA model mappings, identifying 98 models. Section 6 consists of concluding considerations, limitations and ideas for the future research agenda.

#### 2. Theoretical Background

The definition of "impact" has been widely discussed in the literature as cited in the previous section [6,12–16]. Social Impact Assessment combines social research, public engagement, planning and social change management [17].

Social impact measurement aims to assess the social and environmental value produced by the activities or operations of any organization (for-profit, non-profit or public). Although any company can have a social impact, we will always need to distinguish between companies that are socially oriented, such as non-profit organizations, social enterprises and public bodies, and those that create it indirectly. For these reasons, it is necessary to make order among the entire panorama of models, precisely to create tools that can best represent the impact of an organization, enhancing its social vocation.

The relationship between impact and complexity seems particularly interesting considering the systems change perspective and the possibility of legitimizing forms of value generation not recognized by conventional financial metrics [18]. The complexity approach has unveiled the perspectives of SIA from being a performance construct specific to a minority portion of organizations to becoming a determinant for the generation of new cross-sectoral relationships among public, for-profit, and non-profit organizations

Arvidson and Lyon [19] stated that social impact can be perceived as a social construction. The complexity of SIA lies in this: there is no clear definition of what is meant by "social", so discretion must be involved when assessing social impact [19,20].

Through the analysis of the existing literature, we identified five streams of studies that discuss, from different points of view, the "social" meaning:

- (1) social value creation and corporate social responsibility (henceforth CSR) studies;
- (2) social enterprise (henceforth, SE) studies primarily focused on the issues of performance and accountability;
- (3) environmental impact studies;
- (4) public sector and impact finance studies;
- (5) developing economies studies.

(1) The first avenue is rooted in CSR around the notion of "creating social value", as developed by Porter and Kramer [21], in order to explain a deeper relationship between business and society. Organisations should move beyond the traditional belief that their economic value is separate from, and in conflict with, their social value by assuming the perspective of blended value, as coined by Emerson [12]. Within the wider process of value creation, organisations need to be aware of the importance of quantification in unlocking new value and creating valuable opportunities for innovation and growth that would otherwise be missed [6,22,23].

(2) The most consistent field of study concerning SIA is related to SE performance and accountability. Social impact is described as a combination of resources, inputs, processes or policies that occur as a result of the real, implied or imagined presence or actions of individuals in achieving their desired outcomes [24–26]. In order to assess the performance of an SE, Clark and Brennan [27] developed the Balanced Value Matrix (henceforth, BVM), which concludes that separate and balanced indicators exist for outputs (enterprise actions), outcomes (the benefits associated with enterprise actions) and impacts (the results that enterprises desire) [28].

It is worth contrasting this approach with the contribution of Clark et al. [5], who introduced the concept of the impact value chain (henceforth, IVC). Since then, IVCs have been widely used to better understand the relationship between programme inputs and outcomes and to discover which mechanisms of change are involved in moving from inputs to desired results [29]. Bagnoli and Megali [30] suggested that measurement of SE performance should consider economic and financial performance, social effectiveness and institutional legitimacy. According to Dart [31], SIA promotes improved accounting practices, increasing the legitimacy of the organisation with its stakeholders and enhancing the relationship of trust with the organisation's funders. In a context of scarcity and competition for funds, the use of standard procedures for assessing and reporting social outcomes might encourage investment in an SE that adopts them [22,32,33].

Based on a study of five SEs in the UK, Nicholls [34] argued that evaluations and audits are used as a means by which to "enhance social mission rather than merely to respond to regulation". He suggested that "emergent reporting practices constitute a spectrum of disclosure logics that social entrepreneurs exploit strategically to support their various mission objectives with key stakeholders". Impact measurement is thus seen as a part of a negotiation process between stakeholders.

In another study, Nicholls [35] showed how a "flexible reporting format can be used strategically in various ways by companies according to their particular objectives and resource limitations". Therefore, SIA is seen as a strategic opportunity for SE development [34,36].

(3) Bakar et al. [17] consider SIA to be a subfield of Environmental Impact Assessment (henceforth, EIA). EIA refers to the assessment of impacts concerning the environment. Although EIA was intended as an all-inclusive framework for analysing environmental and social issues, it failed to adequately address social issues [6], and, therefore, SIA was developed with a gradual extension of the items under consideration [37,38].

(4) Another field of study relates to the transformation of welfare systems and the role of impact finance. The increasing scarcity of public resources has led to innovation in economic relationships between public bodies and private organisations [39–43]. These changes present a challenge in the form of a hybrid market with unexplored potential, involving financial intermediaries and local bodies, small and medium-sized enterprises (henceforth, SMEs), large enterprises, SEs and civil society [44]. The Organization for Economic Cooperation and Development (OECD) report on "new investment approaches to meet the social and economic challenges" [45] stated that impact investing has declined as a result of the changing relationship between finance and philanthropy.

One of the issues addressed by this field of study is that the assessments are often under pressure to demonstrate short-term effects rather than to emphasise long-term impact. Although there are generally accepted accounting principles that support financial reporting, similar standards related to the measurement and communication of social impact have not been produced yet because it is difficult to arrive at a comprehensive definition of the concept of social impact, and the related measurement models often lack the rigour that characterises accounting approaches aimed at assessing financial returns [39].

(5) The last field of study is related to the "people aspect" of development-induced change by empowering communities with a voice in the EIA process [46]. Countries with emerging economies are especially affected by poverty, and SIA conducted within this context necessitates mitigation of both the direct impacts of development as well as the social legacies that can entrench poverty and inequality. Social development is seen as an approach that can be used to reduce poverty and inequality [46].

The proposed overview sheds light on a nebulous and confused approach to the topic of SIA. As highlighted by the OECD [1], the lack of a common language and understanding of the definition of "social impact" and the best way to measure it has hampered both academic debate and the adoption of SIA models amongst practitioners.

Within managerial studies are these five perspectives on the meaning of social impact. This heterogeneity is the reason why, in the literature, many models for assessing social impact have been developed over time. Therefore, some authors have dedicated their work to clustering, categorising and mapping SIA models.

#### 3. Method

A systematic approach was applied in this study to search for articles about SIA models. We used the Web of Science and Scopus databases to find relevant articles. For both databases, the following keywords were used for article searching: "Social impact assessment" OR "Social performance assessment" OR "Nonfinancial performance assessment" OR "Social return assessment" OR "ESG assessment" OR "Impact investing assessment" AND "Model\*". Table 1 shows the criteria used in searching and selecting articles.

Criteria for Searching and Selecting Articles	Web of Science Database	Scopus Database	Description
Searching articles using the keywords	101	245	The keywords used for searching are "Social impact assessment" OR "Social performance assessment" OR "Nonfinancial performance assessment" OR "Social return assessment" OR "ESG assessment" OR "Impact investing assessment" AND "Model*".
Selecting documents only in the article and review category	84	214	Documents in the article and review category were selected since those in other categories are not peer-reviewed or academic contributions.
Selecting articles written in English	82	207	We selected documents that are written in English.
Adding articles from both databases		289	82 articles from the Web of Science and 207 from Scopus databases were added to a single spreadsheet.
Removing duplicate documents from the lists in the databases		231	58 duplicates were removed.
Checking Title, Abstract and Keywords		187	The focus of 44 articles was not SIA models.
Adding more articles after checking the grey literature (NEF—New economic foundation; Tools and Resources for Assessing Social Impact—TRASI database)		43	An additional 43 articles coming from grey literature are focused on SIA models.
Finalizing the number of articles considered for this study		230	Finally, we reached 230 articles for consideration in this study.

Table 1. Searching and selection of articles from the Web of Science and Scopus databases, 1 December 2020.

For the Web of Science, our search was limited to searching titles, abstracts and keywords, resulting in 101 unique documents. After limiting the search by selecting only the article and review options, we arrived at 84 documents, which was further reduced to 82 by excluding articles not written in English. A list of these 82 articles was then downloaded as a comma-separated values (CSV) file and imported into a spreadsheet.

For the Scopus database with the same three keywords, 245 documents were found on the first search attempt. Limiting the search to the article's category brought this down to 214 articles and focusing further on articles written in English reduced this to 207 documents. Furthermore, we limited this search to the following five categories: Business, Management and Accounting Economics, Econometrics and Finance, Psychology, and Decision Sciences. We merged the 82 articles from the Web of Science and the 207 articles from Scopus into an Excel file. We were aware that some articles would be listed in both databases, so we removed 58 duplicates through spreadsheet filtering. Thus, we arrived at 231 unique articles. Upon checking the title, keywords and abstract of each article, we found 187 relevant articles for inclusion.

We decided to supplement the academic results by collecting 43 articles from the two most acknowledged practitioners' repositories: The Foundation Center (TRASI) and the NEF. We then had 230 articles from the searching process to include in this review. We read these papers with two main purposes:

- 1. Identifying previous studies that mapped the evolution of SIA models and then analysing those studies to verify whether they provided clusters or groups of SIA models;
- Investigating the models already reviewed by previous studies and the new SIA models proposed since the last mapping study to highlight emerging patterns and to shape the future research agenda in this field of research.

During the period under review (1980–2020), scientific production on Social Impact Models started in 1980 and underwent changes in 2003 and 2010. The number of publications in journals on the topic increased significantly between 2015 and 2020 (Figure 1). The data confirm the above. This field of research is constantly evolving, and the horizons are becoming wider and wider.



Figure 1. Articles per year.

#### 4. Six Studies in the Search for SIA Models

First of all, we must clarify that we use the term "model" to mean a named, documented process that is used to assess either the actual social and/or environmental impact and that, in the literature we analysed, the terms "approach", "method" and "model" are used interchangeably.

Six main attempts at SIA model mapping were found from both academics and practitioners. Those doing the mapping have had different approaches to grouping the SIA models, and they have used a variety of sources. Table 2 shows how the authors grouped the models mapped. It is interesting to notice that the groups are presented ex ante in the description of the models—based either on the process or on the usage—while the last mapping from Grieco et al. [6] clusters them ex post using variables identified from previous studies. The authors use the terms models, methods and systems to mean "the ways" of assessing social impact.

	SIA Mapping					
	Clark et al. [5]	Olsen and Galimidi [47]	Zappalà and Lyons [39]	Rinaldo [48]	Maas and Liket [13]	Grieco et al. [6]
	(1) Process models/methods	(1) Rating systems	(1) Social Accounting and Audit (SAA)	(1) Monitoring and evaluation tools	(1) Process methods	(1) Simple Social Quantitative
Groups/clusters	(2) Impact models/methods	(2) Assessment systems	(2) Logic Models	(2) Quality tools	(2) Impact methods	(2) Holistic Complex
	(3) Monetisation models/methods	(3) Management systems	(3) Social Return on Investment (SROI)	(3) Outcome tools	(3) Monetisation	(3) Qualitative Screening
		-	-	-	-	(4) Management

Table 2. Clusters/groups of SIA models between 2004 and 2015.

Although the labelling of the SIA groups is different between authors, there are similarities, such as the emergence of models from quality systems, models based on monetisation and models from rating systems.

The first mapping study [5] aims to provide a measurability framework for the double bottom line approach. The authors tried to respond to the need for more tangible accountability for the social impact created for each invested or granted dollar. Their approach can be defined as "investor driven", and the groups that they identify are thought of as having a three-step scale: the models that provide information only on the process, the models that inform about the social impact and the models that enable the transformation of the social impact information into financial metrics in order to combine the social and financial dimensions into a blended evaluation. They consider four key aspects in order to identify coherent groups of SIA models (the authors refer to them as "models/methods"): the completeness of the IVC along the sequence input-activities-output-outcome-goal alignment, the purpose of the assessment, the data feasibility and the possible sectors of application.

While the first mapping study emphasises the monetisation capacity of the SIA models, the second one [47] intends to shed light on the function of each model. The three possible functional types are rating systems, which are models using a fixed set of indicators and the impact investment's quality or potential quality, summarised by a score or symbol; assessment systems, which are models that use a fixed or customised set of indicators at a point in time, evaluate characteristics, practices and/or results of portfolio investments but do not provide explicit tools to manage the tracking of operational data by the organisation over time; and management systems, which are models that provide tools for organisations to manage detailed operational information about drivers of impact. Olsen and Galimidi [47] used three variables provided by Clark et al. [5] (purpose, data feasibility and sector) and three new variables. The first of these new variables relates to the methodological approach adopted by the SIA model, the second investigates the types of data management required by the SIA models, and the third looks into the presence of verification criteria, thus improving the credibility of the results provided by the model.

The third mapping study provided by Zappalà and Lyons [39] identified three groups of SIA models under a different point of view related to the connections between the SIA process and conventional managerial processes. The first is connected to the accounting and auditing process, and it enables the understanding of the social impact on the surrounding community and beneficiaries and builds accountability by engaging with key stakeholders. The second is connected to a logical framework and project management processes, and it provides a systematic and visual way for individuals to present and share their understanding of the relationships among the resources available to operate their programme (inputs), the activities that they plan to do (strategies) and the changes or results that they hope to achieve (outcomes and impact). The third is explicitly named "social return on investment" and is connected to financial management and investor relationship processes, providing a single number that is the result of a ratio between the monetary value of the positive changes achieved through the activities and the budget invested. Zappala and Lyons [39] used eight variables, of which three are new compared with those used by Olsen and Galimidi [47]. The scope of the analysis here is split into two: clarity of purpose and scope. The presence of comparative impact data or information and the materiality analysis are the two new variables.

The fourth mapping study by Rinaldo [48] is focused on the tools provided by scholars and practitioners in order to improve the organisational capacity to measure and evaluate social impact. She identified three types of tools: monitoring and evaluation tools, quality systems and outcome tools. The monitoring and evaluation tools can inform about what data to collect, when to collect it and who will collect it. This information can provide proof that a performance goal has been met, or it can support social impact measurement. The quality tools focus on how things are done. They look at how an organisation is run, how staff are managed and customer care. A set of standards are defined and used to gauge areas for improvement. Some quality tools focus on how activities are carried out, and others also require evidence about the results of these activities. This evidence requirement would have the additional benefit of providing information for impact measurement. The outcome tools are used to measure and record the progress that a beneficiary makes and pinpoint areas of future need. They make it possible to assess the changes made in a consistent and standardised way. Outcome tools provide information that can be drawn together to give an overview of the change achieved by a service or project. They are therefore a key part of the impact measurement process. The variables used by Rinaldo [48] are for investigating the usability of these tools in terms of cost-efficiency and cost-effectiveness, so they investigate the cost of each tool, its complexity, the time required to embed it within organisational processes, how demanding the tool is for the staff in terms of the training required to implement it and the provision (and its cost) of the support required to process it.

In the fifth mapping study, Maas and Liket [13] used the same groups provided in [5]. Nevertheless, they improved the analysis by considering a larger number of models (30 vs. 9 models) and a more consistent set of variables and by introducing the time frame variable in the analysis. The capacity to determine the time factor of the models opens new perspectives for the SIA, making the measurement and evaluation process circular: social impact forecasting as an ex ante evaluation of the expected results, social impact monitoring as an ongoing step for checking the intermediary data and the ex post evaluation for informing about the changes achieved and the improvement areas to stimulate as drivers for the next strategic planning cycle. Another relevant variable that Maas and Liket [13] added is related to the different perspectives that SIA models can assume. When they originate from business (micro perspective), for example, they include indicators that differ from the indicators used for assessing the impact on the urban/rural territory (meso perspective), and even more different are the indicators used when the perspective is wider and encompasses socioeconomic dimensions at the country or regional level (macro perspective). Depending on the perspective used, different indicators will be used, and therefore, different impacts will be measured.

The latest and most comprehensive mapping study [6] employed a combination of the variables used prior to 2015 to carry out a hierarchical cluster analysis. More specifically, Grieco et al. [6] integrated six variables with those identified by previous authors and added one more variable that refers to the developer of the SIA model (actors involved in the development of SIA models). Table 3 summarises the variables used and their references.

Table 3. Variables and references used by Grieco et al. [6].

Grieco et al. [6]			
Variables	References		
(1) Data typology	Nicholls [14]		
(2) Impact typology	Rinaldo [48]		
(3) Purpose	Clark et al. [5], Rinaldo [48], Maas and Liket [13]		
(4) Model complexity	Zappalà and Lyons [39], Maas and Liket [13]		
(5) Sector	Olsen and Galimidi [47]		
(6) Time frame	Maas and Liket [13]		
(7) Developer	Identified by the authors [6]		

Applying these variables to the SIA models mapped (i.e., 76) resulted in four clusters:

- o Cluster 1 (Simple Social Quantitative) contains models based on quantitative indicators. These models are easy, applicable to any sector and intend to produce a quantitative measure of the social impact and of the impact on employees with a retrospective time frame.
- o Cluster 2 (Holistic Complex) contains models characterised by a holistic purpose, and this explains the presence of both qualitative and quantitative variables. The aim of these models is to provide evidence to obtain funding, so they focus on reporting and

communication of the results achieved. These are also applicable to any sector, but in this case, the complexity is high.

- o Cluster 3 (Qualitative Screening) consists of models based on qualitative variables and are usually focused on holistic impacts. They are retrospective and have a basic level of complexity.
- Cluster 4 (Management) contains models based on qualitative or quantitative variables that aim to measure different types of impacts. They are used for managerial or certification reasons. Usually, they are applied to ongoing activities.

#### 5. Model Mapping

Previous mappings have used different approaches to grouping SIA models and a variety of sources, and therefore, they have analysed a variable number of models. It is important to underline that the purpose of some of the previous mapping was not to collect all models developed but to become a practical guidance for assessing social impact. These mappings aimed to become a reference point for those who have to choose what kind of SIA model to implement according to the purpose of the assessment. As already described, the very first list of models was developed by Clark et al. [5] and was based on grey literature and interviews with key stakeholders. Indeed, the interest in systematising SIA models emerged from practitioners and, over time, has gained importance in academic studies. As shown in Table 4, the number of models has varied over time, as well as the number of patterns identified. It is possible to observe a growing trend in the number of models listed, aside from Zappalà and Lyons [39], who decided not to count the models but to approach them directly by dividing them into groups based on their main characteristics.

Table 4. Summary of SIA mapping (source: authors' elaboration).

Authors	Year of Publication	No. of Models Analysed	Emerging Patterns (No.)	Sources
Clark et al. [5]	2004	9	3	GL + I
Olsen and Galimidi [47]	2008	25	3	GL + I
Zappalà and Lyons [39]	2009	N.A.	3	AL
Rinaldo [48]	2010	19	3	GL
Maas and Liket [13]	2011	30	-	AGL + I
Grieco et al. [6]	2015	76	4	AGL

Notes: GL: Grey literature; I: interviews; AL: Academic literature; AGL: Academic and grey literature.

Ninety-eight social impact measurement models were identified. Appendix A provides the names of the models mapped. In comparison with the latest mapping by Grieco et al. [6], 22 new models were found, 52 were confirmed, and 24 have been renamed but are coherent with the previous list. In addition, their study, as in former SIA mappings, does not include models that assess only the internal organisational efficiency, models that cannot be linked to an organisation (for instance, policy evaluation models are not listed) or models for which the information is too little to be included.

It is important to underpin that there is a semantic issue concerning SIA models; therefore, this mapping, as with the former ones, has to be considered as partial and not exhaustive. Aside from specific models that are establishing themselves as reference points, most of the models might have different names but actually reproduce the same process and consider the same variables to assess social impact. This is the reason why 24 of the latest mappings have been identified; even if the name of the model itself might have changed over time and with different research parameters (for instance, other databases), other names might be found. The most used model is social return on investment (SROI) and other monetisation models that have been developing around it. Another two very common models come from management systems and quality systems such as EMAS (Eco-Management and Audit Scheme) and EFQM (European Foundation for Quality Management Excellence Model).

It is interesting to observe a new category of models related to the issue of sustainability and finance. Among the 22 new models, 10 of them are related to the assessment of environmental, social and governance (ESG) performance [49], and 5 of them are associated with sustainable development goals (SDGs) and sustainability assessment. In terms of ESG, it is crucial to consider the latest OECD (2020) report that underlines the variety of ESG scores and ratings and the complexity of this kind of assessment. The OECD report cites many hundreds of indexes and ratings to assess ESGs that correspond to a similar number of firms providing this service, even though there are only three main market players. The ESG models listed in our mapping are just a small portion of the high number of rating systems that are rapidly developing in consultancy firms and investment funds. Those mapped are the ones that were already mentioned in academic or grey literature involving the discourse on SIA models. This opens up future research questions that could investigate the relationship between SIA studies, sustainability and finance. In particular, ESGs might have an important role in the development of "alternative" finance, since they are identified in the OECD report [50] as a measure for social impact investing and sustainable and responsible investing (SRI).

The other important topic of new models relates to sustainability and expands the already listed Global Reporting Initiative (GRI) type of model. There are five models that focus on the assessment of SDGs. SDGs are a global framework with a shared and specific set of targets; it is interesting that different types of SIA models are arising to assess if and how an organisation is contributing to the reaching of global targets.

The novelty of this mapping comprises the emergence of sustainability as a new keyword of SIA models in relation to both finance and enterprise assessment.

#### 6. Discussion and Conclusions

#### 6.1. Contributions to the Literature

The authors show that the analysis conducted consisted of two steps. First, six major mappings of SIA models between 2004 and 2015 were analysed. As a second step, 98 models were identified. The year 2015 was very important scientifically for the topic because the policy context of sustainability changed with the inclusion of the SDGs [51].

Until now, the research in this field has almost never led to shared solutions, and this finds direct evidence in the plurality of models adopted for social impact measurement and evaluation, representative of highly differentiated approaches and tools. This condition is generated by the fragmentation among SIA models and the variety is high, apart from the very few models that present a clear methodology and features, e.g., the SROI [8], most models are not standardized (at least in the process) [9]. This variety certainly covers a wider range of dimensions for assessing social value and accommodates the diversity of each entity (from for-profit companies to social enterprises, from benefit companies to non-profits), but at the same time, it has the limitation of making it much more difficult to scale assessments [52]. The fuzziness of SIA models also affects impact finance, which should be the system in which these news metrics are considered [53].

For this reason, the purpose of this paper is to provide a literature review of SIA models and an update of the latest mapping from 2015. The analysis of previous studies was critical to better understand how to interpret the mapping and to question how to expand this research study. In fact, the other six mapping studies revealed clusters, groups and variables. The model grouping or clustering usually consists of three main labels: (1) models that come from performance/management system studies, of which the most used is EMAS; (2) models that come from auditing/quality system studies, of which the most used is EFQM; and (3) models that aim to monetise the outcome, called monetisation models, of which the most known and used is SROI. In the 98 models analysed, we can identify these groups, although we cannot yet fully describe the characteristics and frequencies. The next step in this research study was to perform a cluster analysis of our group by selecting some of the variables from previous studies. From our perspective, it appears that a fourth group/cluster of patterns may emerge, and that is the one related

to sustainability. As mentioned in the previous mapping, the topic of sustainability was present through the inclusion of GRI among SIA models but did not have specific importance. From our mapping, sustainability seems to be the main driver of the 22 new models found. The meaning of sustainability is determined by the emergence of new systems in finance requiring specific metrics and in relation to the global agenda towards sustainable development. Further research should focus on the relationship between social impact studies and sustainability studies as two frameworks that could partially overlap and integrate. The integration of the two perspectives consists primarily of a theoretical problem [54].

This brings to light how impactful the grafting of global organisations is to provide a direction for innovation in terms of social impact.

#### 6.2. Implications for Managers

As we have shown with this research, the SIA process is widely implemented and is capturing increasing interest from both scholars and practitioners. In this article, we analysed the models of SIA to understand their main characteristics and the emerging pathways that may determine the future research agenda on this topic [55].

In particular, we noted that the last academic work that attempted to summarise the scientific production on SIA is a 6-year-old paper [6], and since then, the number of publications on this topic has steadily increased, thus suggesting a need for a new systematisation of the literature.

The results show that the benefits of the SIA topic are growing, and without a doubt, companies need to equip themselves with good assessment tools. Managers must also anticipate the need for information by their stakeholders, and therefore, it is essential to identify an excellent reporting tool that may require dedicated efforts and procedures [56,57]. Based on this study's preliminary evidence, and if supported by further research, business decision makers can improve the effects of their actions internally and externally, even when reflection on best practices is not perceived as urgent. The findings suggest that these micro-processes can be supported by an entrepreneurial attitude that allows business managers to regularly take stock and be ready to act quickly by being aware of their company's financial and non-financial data, especially in a language that is clearly accessible. Too often, smaller companies entering the market do not have the time or resources to make these assessments, and this can lead to inefficiencies that last longer than necessary, resulting in wasted resources and poor returns, as well as reduced opportunities for learning and adapting practices.

As mentioned earlier, the OECD report discusses many hundreds of indices and ratings to assess ESGs that correspond to a similar number of companies providing this service, although there are only three main players in the market. The ESG models listed in our mapping are only a small part of the large number of rating systems that are rapidly developing in advisory firms and investment funds. It is therefore crucial for managers to equip themselves with professionals who know how to anticipate problems and manage this type of know-how.

#### 6.3. Limitations and Future Research

The scenario demonstrated has strong illustrative and exploratory potential, and the steps identified in this study can be adapted to other contexts [58].

The purpose of our exploratory study is to provide insights that other scholars can draw upon and explore further in the theory development process. Therefore, this study invites scholars to investigate the transferability of our insights and provides several promising avenues for future research.

Our contribution has some limitations. First, SIA models have a fragmentation problem in their taxonomy, so it is possible that some models may not have been found, or others may have been created and not intercepted through the four databases searched. Particularly with regard to the grey literature, it is very difficult to capture the continued innovation around SIA models [59,60].

These findings provide the basis for providing support to the scientific research sector, third sector agents, investors and all stakeholders working with social entrepreneurs to better understand what research and SEs will need to focus on to generate social impact [61–63]. Researchers question whether the focus on social impact can generate transparency and accountability in all non-profit [64] but also for-profit contexts, as the current literature provides us with additional distinctions in this category: for impact and without impact.

Second, the listed models have not yet been investigated through a cluster analysis to identify key characteristics. Furthermore, the methodology adopted may be a limitation in that it requires some discretion on the part of the researchers and, consequently, introduces the potential for bias in conducting the analysis.

However, this paper offers quantitative results on the SIA model, whereas most studies focus on qualitative insights. Limitations arising from the methodology can be addressed in future research by, for example, extending the analysis to a different sample of reports or, even more importantly, supplementing the cluster identification with a field analysis that would allow for a deeper understanding of actual practices [4].

The findings highlight this study as an opportunity to direct future research to fill gaps in the literature. Since the SDGs are goals to be achieved by 2030 (2030 Agenda), evidence of these gaps can increase awareness of scientific production and thus facilitate the achievement of the goals. In this sense, the literature places SIA at the centre of social innovation [4].

Third, to reach more general conclusions, the 2030 goals towards which the world is racing seem to be set by large institutions and public bodies. This brings to light how impactful the work of global organisations is in providing direction for social impact innovation [65].

The generation of social impact creates a need for reporting on this data and thus for identifying good reporting tools on the global goals that humanity must now aim for.

In conclusion, with the aim of mapping SIA models, addressing these issues contributes to the achievement of sustainable goals, crystallises the academic studies in this field and demonstrates the evolution of the perception of assessment models in the literature. This research confirms that it plays a central role in the topic of the innovation-oriented social impact of the entire ecosystem. This condition increases accountability, transparency and stakeholder engagement. Stakeholders will be able to better understand the strong values of the company itself and all the partnerships that are normally created to achieve its goals, in line with SDG 17.

**Author Contributions:** Conceptualization, L.C., L.P.; methodology, L.C.; data curation, A.M.; validation, L.P., L.C.; writing—original draft preparation, L.C. (Sections 3–5), L.P. (Sections 1, 2 and 5), D.I. (Section 6); writing—review and editing L.P., L.C.; resources, D.I., A.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Conflicts of Interest:** The authors declare no conflict of interest.

#### Appendix A

SIA MODELS				
1	AA1000AP	47	Logic model builder	
2	Acumen Lean Data	48	LuxFLAG ESG Label	
3	Acumen scorecard	49	Measuring impact framework	
4	Anticipated Impact Measurement and Monitoring (AIMM)	50	Methodology for impact analysis and assessment	
5	AtKisson compass assessment for investors	51	MetODD-SDG	
6	Best available charitable option	52	MicroRate	
7	Bridges Ventures Impact Radar	53	Movement above the US\$1 a day threshold	
8	CERISE-IDIA	54	MSCI ESG Ratings Methodology	
9	Charity analysis framework	55	Ongoing Assessment of Social Impacts (OASIS)	
10	Cost per impact	57	Outcome star	
11	Cradle-to-cradle certification	58	Practical quality assurance system for small organisations (PQASSO)/Trusted Charity	
12	Dalberg Approach	59	Progress out of poverty index	
13	DTA Fit for purpose	60	Prove it!	
14	Eco-mapping	61	Public value scorecard	
15	EFQM	62	Quality first	
16	EMAS	63	RobecoSam 3 step SDG Framework	
17	Environmental, Social and Governance (ESG) Scores	65	SASB Standard	
18	EPIC	67	SDG Impact Practice Standard	
19	ESG Disclosure score	68	Social accounting and audit	
20	ESG Relevance Score	69	Social Business Scorecard	
21	ESG Risk Rating	70	Social enterprise balanced scorecard	
22	European Impact Investing Luxembourg	71	Social enterprise mark	
23	Expected return	73	Social Impact Measurement for Local Economies (SIMPLE)	
24	Family of measures	74	Social rating	
25	Finance Initiative Impact Radar	75	Social return assessment	
26	FMO ESG Toolkits	76	Social return on investment	
27	FTSE ESG Ratings	77	Social Value Maturity Index	
28	Global Alliance for Banking on Values (GABV)	78	Social value metrics	
29	Global Impact Investing Rating System (GIIRS)	79	Sopact-tool	
30	GOGLA Impact Metrics	80	SPI4	
31	GRI sustainability reporting framework	82	Standard Ethics Rating (SER)	
32	HIP Rating	83	Star social firm	
33	HIPSO Harmonized Indicators for Private Sector Operations	84	Success measures data system	
34	VALORIS method	85	The B impact rating system	
35	Impact Analysis for Corporate Finance & Investments (Tool prototype)	86	The big picture	
36	Impact Due Diligence Tools	87	The Committee on Sustainability Assessment (COSA) Methodology	
37	Impact Management Project (IMP) Five Dimensions	88	The FINCA client assessment tool	
38	Impact Multiple of Money (IMM)	89	The Impact Due Diligence Guide	
39	Impact Risk Classification (IRC)	90	The SRI LABEL	
40	Impact-Weighted Accounts	91	Third sector performance dashboard	
41	Inrate ESG Impact Rating Methodology	92	TIMM	
42	Inventory of Business Indicators (SDG Compass)	94	Trucost	
43	IRIS + (and IRIS)	96	Volunteering impact assessment toolkit	
44	ISS ESG Corporate Rating	97	Wallace assessment tool	
45	ISS SDG Impact rating	98	Y Analytics	
46	LM3			

#### References

- 1. OECD. Policy Brief on Social Impact Measurement for Social Enterprises. In *Policies for Social Entrepreneurship*; European Commission Luxembourg: Luxembourg, 2015; ISBN 978-92-79-47475-0. Available online: https://www.oecd.org/social/PB-SIM-Web\_FINAL.pdf (accessed on 15 April 2021).
- 2. Young, D.R.; Searing, E.A.; Brewer, C.V. The Social Enterprise Zoo: A Guide for Perplexed Scholars, Entrepreneurs, Philanthropists, Leaders, Investors, and Policymakers; Edward Elgar Publishing: Cheltenham, UK, 2016.
- 3. OECD. International Migration Outlook 2018; OECD Publishing: Paris, France, 2018. [CrossRef]
- 4. Tang, M.; Liao, H.; Wan, Z.; Herrera-Viedma, E.; Rosen, M.A. Ten Years of Sustainability (2009 to 2018): A Bibliometric Overview. *Sustainability* 2018, 10, 1655. [CrossRef]
- 5. Clark, C.; Rosenzweig, W.; Long, D.; Olsen, S. *Double Bottom Line Project Report: Assessing Social Impact in Double Bottom Line Ventures*; Working Paper Series No. 13; University of California: Berkeley, CA, USA, 2004.
- Grieco, C.; Michelini, L.; Iasevoli, G. Measuring value creation in social enterprises: A cluster analysis of social impact assessment models. *Nonprofit Volunt. Sect. Q.* 2015, 44, 1173–1193. [CrossRef]
- Lyon, F.; Sepulveda, L. Mapping social enterprises: Past approaches, challenges and future directions. *Soc. Enterp. J.* 2009, *5*, 83–94. [CrossRef]
- 8. Then, V.; Schober, C.; Rauscher, O.; Kehl, K. Social Return on Investment Analysis; Springer: Berlin, Germany, 2017.
- 9. Corvo, L.; Pastore, L. The Usefulness of Sharing Social Impact Data. Early Findings from an International Benchmarking on SROI Assessments. *J. Entrep. Organ. Divers. (JEOD) Creat. Commons Attrib.* **2020**, *9*, 45–61. [CrossRef]
- 10. Massey, A.; Johnston-Miller, K. Governance: Public governance to social innovation? Policy Politics 2016, 44, 663–675. [CrossRef]
- 11. Klemelä, J. Licence to operate: Social Return on Investment as a multidimensional discursive means of legitimating organisational action. *Soc. Enterp. J.* 2016, *12*, 387–408. [CrossRef]
- 12. Emerson, J. The Blended Value Proposition: Integrating Social and Financial Returns. *Calif. Manag. Rev.* 2003, 45, 35–51. [CrossRef]
- 13. Maas, K.; Liket, K. Social impact measurement: Classification of methods. In *Environmental Management Accounting and Supply Chain Management*; Springer: Berlin/Heidelberg, Germany, 2011.
- 14. Nicholls, A. *Measuring Impact in Social Entrepreneurship: New Accountabilities to Stakeholders and Investors?* ERSC Seminar, Local Government Research Unit: London, UK, 2005.
- 15. Dietz, T. Theory and method in social impact assessment. Sociol. Ing. 1987, 57, 54-69. [CrossRef]
- 16. Vanclay, F. Conceptual and methodological advances in social impact assessment. The international handbook of social impact assessment. *Concept. Methodol. Adv.* **2003**, 1–9. [CrossRef]
- 17. Bakar, A.A.; Osman, M.M.; Bachok, S.; Zen, I. Social impact assessment: How do the public help and why do they matter? *Procedia-Soc. Behav. Sci.* 2015, 170, 70–77. [CrossRef]
- 18. Hervieux, C.; Voltan, A. Toward a systems approach to social impact assessment. Soc. Enterp. J. 2019, 15, 264–286. [CrossRef]
- 19. Arvidson, M.; Lyon, F. Social Impact Measurement and Non-profit Organisations: Compliance, Resistance, and Promotion. *Volunt. Int. J. Volunt. Nonprofit Organ.* **2014**, *25*, 869–886. [CrossRef]
- 20. Manzoor, F.; Wei, L.; Nurunnabi, M.; Subhan, Q.A.; Shah, S.I.A.; Fallatah, S. The Impact of Transformational Leadership on Job Performance and CSR as Mediator in SMEs. *Sustainability* **2019**, *11*, 436. [CrossRef]
- 21. Porter, M.E.; Kramer, M.R. The Big Idea: Creating Shared Value. How to reinvent capitalism—And unleash a wave of innovation and growth. *Harv. Bus. Rev.* 2011, *89*, 62–77.
- 22. Porter, M.E.; Hills, G.; Pfitzer, M.; Patscheke, S.; Hawkins, E. Measuring Shared Value: How to Unlock Value by Linking Business and Social Results; by FSG Creative Commons Attribution-NoDerivs 3.0. 2012. Available online: https://www.hbs.edu/ris/Publication%20Files/Measuring\_Shared\_Value\_57032487-9e5c-46a1-9bd8-90bd7f1f9cef.pdf (accessed on 15 April 2021).
- 23. Kozień, A. The Principle of Sustainable Development as the Basis for Weighing the Public Interest and Individual Interest in the Scope of the Cultural Heritage Protection Law in the European Union. *Sustainability* **2021**, *13*, 3985. [CrossRef]
- 24. Emerson, J.; Wachowicz, J.; Chun, S. Social return on investment: Exploring aspects of value creation in the nonprofit sector. In *Social Purpose Enterprises and Venture Philanthropy in the New Millennium*; Investor Perspectives, REDF Workshop; REDF: San Francisco, CA, USA, 2000; Volume 2, pp. 130–173. Available online: https://redf.org/wp-content/uploads/REDF-Box-Set-Vol.-2 -SROI-Paper-2000.pdf (accessed on 15 April 2021).
- 25. Latané, B. The psychology of social impact. Am. Psychol. 1981, 36, 343. [CrossRef]
- 26. Bergmann, T.; Utikal, H. How to Support Start-Ups in Developing a Sustainable Business Model: The Case of an European Social Impact Accelerator. *Sustainability* **2021**, *13*, 3337. [CrossRef]
- 27. Clark, C.; Brennan, L. Entrepreneurship with social value: A conceptual model for performance measurement. *Acad. Entrep. J.* **2012**, *18*, 17.
- Yang, C.-L. Building a Performance Assessment Model for Social Enterprises-Views on Social Value Creation. *Sci. J. Bus. Manag.* 2014, 2, 1. [CrossRef]
- 29. Ebrahim, A.S.; Rangan, V.K. The Limits of Nonprofit Impact: A Contingency Framework for Measuring Social Performance. *SSRN Electron. J.* **2010**. [CrossRef]
- 30. Bagnoli, L.; Megali, C. Measuring Performance in Social Enterprises. Nonprofit Volunt. Sect. Q. 2009, 40, 149–165. [CrossRef]
- 31. Dart, R. The legitimacy of social enterprise. *Nonprofit Manag. Leadersh.* 2004, 14, 411–424. [CrossRef]

- 32. Ruttman, R. New ways to invest for social and environmental impact. In *Investing for Impact: How Social Entrepreneurship Is Redefining the Meaning of Return;* Credit Suisse with Schwab Foundation for Social Entrepreneurship: Zurig/Davos, UK, 2012. Available online: https://www.longfinance.net/media/documents/cs\_impactinvesting\_2012.pdf (accessed on 15 April 2021).
- 33. Esposito, P.; Brescia, V.; Fantauzzi, C.; Frondizi, R. Understanding Social Impact and Value Creation in Hybrid Organizations: The Case of Italian Civil Service. *Sustainability* **2021**, *13*, 4058. [CrossRef]
- 34. Nicholls, A. 'We do good things, don't we?': 'Blended Value Accounting' in social entrepreneurship. *Account. Organ. Soc.* 2009, 34, 755–769. [CrossRef]
- 35. Nicholls, A. Institutionalizing social entrepreneurship in regulatory space: Reporting and disclosure by community interest companies. *Account. Organ. Soc.* 2010, *35*, 394–415. [CrossRef]
- 36. Di Fabio, A.; Peiroó, J.M. Human Capital Sustainability Leadership to Promote Sustainable Development and Healthy Organizations: A New Scale. *Sustainability* **2018**, *10*, 2413. [CrossRef]
- 37. Esteves, A.M.; Franks, D.M.; Vanclay, F. Social impact assessment: The state of the art. *Impact Assess. Proj. Apprais.* **2012**, *30*, 34–42. [CrossRef]
- Richmond, B.J.; Mook, L.; Jack, Q. Social accounting for nonprofits: Two models. *Nonprofit Manag. Leadersh.* 2003, 13, 308–324. [CrossRef]
- Zappalà, G.; Lyons, M. Recent Approaches to Measuring Social Impact in the Third Sector: An Overview; Centre for Social Impact: Sydney, NSW, Australia, 2009. Available online: https://www.socialauditnetwork.org.uk/files/8913/2938/6375/CSI\_Background\_ Paper\_No\_5\_-\_Approaches\_to\_measuring\_social\_impact\_-\_150210.pdf (accessed on 15 April 2021).
- 40. Zamagni, S.; Venturi, P.; Rago, S. Valutare l'impatto sociale. La questione della misurazione nelle imprese sociali. *Impresa Soc.* **2015**, *6*, 77–97.
- 41. Corvo, L.; Pastore, L. The challenge of Social Impact Bond: The state of the art of the Italian context. *Eur. J. Islam. Financ.* **2019**. [CrossRef]
- 42. Meneguzzo, M.; Galeone, P. La finanza sociale. In *Pubblico, Privato, Non Profit: Le Prospettive Comuni in Europa e in Italia*; Rubbettino: Soveria Mannelli, Italy, 2016.
- 43. Biancone, P.P.; Radwan, M. Social Finance and Unconventional Financing Alternatives: An Overview. *Eur. J. Islam. Financ.* 2018. [CrossRef]
- 44. Brown, A.; Swersky, A. The First Billion; The Boston Consulting Group, Big Society Capital: London, UK, 2012.
- 45. Wilson, K.E. Social Investment: New Investment Approaches for Addressing Social and Economic Challenges. In *OECD Science, Technology and Industry Policy Paper*; No. 15; OECD Publishing: Paris, France, 2014; Available online: https://ssrn.com/abstract= 2501247 (accessed on 15 April 2021).
- 46. Aucamp, I.; Lombard, A. Can social impact assessment contribute to social development outcomes in an emerging economy? *Impact Assess. Proj. Apprais.* 2017, *36*, 173–185. [CrossRef]
- Olsen, S.; Galimidi, B. Catalog of Approaches to Impact Measurement: Assessing Social Impact in Private Ventures; Rockfeller Foundation: New York, NY, USA, 2008. Available online: http://www.midot.org.il/Sites/midot/content/Flash/CATALOG%20OF%20 APPROACHES%20TO%20IMPACT%20MEASUREMENT.pdf (accessed on 15 April 2021).
- Rinaldo, H. Getting Started in Social Impact Measurement: A Guide to Choosing How to Measure Social Impact; Norwich Guild: Norwich, UK, 2010. Available online: https://www.socialauditnetwork.org.uk/files/8113/4996/6882/Getting\_started\_in\_ social\_impact\_measurement\_-\_270212.pdf (accessed on 15 April 2021).
- 49. Boffo, R.; Patalano, R. Esg Investing: Practices, Progress and Challenges; Technical Report; OECD: Paris, France, 2020.
- 50. OECD. OECD Social Impact Investment 2019: The Impact Imperative for Sustainable Development; OECD Publishing: Paris, France, 2019.
- 51. Faraudello, A.; Barreca, M.; Iannaci, D.; Lanzara, F. The Impact of Social Enterprises: A Bibliometric Analysis from 1991 to 2020. *Int. J. Financ. Res.* 2021, 12, 3. [CrossRef]
- 52. Arce-Gomez, A.; Donovan, J.D.; Bedggood, R.E. Social impact assessments: Developing a consolidated conceptual framework. *Environ. Impact Assess. Rev.* 2015, *50*, 85–94. [CrossRef]
- 53. Spiess-Knafl, W.; Scheck, B. Impact Investing: Instruments, Mechanisms and Actors; Springer: Berlin, Germany, 2017.
- 54. Bonilla-Alicea, R.J.; Fu, K. Systematic Map of the Social Impact Assessment Field. Sustainability 2019, 11, 4106. [CrossRef]
- 55. Lenzo, P.; Traverso, M.; Salomone, R.; Ioppolo, G. Social Life Cycle Assessment in the Textile Sector: An Italian Case Study. *Sustainability* 2017, 9, 2092. [CrossRef]
- 56. Biancone, P.P.; Secinaro, S.; Brescia, V.; Iannaci, D. Communication and Data Processing in Local Public Group: Transparency and Accountability. *Int. J. Bus. Manag.* 2018, *13*, 20–37. [CrossRef]
- 57. Biancone, P.; Secinaro, S.; Brescia, V.; Iannaci, D. The Popular Financial Reporting between Theory and Evidence. *Int. Bus. Res.* **2019**, *12*, 45. [CrossRef]
- 58. Welch, E.W. The relationship between transparent and participative government: A study of local governments in the United States. *Int. Rev. Adm. Sci.* **2012**, *78*, 93–115. [CrossRef]
- 59. Secinaro, S.; Calandra, D.; Petricean, D.; Chmet, F. Social Finance and Banking Research as a Driver for Sustainable Development: A Bibliometric Analysis. *Sustainability* **2020**, *13*, 330. [CrossRef]
- 60. Baraibar-Diez, E.; Luna, M.; Odriozola, M.D.; Llorente, I. Mapping Social Impact: A Bibliometric Analysis. *Sustainability* **2020**, *12*, 9389. [CrossRef]

- 61. Burdge, R.J. Benefiting from the practice of social impact assessment. *Impact Assess. Proj. Apprais.* 2003, 21, 225–229. [CrossRef]
- 62. Mitzinneck, B.C.; Besharov, M.L. Managing Value Tensions in Collective Social Entrepreneurship: The Role of Temporal, Structural, and Collaborative Compromise. J. Bus. Ethics **2019**, 159, 381–400. [CrossRef]
- 63. Secinaro, S.; Corvo, L.; Brescia, V.; Iannaci, D. Hybrid organizations: A Systematic Review of the Current Literature. *Int. Bus. Res.* **2019**, *12*, 1–21. [CrossRef]
- 64. Barman, E. What is the Bottom Line for Nonprofit Organizations? A History of Measurement in the British Voluntary Sector. *Volunt. Int. J. Volunt. Nonprofit Organ.* **2007**, *18*, 101–115. [CrossRef]
- 65. Aznar-Crespo, P.; Aledo, A.; Melgarejo-Moreno, J.; Vallejos-Romero, A. Adapting Social Impact Assessment to Flood Risk Management. *Sustainability* **2021**, *13*, 3410. [CrossRef]





### The Principle of Sustainable Development as the Basis for Weighing the Public Interest and Individual Interest in the Scope of the Cultural Heritage Protection Law in the European Union

Adam Kozień 匝

Article

Doctoral School in the Social Sciences, Jagiellonian University in Kraków, 31-007 Kraków, Poland; a.kozien@doctoral.uj.edu.pl

Abstract: The concept of sustainable development is widely used, especially in social, environmental and economic aspects. The principle of sustainable development was derived from the concept of sustainable development, which appears in legal terms at the international, EU, national and local levels. Today, the value of cultural heritage that should be legally protected is indicated. A problematic issue may be the clash in this respect of the public interest related to the protection of heritage with the individual interest, expressed, e.g., in the ownership of cultural heritage designates. During the research, scientific methods that are used in legal sciences were used: theoretical-legal, formal-dogmatic, historical-legal methods, as well as the method of criticism of the literature, and legal inferences were also used. The analyses were carried out on the basis of the interdisciplinary literature on the subject, as well as international, EU and national legal acts—sources of the generally applicable law. Research has shown that the interdisciplinary principle of sustainable development, especially from the perspective of the social and auxiliary environmental aspect, may be the basis for weighing public and individual interests in the area of legal protection of cultural heritage in the European Union. It was also indicated that it is possible in the situation of treating the principle of sustainable development in terms of Dworkin's "policies" and allows its application not only at the level of European Union law (primary and secondary), but also at the national legal orders of the European Union Member States.

**Keywords:** principle of sustainable development; public interest; individual interest; cultural heritage; law

#### 1. Introduction

At the basis of the formulation of the concept of sustainable development, which the principle of sustainable development derives from, was an issue of protection of the environment from broadly perceived human activity, including the activity contributing to economic development [1] (p. 81). The very idea of sustainable development can already be noticed in the views expressed by M. Sommeville [2]. Reflections over the human impact on the natural environment led to a formulation of a concept of sustainable development. A concept, which initially concerned mainly environmental issues, is of an open character, since sustainable development may refer to many aspects of social, political, economic and cultural life. The concept of sustainable development, and *sensu stricto* its elements, has been gradually implemented in the international, national and local law. Next, it turned into the principle of sustainable development formulated in various legal acts. However, for the implementation of the principle of sustainable development to be effective, it is necessary to introduce the appropriate legal instruments at the international, European Union, national and local level, especially in the branch of administrative law [3]. The cultural heritage protection law, and more narrowly perceived historic buildings protection law, derives

Citation: Kozień, A. The Principle of Sustainable Development as the Basis for Weighing the Public Interest and Individual Interest in the Scope of the Cultural Heritage Protection Law in the European Union. *Sustainability* 2021, *13*, 3985. https://doi.org/ 10.3390/su13073985

Academic Editor: Zachary A. Smith

Received: 17 February 2021 Accepted: 30 March 2021 Published: 2 April 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). from the administrative law. However, the cultural heritage protection requires taking into account on the legal level numerous branches of law, such as, for example, administrative, civil, criminal constitutional, international, European union law. Moreover, in professional literature, a separation of a branch of the cultural heritage protection law is postulated [4] (pp. 23–33). Protection of cultural heritage appears on the international, European Union, national and local level, and a form of historic heritage protection is subject to a process of continuous evolution [5] (pp. 219–243). Cultural heritage as a carrier of cultural identity is one of the key elements of the state and social life, so, therefore, its protection also has its public and legal dimension, especially that the preservation of heritage, and its transfer to future generations is included into the public best interest from the perspective of administrative law. At the same time, the designates (objects assigned to the given name) of cultural heritage, such as, for example, historic buildings or objects, may belong to various entities of public and private law. As a consequence, a dilemma may appear of weighing the public interest and individual interest in cases connected with designates of cultural heritage in situations in which there is a conflict between the right of ownership or the rights arising from contracts and the necessity of preservation of authenticity and integrity of designates of cultural heritage [6] (pp. 187–210), [7] (pp. 21–31). In the contemporary globalized world in which the state borders become blurred, the existence of sui generis clauses at the international level is necessary, which would allow for the solution of conflicts between public and individual interest. A principle of sustainable development, also in some legal acts and professional literature, is connected with cultural heritage issues [3,8], so, therefore, one should ponder whether the principle of sustainable development, as a principle present in numerous acts of international law, especially in the European Union law, could become the basis of weighing public interest and individual interest in issues connected with designates of cultural heritage.

The goal of this paper is an analysis of the possibility of using the principle of sustainable development derived from the concept of sustainable development to weigh public interest and individual interest in the scope of protection of cultural heritage in the European Union, as well as formulation of appropriate remarks de lege ferenda for the European Union legislator. As a preliminary research hypothesis, it was assumed that the principle of sustainable development is important and may be the basis for weighing the public and individual interest in the field of cultural heritage designations in the European Union. Cultural heritage designations, both movable and immovable, are often owned by bodies governed by public or private law. In this respect, these entities have a wide power over these designations, but its implementation may not necessarily coincide with the public interest, which is to preserve the designations of cultural heritage in the richness of their authenticity and integrity. As a consequence, it is often necessary to weigh the public and private interests in individual cases. Traditionally, this weighing may be based on the principle of proportionality, but it is not always effective to rely on it and, consequently, may lead to an ultimate advantage of one interest over the other. This problem, instead of the proportionality principle, could be solved using the principle of sustainable development. It is worth emphasizing that the concept itself, as well as the principle of sustainable development, has an interdisciplinary character, and the issues of protecting cultural heritage are also of a similar nature. As a consequence, the application of the principle of sustainable development as the basis for weighing public and individual interests would make it possible to find a certain compromise in specific situations and, to the greatest possible extent, enable the implementation of both public and individual interests. The application of the principle of sustainable development seems to be expedient also because it is present in the law of the European Union and, therefore, in all Member States of the European Union, which is important because nowadays the issue of protection of cultural heritage is often not international in nature only from the theoretical perspective (world or European cultural heritage), but also from a practical perspective, especially since the boundaries between the Member States of the European Union are becoming blurred, especially in the economic, social and cultural areas.

#### 2. Methods of Research

During the analyses, the scientific methods characteristic for legal sciences have been employed. In order to analyze the principle of sustainable development based on the concept of sustainable development, as well as the issues of weighing legal and public interest in reference to the protection of cultural heritage in the perspective of theory of law, as well as axiology of law, a theoretical and legal method was used. The theoretical and legal method in the context of this article is used to analyze legal institutions in the field of administrative law, civil law and European Union law on the basis of legal acts and literature on the subject and also enables the analysis of tools and bases for the operation of public and private law entities. The area of the theoretical-legal method also includes considerations based on legal logic by identifying a set of referents of given concepts, analyzing the range relations of given concepts, analyses of functions and features of given concepts, taking into account the categories of syntactic names and functors, semantics of names, theory of definition, as well as the theory of relations between particular designates and notions. In order to analyze legal acts, which are relevant from the perspective of the subject of research, the formal and dogmatic method was used. The formal-dogmatic method in relation to this article is based on the analysis of legal acts and, as an auxiliary, legal doctrine. As far as the analysis of legal acts is concerned, it is based on normative material, as well as on the reconstruction of legal norms from various legal provisions. These legal provisions, which are in different hierarchies and degrees of interdependence, are then verified by applying conflict of laws rules (hierarchy, chronology and detail). Moreover, the normative material itself, after reconstruction, is analyzed and interpreted in the process of law interpretation, first at the semantic level (linguistic analysis), then at the system level (reference and verification of linguistic analysis in the context of a given branch or legal system) and, finally, the purposeful level (reference previous considerations for the purposes of a given regulation). The goal of the analysis of the subject of research in legal acts previously in effect, the historical and legal method was used. The historical-legal method in the context of this article made it possible to analyze previous legal regulations from the perspective of definitive issues, as well as the location of given concepts in the legal system in the past legal acts. This was of particular importance in the context of the analysis of the primary law of the European Union, which is no longer in force, but was the basis for the creation and shape of the current primary law of the European Union, also taking into account the issues of sustainable development and cultural heritage. In the scope of interdisciplinary analysis of professional literature from the scope of management science, economic science as well as legal sciences, the method of literary criticism was used. The analyses have been based on interdisciplinary professional literature as well as legal acts-international European Union law and national sources of the generally applicable law. The method of literary criticism was applied taking into account the selection of basic literature, its query, selection, preparation of the publication database, analysis of the content and, on this basis, the preparation of a report, which is the basis for the theoretical considerations of this article, as well as the formulated conclusions. The applied methods and legal conclusions are the means enabling for the implementation of the scientific output of economic science and management science in the scope of the concept of sustainable development into law, and specifically the cultural heritage protection law.

The research methodology of the article was based on the following stages:

- 1. Identification of the research problem and research gap.
- 2. Formulating the aim of the research and the initial research hypothesis.
- 3. Identification of the state of knowledge through the analysis of the literature on the subject and legal acts (normative material) of the ius cogens, ius dispositivum and soft law nature.
- 4. Collecting text and normative data.
- 5. Data analysis, coding and interpretation.
- 6. Development of research results.
- 7. Formulating conclusions.

#### 3. Results

#### 3.1. Assumptions of a Concept of Sustainable Development

A concept of sustainable development as a response to the effects of human activity in the natural environment, but also numerous crises of the political, military, economic, social and environmental character, appeared in the 20th century, although its ideas had already been visible already in the 19th century. Most frequently, referring to the concept of sustainable development, the issue of development is indicated; however, from a historical perspective, the key issue was a concept of conservation presented by M. Sommerville, but also by G.P. Marsch and G. Pinchot [9] (pp. 2–3), postulating the preservation of the environment in the best possible condition for future generations [1] (p. 83). G. Pinchot acknowledges the necessity of preserving the environment and its benefits for future generations comparing this activity to the operations of an enterprise [10] (p. 136), which is connected strictly with the intergenerational justice principle [11] and the necessity of preservation and transferring the environment and its benefits in the best possible condition for future generations. M.C. Cordonier Segger notices that a concept of sustainable development should be treated as a compromise to preserve the natural environment for future generations [12] (p. 2). The significant economic growth impacts the environment as well as its natural surroundings, especially since some of them are non-renewable [13] (pp. 12–13). Economy in principle had been tackling the issue of economic benefits; however, currently, it is necessary to present responsible economic approach, which will also take into consideration ecological benefits [13] (p. 13), as well as social issues. In economy the problems of long-term and sustainable economic growth is analyzed in the perspective of the following theories of development:

- 1. Neoclassical theory of growth, subsequently connected with the liberal growth policy and the economic development.
- 2. New theory of growth (endogenic growth).
- 3. Theory of real business cycle.
- 4. Theory of sustainable development [13] (pp. 13–14), [14].

However, in principle, only in the sustainable development theory the condition of sustainability of growth is the sustainable development [13] (p. 14). Additionally, the fact must be emphasized that the concept of sustainable development is not only the economic theory but the ethical and philosophical theory as well [13] (p. 14); therefore, it has become possible to implement it into the generally applicable law. The interdisciplinary character of the concept of sustainable development is well described in the approach presented by D. Pearce and R.K. Turner who had stated that sustainable development "means the maximization of the net profits from the economic development, at the same time protecting and securing the reproduction of usability and quality of natural resources in a long term. Economic development of other elements of social wellbeing. It must also include the necessary structural changes in economy, as well as in the entire society" [15], [13] (p. 17). W. Pearce, E. Barbier and A. Markandya think that the sustainable development means the implementation of defined goals, which are socially desirable, among which one must mention:

- 1. growth of the actual *per capita* income;
- 2. improvement of the health condition of the society;
- 3. fair access to natural resources;
- 4. improvement of the level of education." [13] (p. 17), [16] (p. 2)

T. Tietenberg acknowledges the formation of an individual's wellbeing in time as a starting point for sustainable development [17].

R. Carson noticed the relationship of economic development with the natural environment and social wellbeing [18]. One can notice, though, that in a concept of sustainable development one may differentiate three basic aspects, complementary towards one another, so, therefore, the environmental, social and economic aspects. They are of a character of primary aspects, which form a concept; however, the elements of a concept began to be gradually implemented to the generally applicable law, and next, a principle of sustainable development was formed, which is present in some acts of international, European Union and the national law. The principle is of a general clause character, so it may refer to various aspects of social life; therefore, one should indicate secondary aspects, arising directly from the principle of sustainable development, and indirectly from the concept of sustainable development, which may include, inter alia, the political, legal, cultural aspects, related to the rights and freedoms of a human being and a citizen, and also safety. In the perspective of reflections over the weighing of the individual interest and the public interest, it is worth noting that taking into consideration the social aspect of economic development, which is compliant with sustainable development as an evolution could be perceived—from the wellbeing of an individual towards the social wellbeing, as well as an attempt to find a compromise between these interests [13] (p. 14). As a consequence, it should be stated that the necessity of prioritizing the social best interests should not be imposed immediately, but development of an individual and his/her wellbeing should be made possible, so that next, an individual should be gradually persuaded to reject one's own egoism and turn towards the social milieu—the communities. In professional literature, it is indicated that the connection of social interests and individual interests may be a concept of a "visible hand", which according to A. Chandler, should be based on undertaking actions at various decisive levels in order to harmonize individual interest with social interest [19].

The above review of various approaches to the concept of sustainable development, and also the presentation of basic assumptions, does not have an exhaustive character, since due to its capacity and a broad scope of the notion, sustainable development in professional literature, one can notice various approaches and interpretations, especially in the concept of economic aspect and social aspect of the concept of sustainable development [1,9,12,20–23]. Analyzing the views and interpretations of the concept of sustainable development in the professional literature, one may notice that it is not homogenous, and even in some places, it is internally contradictory, which may particularly hinder its implementation into the generally applicable law, which should be internally cohesive [24].

## 3.2. A Concept of Sustainable Development as the Basis of Formulation of the Principle of Sustainable Development

The problem regarding the relations of human activity in relation to the environment was a subject of various analyses at the scientific, organizational, political and legal level in the 20th century. In 1968, during the first Intergovernmental Conference of UNESCO Scientific Experts, during which an issue was dealt with of interconnections between the environment and the development, an interdisciplinary program (MAB) [25] was formed. One of the conferences, which also significantly impacted the further development of sustainable development in the political, social and legal perspective, was the United Nations Conference on the Human Environment (the so-called Stockholm Conference) in 1972, which tackled the issue of environmental protection and its relationship in reference to economic and social aspects [1] (p. 88). It is worth quoting Principle 8 of the Declaration of the United Nations Conference in the Human Environment: "Economic and social development is essential for ensuring a favourable living and working environment for man and for creating conditions on earth that are necessary for the improvement of the quality of life" [26]. In the same year, the Club of Rome published "The Limits to Growth" [22], undertaking the issue of sustainable development. The issue of sustainable development, especially in a perspective of environmental protection, was subsequently referred to in the Brandt Report titled: "North: South—A Programme for Survival", written within the scope of the Independent Commission on International Development Issues [1]. The requirements of the environmental protection were undertaken in the World Conservation Strategy [27] document of 1980, where for the first time in an official document a notion of sustainable development was used [28] (p. 28). A key document concerning sustainable development "Our Common Future" [29], i.e., the so-called Brundtland Report of 1987, contains the

following definition of sustainable development: "In essence, sustainable development is a process of change in which the exploitation of resources, the direction of investments, the orientation of technological development; and institutional change are all in harmony and enhance both current and future potential to meet human needs and aspirations" [29]. The definition of sustainable development in the Brundtland Report has a character of a program [24], which from the legal perspective may create difficulties connected with inaccuracy of a notion and heterogeneity of a concept, and which follows the principles of sustainable development; however, from the other side, the flexible definition of program gives the possibility of adjustment to the dynamically varying social-economic-political situation in the world, and also, it enables the vast application of a concept in various fields of social life. Further development of the concept was continued during the Earth Summit in Rio de Janeiro, which took place in 1992. Its important output from the legal perspective is the Declaration Rio and the Agenda 21 [30,31]. Another important Earth Summit from the perspective of the concept of sustainable development formation was the Earth Summit in Johannesburg in 2002, during which the Johannesburg Declaration on Sustainable Development was adopted [32]. Additionally, a fact needs emphasizing that many international conventions, especially in the scope of environmental protection, are based in a defined measure on the concept of sustainable development. At the same time, one of the views in the doctrine of international law indicates that the concept of sustainable development is too general to be treated as a principle or a rule of common law [33] (p. 78); however, another view in this matter—just to the contrary—emphasizes the normativity of the concept and also allows us to declare that currently a concept of sustainable development law already exists [34]. It is worth emphasizing that the concept of sustainable development increasingly often has impact on the international but also on the European Union, national or local legislation, especially in the branches of public law, including administrative law [3].

In the European Union law, both primary and secondary, one may perceive a certain impact of the concept of sustainable development, especially in the scope of the protection of the environment. At the level of the European Union law, sustainable development is of a character of a political system founding rule. In the doctrine of European Union law, it is indicated that the European legal system does not consist only from the written law norms [35], but it is supplemented by general principles of law, which are a part of the legal order, which also finds a confirmation in the jurisdiction of the European Court of Justice. These principles from the perspective of the normative character are imposed by appropriate legal decisions of creators of policies, as well as the court jurisdiction [35]. J. Bengoextea, in the classification of the principles constituting the European Union law, distinguished as the last ones such principles, which allow it to measure up to the arising new social problems among which, next to such notions as subsidiarity, environmental protection, proportionality, a principle of sustainable development appears [36] (p. 100–110). It is worth noting, though, that in this division the principles and values have been combined [35] (p. 106), [37] (p. 29). J. de Cendra Larragan distinguishes principles from values. In this respect, the principles should be treated as directing the actions, which they are a source of onto a goal, and also, they provide guidelines for a legislator and allow the courts to issue verdicts based on them in a situation of any gaps appearing in the law, whereas the values are the goal in themselves [35] (p. 106–107). In principle in the primary law of the European Union, one may notice the literal approach to the issue of sustainable growth (not yet development) only in the Treaty of Maastricht of 1992, through which, inter alia, the amendment of art. 2 was introduced of the Treaty establishing the European Economic Community, which formed the following wording: "The Community shall have as its task, by establishing a common market and an economic and monetary union and by implementing the common policies or activities referred to in Articles 3 and 3a, to promote throughout the Community a harmonious and balanced development of economic activities, sustainable and non-inflationary growth respecting the environment, a high degree of convergence of economic performance, a high level of employment and of

social protection, the raising of the standard of living and quality of life, and economic and social cohesion and solidarity among Member States" [38]. Sustainable development was already recorded as a political system founding rule in art. 2 of the Treaty of Amsterdam of 1997 in the following wording: "The Community shall have as its task, by establishing a common market and an economic and monetary union and by implementing common policies or activities referred to in Articles 3 and 3a, to promote throughout the Community a harmonious, balanced and sustainable development of economic activities, a high level of employment and of social protection, equality between men and women, sustainable and non-inflationary growth, a high degree of competitiveness and convergence of economic performance, a high level of protection and improvement of the quality of the environment, the raising of the standard of living and quality of life, and economic and social cohesion and solidarity among Member States" [39]. The European Union legislator changed the narrower notion of sustainable growth to the wider scope of a notion of sustainable development.

In the currently applicable Treaty of Lisbon, which consists of the Treaty on European Union (TEU) and the Treaty on the Functioning of the European Union (TFUE) [40], the notion of sustainable development appears mainly in reference to environmental issues, but also to social and economic issues. In the Treaty on the European Union the principle of sustainable development treated as a political system founding rule [1] (p. 39), [41] (p. 282) appears in the Preamble of the Treaty on European Union: "Determined to promote economic and social progress for their peoples, taking into account the principle of sustainable development and within the context of the accomplishment of the internal market and of reinforced cohesion and environmental protection, and to implement policies ensuring that advances in economic integration are accompanied by parallel progress in other fields" [40]. One could analyze, though, whether from the text of the Preamble one could conclude on the existence of principles, especially that the Preamble should rather indicate the directions of interpretation of a given act of law. However, regardless of these reservations of a normative character, one must emphasize that the European Union legislator expanded the use of a principle of sustainable development basically to many fields of social life. On the grounds of the Treaty on the European Union the sustainable development appears in:

Art. 3 section 3 paragraph 1 TEU among the European Union goals in the social, economic and social context: "The Union shall establish an internal market. It shall work for the sustainable development of Europe based on balanced economic growth and price stability, a highly competitive social market economy, aiming at full employment and social progress, and a high level of protection and improvement of the quality of the environment. It shall promote scientific and technological advance" [40].

Art. 3 section 5 TEU, in which the European Union legislator decides that the goal of the European Union is the contribution to "the sustainable development of the Earth" [40].

Art. 21 section 2 letter d TEU and art. 21 section 2 letter f [40], which in principle transfer the European Union laws onto the level of its international cooperation with other states [24].

It is worth emphasizing a consequence of the European Union legislator who adopts sustainable development among the goals, not the values of the European Union, which from the axiological perspective and the nature of a concept of sustainable development is a correct approach [24]. The sustainable development was also undertaken on the grounds of a Treaty on the Functioning of the European Union in the following legal regulations:

Art. 5 TFEU—a principle of subsidiarity, which is strictly related to the principle of sustainable development.

Art. 11 TFEU, which states that "Environmental protection requirements must be integrated into the definition and implementation of the Union's policies and activities, in particular with a view to promoting sustainable development" [40]. Art. 11 can be found in Title II containing the EU principles. Analyzing art. 11 TFEU one can indicate, firstly, the principle of integration and, secondly, the principle of sustainable development. At the

same time, it must be pointed out that in principle they refer to environmental issues, so, therefore, they narrow down the principle of sustainable development.

Title XX named: "Environment", which is based on the concept of sustainable development implements it to environmental issues. It is especially perceivable in the scope of the goals of the European Union environmental policy—in art. 191 section 1 TFEU: "preserving, protecting and improving the quality of the environment, protecting human health, prudent and rational utilisation of natural resources, promoting measures at international level to deal with regional or worldwide environmental problems, and in particular combating climate change" [40], as well as the principles of the European Union environmental policy (especially art. 191 section 3 TFEU).

Art. 309 TFEU in principio, constitutes that "The task of the European Investment Bank shall be to contribute, by having recourse to the capital market and utilising its own resources, to the balanced and steady development of the internal market in the interest of the Union" [40], which can be perceived as sensu largo a reference to the economic aspect of a concept of sustainable development in conjunction with art. 3 TEU.

Declaration on Article 126 of the Treaty on the Functioning of the European Union, paragraph 5, constitutes that "The Union aims at achieving balanced economic growth and price stability. Economic and budgetary policies thus need to set the right priorities towards economic reforms, innovation, competitiveness and strengthening of private investment and consumption in phases of weak economic growth. This should be reflected in the orientations of budgetary decisions at the national and Union level in particular through restructuring of public revenue and expenditure while respecting budgetary discipline in accordance with the Treaties and the Stability and Growth Pact" [40]. In this Declaration one can also notice a reference to the economic growth", which is derived from the concept of sustainable development; however, this Declaration refers to the issues connected with the budget deficit, so, therefore, this is a very narrowly defined economic perception.

In the primary and secondary law of the European Union, there is no definition of sustainable development, so, therefore, pursuant to the views of L. Kramer and D. Pyć [41,42], it should be stated that in the European Union law a definition of sustainable development is identical with the definition existing on the basis of the international law and, therefore, adopted in the Brundtland Report, pursuant to which "Humanity has the ability to make development sustainable to ensure that it meets the needs of the present without compromising the ability of future generations to meet their own needs. The concept of sustainable development does imply limits—not absolute limits but limitations imposed by the present state of technology and social organization on environmental resources and by the ability of the biosphere to absorb the effects of human activities. But technology and social organization can be both managed and improved to make way for a new era of economic growth" [29]. A problematic issue is the implementation of the heterogenous concept of sustainable development onto the legal grounds, as well as its effectiveness [43]. In the professional literature, inter alia, the following approaches to sustainable development in the European Union law are indicated:

A concept of sustainable development as the political system founding rule of the European Union [41] (p. 282).

Art. 191 of the TFEU, undertaking an environmental aspect of a concept of sustainable development is the treaty basis to issue the acts of the EU law in a situation when the main goal of the given act of law is environmental protection [44] (pp. 10–11).

Principles that may be interpreted based on art. 191 TFEU are the guidelines in the scope of the environmental protection policy [42].

The principle may be transformed into law in a situation when it is contained in the legal regulations; however, the policies cannot become the generally applicable law, and a concept of sustainable development should be treated as a policy [45] (p. 13).

Sustainable development is legally effective in European Union law [46].

Legal significance of a concept may be referred only to art. 11 TFEU (the environmental aspect) [47].

J.H. Jans indicates that the sustainable development is a guideline connected with political activity, and not the normative character [48] (p. 496).

A concept of sustainable development should be treated as an idea or a policy [49] (p. 19). The analysis concerns European Union law, but it is worth noting that a principle of sustainable development also appears in national and local legal acts of various countries. As an example, it can be pointed out that in the Constitution of the Republic of Poland, the constitutional legislator in art. 5 decides that "The Republic of Poland shall safeguard the independence and integrity of its territory and ensure the freedoms and rights of persons and citizens, the security of the citizens, safeguard the national heritage and shall ensure the protection of the natural environment pursuant to the principles of sustainable development" [50]. On the grounds of this article, a dispute arose in the doctrine of law, whether the principle of sustainable development clearly pointed out by the Polish constitutional legislator refers only to the environmental protection or the issue of territorial sovereignty, human freedoms and rights, security of citizens and national heritage. In case of broad interpretation, it should be pointed out that the Polish constitutional legislator not only holistically implemented the principle of sustainable development, but also expanded its conceptual range as well, referring it also to the issue of national heritage protection. I. Niżnik-Dobosz and P. Dobosz state that "The principle of sustainable development belongs to the so-called programmatic norms of the Polish Constitution which—under the theory of the national law—outline only the general direction of operation for the state bodies, at the same time constituting the legal basis for formulation of specific norms, defining possibly unambiguously the addressee's behaviour" [8]. One should lean towards the voices in professional literature, which point out to the broad implementation of the principle of sustainable development to the Polish law and, therefore, also to the issues connected with the protection of cultural heritage [8]. It can also be confirmed in the case law of the Polish Constitutional Tribunal, which had stated that "the principle of sustainable development includes not only the environmental protection or development of spatial planning, but also sufficient care for social and civilizational development, connected with the necessity of building the required infrastructure, necessary for human life and life of particular communities-taking into account the civilizational needs. Therefore, the idea of sustainable development includes a need to consider various constitutional norms and their appropriate balancing out" [51]. I Niżnik-Dobosz and P. Dobosz, developing the analysis of the principle of sustainable development in reference to the cultural heritage pursuant to the Polish Constitution, distinguish three meanings (qualifications) of the principle of sustainable development: "firstly-the sustainable, harmonious development of a given sphere of social relations: business, economic, spatial, regional, secondly—is to be at the same time cohesive, friendly, corresponding to the principles of environmental protection. It is also possible to deliberate upon triple qualification, according to which the idea is about the sustainable development of harmoniously developing spheres of social relationships referred to above, which are compliant at the same time with the principles, notions and institutions of the Environmental Protection Law" [8]. I. Niżnik-Dobosz and P. Dobosz deem the adoption of double qualification as appropriate, which eventually, however, leads them to adopt the triple qualification as well [8].

#### 3.3. Cultural Heritage as an Element of a Concept of Sustainable Development

Prima facie, it would seem that the protection of cultural heritage does not become an inherent part of the concept of sustainable development; however, the very concept, as it had already been presented above, consists of three basic cornerstones: economic, social and ecological [13] (pp. 16–17), [52] (p. 148). The environmental aspect seems to be the most developed one. The social aspect of the concept has a broad conceptual range, so, therefore, its scope may contain an issue of cultural heritage protection. It has a particular justification from a perspective of the principle of intergenerational justice, which stands at the base
of social and environmental aspects of the concept of sustainable development. Cultural heritage is also frequently connected to the environmental milieu, in which it appears. The concept of cultural heritage is defined differently in different scientific disciplines. For example, one can mention the definition of J. Pruszyński, who considered cultural heritage as "the stock of movable and immovable things along with the related spiritual values, historical and moral phenomena, considered worthy of protection of the law for the good of society and its development and passing on to the next generations, due to understandable and accepted historical values, patriotic, religious, scientific and artistic, important for the identity and continuity of political, social and cultural development, proving the truths and commemorating historical events, cultivating a sense of beauty and civilization community" [53] (p. 50), and K. Zeilder treats cultural heritage as "the entire material and spiritual heritage of a given social group" [54] (p. 26). Pursuant to the acts of international law [55,56] and the professional literature, tangible and intangible cultural heritage can be distinguished; however, in principle the connection of tangible and intangible heritage creates a certain entirety and the complete dimension of cultural heritage. According to Art. 1 Convention Concerning the Protection of the World's Cultural and Natural Heritage [55], the concept of cultural heritage (especially in relation to tangible cultural heritage) covers:

- "monuments: architectural works, works of monumental sculpture and painting, elements or structures of an archaeological nature, inscriptions, cave dwellings and combinations of features, which are of outstanding universal value from the point of view of history, art or science;
- groups of buildings: groups of separate or connected buildings which, because of their architecture, their homogeneity or their place in the landscape, are of outstanding universal value from the point of view of history, art or science;
- sites: works of man or the combined works of nature and man, and areas including archaeological sites which are of outstanding universal value from the historical, aesthetic, ethnological or anthropological point of view" [55]. On the other hand, when looking for a legal definition of intangible cultural heritage, art. 2 clause 1 sentence 1, which states that: "The »intangible cultural heritage« means the practices, representations, expressions, knowledge, skills—as well as the instruments, objects, artefacts and cultural spaces associated therewith—that communities, groups and, in some cases, individuals recognize as part of their cultural heritage" [56].

Moreover, a fact of strong connection of heritage with religion and values needs emphasizing. In Europe, heritage has a religious dimension [57], and it is connected with Christianity and its values. In reference to the reflections over the cultural heritage in the context of the principle of sustainable development it is worth quoting J. Purchla's view that "Relationship between the past and the future is not limited today to the issue of historic buildings and their protection. A notion which has been enjoying spectacular career lately, is the cultural heritage (cultural heritage). More and more often it replaces the classical notion of a historic building. Here one should absolutely state that although the historic building belongs to the past, the heritage serves contemporary goals, whereas this heritage is not limited only to tangible cultural assets, but also our memory and identity. Cultural heritage is a process with its own dynamics. It reflects both the relationship of the society to the world of values, but also a process of reinterpretation of values in itself. Herein lies also the growing significance of cultural heritage. Because the heritage belongs to all of us, and access to it is one of the most fundamental human rights. Therefore, the heritage always has the human dimension. From this fact the key significance of social capital stems from not only for the dynamic process of continuous creation and reinterpretation of heritage, but also for its effective defense" [58] (p. 44). Therefore, one can state that the heritage has dynamic character when it is a certain process. The dynamics of changes and also a definite process of changes is also entered into the concept of sustainable development, and implementation of the principle of sustainable development allows for flexible adjustment to dynamically changing conditions of the social, economic and environmental milieu, respectively. It is also worth noting that contemporary literature is concerned with the issue of sustainability in various contexts related to cultural heritage [59–61]. Connection of a principle of sustainable development with the protection of cultural heritage also allows for a multi-aspect approach to the issue of protection of cultural heritage, which should also contribute to the increasing the effectiveness of cultural heritage management, especially that in contemporary times the heritage must be appropriately managed [62,63]. The only problem can be the fact that the concept of sustainable development has not been holistically implemented to any legal order, which may cause a decrease in the effectiveness of the principle of sustainable development [24,43]. On the other side, however, on account of heterogeneity of a concept at the normative level, it is not possible to implement it holistically, so, therefore, perhaps, it should be treated as a sui generis programmatic norm or even the policy as well, which although, from the international perspective may be easier to introduce, and which also follows then to use in defining specified standards of policy in a given scope.

As a consequence, it can be noted that cultural heritage with all its material and nonmaterial wealth, as well as taking into account the principle of intergenerational justice, which is the basis of the concept of sustainable development, as well as the protection of cultural heritage, is part of the concept of sustainable development, especially in the social aspect, but also environmental. In addition, the protection of cultural heritage should be subject to the processes of sustainable development and is part of them through the necessity to preserve the cultural heritage of the past and the present generation for future generations.

# 3.4. Weighing the Public Interest and Individual Interest in the Scope of Protection of Cultural Heritage

Protection of cultural heritage requires from the legislator the introduction of certain restrictions of individual freedoms. Usually, these restrictions pertain mainly to the issue of the right of ownership, as well as the freedom of business activity. Protection of cultural heritage is connected with public interest, so, therefore, the necessity of preservation and transfer to future generations of the heritage of present and future generations, whereas the issues connected with individual freedoms, so, therefore, inter alia, the right of ownership and the freedom of business activity, relate to an individual's best interest. Public (social) interest is a diverse, variable and connected with a specific factual status [64] (p. 137). Evaluating the justifiability of protection of public interest in a given case one should consider numerous factors such as, inter alia, the ratio legis of a given legal act, its objectives and tasks, as well as the general principles of law [64] (p. 137). In the doctrine of administrative law, sometimes the public interest is identified with public welfare [65] (p. 106), which may include, inter alia, cultural heritage. Public interest is not the sum of private interests, either [65] (p. 107), but something above the private interest, because it includes various entities of public and private law. Public interest as an undetermined notion should be reviewed in a perspective of the values of the generally applicable law (especially legal rights, which are given the highest position in the hierarchy) [65] (p. 109). A notion of the public interest is flexible and is of dynamic character, allowing for adjustment to the changing political, social and economic conditions, with the concurrent preservation of legal foundations based on legal values and principles in a given legal order. Public interest on one side allows for effective protection of an individual's freedoms, which lies in the public interest, but at the same time, this notion often stands at the basis of restriction of an individual's freedom on account of the public interest [65] (p. 110), [66] (p. 34). Individual interest is the interest of an individual person, perceived very broadly, because also in the context of the factual interest [64] (p. 138). It is also worth emphasizing that legislators sometimes use terms qualifying this interest, such as for example a principle of equity, which indicates that the equitable interest of an individual cannot be illegal and principle of social coexistence [64] (p. 138).

Taking into account the public and individual interest, two underdefined general clauses, one should consider a situation in which the conflict of both of these interests

occurs [67] (p. 39–40). Therefore, one can present the following possibilities of solution to this conflict [67] (p. 40–41):

- a. In a situation of conflict, one should firstly implement the public interest.
- b. One should protect individual interest as far as the border of conflict with the public interest.
- c. Both interests should be balanced out and, bestowing the principle of equity to the individual interest, indicate the lack of protection of any subjective individual interest.
- d. The theory of sustainable development [13] (pp. 13–14).

The first of the possibilities is a characteristic for non-democratic states, for which the overriding goal is the public interest, even though de facto, it was of an individual character (public interest is equal to the interest of persons exercising power and authority). The other possibility de facto also in principle prejudges on the primacy of the public interest. The third presented possibility assumes balancing out the interests and restricts the subjectivism of an individual interest. It seems to be the most matching to balancing out the public and individual interest in the scope of the cultural heritage protection. Balancing out is also entered in a certain sense into the principle of sustainable development. Therefore, it is necessary to review this possibility from the perspective of the principle of sustainable development. As a consequence, reviewing a specific case using the principle of sustainable development, one should first reflect on the basis of analysis of various factors (legal, social, economic, environmental, cultural) upon the necessity of protection of a given designate of cultural heritage and, next, considering the scope of action that an owner of a given designate of cultural heritage wishes to undertake towards it, using the principle of balancing out the interests, make a decision regarding how much the protection of public interest is necessary and to what extent. Balancing out the interests based on the principle of sustainable development will also allow us to find a compromise between the protection of public interest and fulfillment of the individual interest by, for example, indicating certain elements, which have to be preserved, and allowing for modification/adjustment of elements less significant from the perspective of a given cultural asset in order to allow for exercising, for example, the right of ownership.

#### 4. Discussion

The principle of sustainable development, expressed in European Union law, is of interdisciplinary character and may be used in order to solve conflicts of legal interests supportively towards the principle of proportionality, which is of the primary and more general character. Although sustainable development is identified in majority with environmental issues, one should clearly emphasize that the assumptions of the concept significantly extend beyond the environmental protection, because they refer to broadly perceived social, economic and environmental aspects, and moreover, they are strongly connected with the principle of intergenerational justice, which also stands at the foundation of the cultural heritage protection. It is also worth noting that the concept of sustainable development derives also from the concept of conservation, a notion that is particularly related to the conservation of historic buildings. Protection of cultural heritage is entered into the social aspect of a concept of sustainable development. It should also be noted that the protection of cultural heritage may be also connected with conservation of greenery, entering into the wholeness and integrity of a given designate of cultural heritage. In a particular way, it concerns historic buildings, and especially the palace and park complexes, and protection of a given landscape is of significance to cultural heritage. It can be confirmed, though, that in specific cases the implementation of the principle of sustainable development in reference to the conservation of designates of cultural heritage is not even proper, but necessary indeed, because it also refers to the environmental aspect of the subject principle. Various approaches in professional literature presented in item 3.2 of this article to the principle of sustainable development at the level of the European Union law arise from heterogeneity, multi-dimensionalism and multilevel character of the concept of sustainable development, being the foundation of forming the principle of sustainable development. Vast interpretational possibilities of this concept may lead to the danger of a lack of cohesiveness of the principle of sustainable development at the normative level, and this is the cohesiveness and clarity of law, which is necessary from the perspective not only of the very nature of law, but of the effectiveness of law as well. The European Union legislator, perceiving these problems, entered sustainable development into the European Union goals, and not values. The author thinks that at the level of European Union law, sustainable development cannot be treated as a political system founding rule and also as a legally binding one; the differentiation of using the principle of sustainable development should not be applied depending on which aspect it refers to. Additionally, it is difficult to treat the sustainable development only as an European Union/international idea or policy. As a consequence, in the author's opinion, it is worth invoking R. Dworkin's [68] concept, and it should be declared that the principle of sustainable development cannot be a legal rule but a standard, and specifically one of the "policies", outlining general goals, in this case referring to social, economic and environmental issues. One should emphasize, though, that Dworkin's "policies" are a part of the legal system, and they are not a sole result of the current international, European Union, national or regional policy. Taking the above into consideration, it seems fair to state that the principle of sustainable development should be treated as the principle outlining the goal, and also indicating the directions of interpretation of given legal provisions (legal regulations on the normative contents). As a consequence, then, a given legal interest should be interpreted in the context of the implementation of sustainable development. The principle of sustainable development should also be used in a situation of conflicts of common good and legal interests. In case of a conflict of common goods, it may be used supportively next to the principle of proportionality; whereas, in case of a conflict of interests within the environmental, economic or social aspects, the principle of sustainable development may fulfil a function of weighing the individual and public interest in a situation of their conflict. The linking of sustainable development with the individual and public (social) interest, especially in the context of harmonizing the individual interest with the social interest, is also perceivable in the professional literature [19], which had already been indicated in the course of reasoning contained in item 3.1. The concept of sustainable development is interdisciplinary, just like the approach to cultural heritage should be of such nature. Paying attention to the fact that cultural heritage is part of the concept, and thus the principle of sustainable development, should make it possible to look at the issue of cultural heritage protection not only from a legal or conservation perspective, but also from a social, environmental and even economic perspective. In principle, each person is responsible for some designations of cultural heritage, which are either in the public space, and then naturally belong to the entire society, or belong to individual bodies of public or private law. The necessity to weigh the public and private interest in the context of cultural heritage may, therefore, appear at any time, e.g., during construction, renovation, extension, reconstruction, sale, rental, maintenance and running a business. From the perspective of the protection of cultural heritage, it is also crucial to protect the authenticity and integrity of the designates of this heritage [6,7]. A holistic view of the issues of heritage protection shows many similarities in environmental protection; therefore, the choice of the principle of sustainable development seems to be purposeful. The above analyses have shown that the principle of sustainable development can be the basis for weighing public and individual interests in matters related to cultural heritage at the level of European Union law; however, the consequences of such an approach should also be indicated. So far, the principle of proportionality has been largely based on the necessity to choose between the individual interests of the owner of monuments and their expectations, and the public interest of the state or society and their expectations regarding the protection of a given heritage designate. However, the principle of proportionality is general and fundamental. The principle of sustainable development would allow for a more detailed and systematic approach to this type of issues and would also allow for an interdisciplinary approach, taking into account, apart from social issues, also environmental and economic issues. In addition, the principle of sustainable development in terms of Dworkin's "policies" would

allow to provide certain general rules and guidelines in such situations and would indicate the directions of solving this type of issues so as to reconcile the public and individual interests as much as possible. In addition, the principle of sustainable development occurs throughout the European Union and, therefore, would be a common determinant and direction for actions of public administration bodies and courts in this type of cases and, therefore, would enable more effective protection of cultural heritage and ensure uniformity and certainty of legal transactions in the European Union. In consequence, one can state then that the principle of sustainable development may be used in reference to the cultural heritage protection at the level of the European Union law, because it becomes an inherent part of the social aspect of the concept of sustainable development, which the principle of sustainable development is based on. Moreover, the principle of sustainable development may be the basis for weighing the public and individual interest in reference to the cultural heritage protection also at the level of the European Union, which confirms the adopted research hypothesis as well.

#### 5. Conclusions

In summary, one should state that the principle of sustainable development, present in the European Union law, deriving from the broad, interdisciplinary, multi-aspect and multi-level concept of sustainable development, may be used in reference to the broadly perceived problems of cultural heritage protection in its social aspect, especially taking into consideration the principle of intergenerational fairness. Moreover, the principle of sustainable development treated in the categories of Dworkin's "policies", being a part of the European Union legal system may be used as the foundation of weighing public interest and individual interest in the scope of protection of the cultural heritage, enabling more effective exercising of law in specific situations. Using this principle in this context should include not only the level of the European Union law (primary and secondary), but also the level of national legal frameworks of the European Union Member States, which rather should not be a problematic issue, especially that many countries in a specific scope use the concept of sustainable development—or more specifically—the principle of sustainable development in their legal framework. Ultimately, the application of the principle of sustainable development to weighing public and private interests in matters relating to cultural heritage would contribute to increasing the effectiveness of protection, uniformity and certainty of legal transactions, as well as cooperation in the European Union, as well as public trust in state and EU bodies.

**Funding:** This research was funded from the budget for science in 2019–2022, as a research project under the "Diamond Grant" program, project No. DI2018 018348.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** The data is based on the cited literature and legal acts and is available upon request.

**Conflicts of Interest:** The author declares no conflict of interest. The funders had no role in the design of the study; in the collection, analyses or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

#### References

- Kielin-Maziarz, J. Koncepcja Zrównoważonego Rozwoju w Prawie Unii Europejskiej; Wydawnictwo KUL: Lublin, Poland, 2013; pp. 35–48.
- 2. Sommerville, M. Psyhical Geography; John Murray: London, UK, 1862.
- Skrzydło-Niżnik, I.; Dobosz, P. The Principle of Sustainable Develepment in the Light of the Polish Constitution and General Principles of Administrative Law. In Facing the European Union—Challenges for Polish Law. *Archivum Iuridicium Cracoviense* 2004, 35–36, 59–81.
- 4. Zeidler, K. (Ed.) Prawo ochrony dziedzictwa kultury jako nowa gałąź prawa. In *Prawo Ochrony Zabytków;* Wolters Kluwer: Warszawa–Gdańsk, Poland, 2014.

- 5. Dobosz, P. Ewoluujące prawne formy ochrony zabytków w Polsce. In *Prawo Ochrony Zabytków;* Zeidler, K., Ed.; Wolters Kluwer: Warszawa–Gdańsk, Poland, 2014.
- 6. Kozień, A. Autentyczność i integralność dóbr kultury w świetle źródeł prawa powszechnie obowiązującego w Polsce. Zeszyty Naukowe Towarzystwa Doktorantów Uniwersytetu Jegiellońskiego—Nauki Społeczne **2018**, 21, 187–210.
- 7. Kozien, A. Protection of the authenticity and integrity of monuments as a determinant affecting the activities of entrepreneurs-the owners of monuments. In *Economic and Social Development (Book of Proceedings), 64th International Scientific Conference on Economic and Social Development;* Cingula, M., Misevic, P., Nedzhad, A., Eds.; Varazdin Development and Entrepreneurship Agency: Varazdin, Croatia, 2021.
- 8. Skrzydło-Niżnik, I.; Dobosz, P. Significance of the Principle of Sustainable Development for the Protection of National Heritage. *Int. Cult. Cent. Crac.* **2004**, *12*, 25–33.
- 9. Ramlogan, R. Sustainable Development: Towards Judicial Interpretation; Boston Martinus Nijhoff Publishers: Leiden, The Netherlands, 2011.
- 10. Pinchot, G. Conservation. In *Green History: A Reader in Environmental Literature, Philosophy and Politics;* Wall, D., Ed.; Routledge: London, UK, 1998.
- 11. Dobosz, P. Sprawiedliwość międzypokoleniowa w prawie administracyjnym. In *Sprawiedliwość i Zaufanie Do Władz Publicznych w Prawie Administracyjnym*; Stahl, M., Kasiński, M., Wlaźlak, K., Eds.; Lex a Wolters Kluwer Business: Warsaw, Poland, 2015.
- 12. Cordonier-Segger, M.C.; Khalfan, A. Sustainable Development Law: Principles, Practices, and Prospects; Oxford Scholarship Online: Oxford, UK, 2012.
- 13. Adamczyk, J.; Nitkiewicz, T. Programowanie Zrównoważonego Rozwoju Przedsiębiorstw; PWE: Warsaw, Poland, 2007; pp. 11-42.
- 14. Fiedor, B. Teoretyczne podstawy badania trwałości wzrostu gospodarczego. In *Sterowanie Ekorozwojem, Materiały Konferencyjne;* Poskrobko, B., Ed.; Białystok University of Technology: Białystok, Poland, 1990.
- 15. Pearce, D.; Turner, R.K. Economics of Natural Resources and Environment; Harvester Wheatsheaf: New York, NY, USA, 1990.
- 16. Pearce, W.; Barbier, E.; Markandya, A. Sustainable Development. Economics and the Environment in the Third World; Brookfield: Aldershot, UK, 1990.
- 17. Tietenberg, T. Environmental and Natural Resource Economics; Addison Wesley: Reading, PA, USA, 2003.
- 18. Carson, R. Silent Spring; Mariner Books: Boston, MA, USA, 2002.
- 19. Chandler, A.D. A Visible Hand. The Managerial Revolution in American Business; Belknap Press: Cambridge, UK, 1980.
- Baker, S. The European Union: Integration, Competition, Growth-And Sustainability. In Implementing Sustainable Development: Strategies and Initiatives in High Consumption Societes; Lafferty, W.M., Meadowcroft, J., Eds.; Oxford Scholarship Online: Oxford, UK, 2003.
- 21. Boyle, A.; Freestone, D. International Law and Sustainable Development: Past Achievements and Future Challenges; Oxford Scholarship Online: Oxford, UK, 2012.
- 22. Meadows, D.H.; Meadows, D.L.; Randers, J. The Limits to Growth; Potomac Associates-Universe Books: New York, NY, USA, 1972.
- 23. Jacobs, M. The Green Economy, Sustainable Development and the Politics of the Future; Pluto Press: London, UK, 1992.
- 24. Kozień, E.; Kozień, A. The sustainability development concept under the regulations in force of the Treaty on European Union and the Treaty on the Functioning of the European Union–legal and economical view. In *Economic and Social Development (Book of Proceedings), 35th International Scientific Conference on Economic and Social Development–"Sustainability from an Economic and Social Perspective"*; Ribeiro, H., Naletina, D., da Silva, A.L., Eds.; Varazdin Development and Entrepreneurship Agency: Varazdin, Croatia, 2018; pp. 402–412.
- 25. Program Man and Biosphere. 1968. Available online: http://www.unesco.pl/edukacja/dekada-edukacji-nt-zrownowazonego-rozwoju/unesco-a-zrownowazony-rozwoj (accessed on 12 February 2021).
- 26. Declaration of the United Nations Conference in the Human Environment (Stockholm); United Nations: New Park, NY, USA, 1972.
- 27. World Conservation Strategy. 1980. Available online: https://portals.iucn.org/library/efiles/documents/wcs-004.pdf (accessed on 12 February 2021).
- 28. Bosselmann, K. The Principle of Sustainability, Transforming Law and Governance; Routledge: Hampshire, UK, 2008.
- 29. Brundtland Report. *Report of the World Commission on Environment and Development: Our Common Future;* Oxford University Press: Oxford, UK, 1987; Available online: https://sustainabledevelopment.un.org/content/documents/5987our-common-future.pdf (accessed on 15 March 2021).
- 30. *The Rio Declaration on Environment and Development;* United Nations: Rio de Janeiro, Brazil, 1992; Available online: https://www.un.org/en/development/desa/population/migration/generalassembly/docs/globalcompact/A\_CONF. 151\_26\_Vol.I\_Declaration.pdf (accessed on 15 March 2021).
- 31. Agenda 21. *United Nations Conference on Environment & Development Rio de Janerio*; United Nations: Rio de Janeiro, Brazil, 1992; Available online: https://sustainabledevelopment.un.org/content/documents/Agenda21.pdf (accessed on 20 March 2021).
- 32. *Johannesburg Declaration on Sustainable Development of 4 September 2002;* United Nations: New Park, NY, USA, 2002; Available online: https://digitallibrary.un.org/record/478154?ln=en (accessed on 15 March 2021).
- 33. French, D. International Law and Policy of Sustainable Development; Manchester University Press: Manchester, UK, 2005.
- 34. Decleris, M. *The Law of Sustainable Development: General Priciples*; Office for Official Publications of the European Communities: Luxembourg, 2000.

- 35. De Cedra Larragan, J. *Distributional Choices in EU Climate Change Law and Policy, Towards a Principled Approach?* Kluwer Law International: Alphen aan den Rijn, The Netherlands, 2011.
- 36. Bengoetxea, J. Principles in the European Constitutionalising Process. King's Coll. Law J. 2001, 12, 100–110. [CrossRef]
- 37. Mac Cormic, N. Institutions of Law: An Essay in Legal Theory (Law, State and Practical Reason). Oxford University Press: New York, NY, UK, 2008.
- 38. Treaty on European Union, Official Journal of the European Union, No C 191/1 of 29.7.92, (Treaty of Maastricht). Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=OJ:C:1992:191:FULL&from=NL (accessed on 1 April 2021).
- 39. Treaty of Amsterdam amending the Treaty on European Union, the Treaties Establishing the European Communities and Certain Related Acts—Final Act, Official Journal of the European Union, C 340 of 10.11.1997. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A11997D%2FTXT (accessed on 1 April 2021).
- 40. Consolidated versions of the Treaty on European Union and the Treaty on the Functioning of the European Union, Official Journal of the European Union, C 202 of 7.6.2016, (Treaty of Lisbon). 2016. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=OJ%3AC%3A2016%3A202%3ATOC (accessed on 1 April 2021).
- 41. Pyć, D. Zasada zrównoważonego rozwoju. In Europe of Judges; Brodecki, Z., Ed.; LexisNexis: Warsaw, Poland, 2007.
- 42. Kramer, L. EU Environmental Law; Sweet&Maxwel: London, UK, 2012.
- 43. Kozień, E.; Kozień, A. Efficiency of the Principle of Sustainable Development in the European Union Law. In *Quality Production Improvement–QPI 2019;* Ulewicz, R., Hadzima, B., Eds.; Walter de Gruyter (Sciendo): Warsaw, Poland, 2019; pp. 206–211.
- 44. Jans, J.H. European Environmental Law; Europa Law Publishing: Groningen, The Netherlands, 2000.
- 45. Winter, G. The Legal Nature of Environmental Principles i International, EC and German Law. In *Europe and the Environment*, *Legal Essays in Honor of Ludwig Kramer*; Onida, M., Ed.; Europa Law Publishing: Groningen, The Netherlands, 2004.
- 46. Macrory, R. Principles of European Environmental Law; Europa Law Publishing: Groningen, The Netherlands, 2004.
- 47. Epiney, A. Environmental Principles. In *Reflection on 30 Years of EU Environmental Law: A High Level of Protection?* Macrory, R., Ed.; Europa Law Publishing: Groningen, The Netherlands, 2006.
- 48. Jans, J.H.; Vedder, H.H.B. European Environmental Law; Europa Law Publishing: Groningen, The Netherlands, 2008.
- 49. Verschuuren, J. Principles of Environmental Law; Nomos: Baden-Baden, Germany, 2003.
- 50. Constitution of the Republic of Poland of April 2, 1997, Dz.U.—Journal of Laws of 1997, No. 78, Item 483 as Amended. Available online: https://www.sejm.gov.pl/prawo/konst/angielski/kon1.htm (accessed on 1 April 2021).
- 51. Judgment of the Constitutional Tribunal in Poland dated 6 June 2006, file ref. K 23/05, OTK-A 2006, No. 6, item 62. 2006. Available online: https://ipo.trybunal.gov.pl/ipo/view/sprawa.xhtml?sprawa=3864&dokument=226 (accessed on 1 April 2021).
- 52. Kozień, A. Uwarunkowania administracyjnoprawne ekonomicznej koncepcji zrównoważonego rozwoju. In *Tendencje Rozwojowe Prawa Administracyjnego;* Kruk, E., Lubeńczuk, G., Drab, T., Eds.; Wydawnictwo UMCS: Lublin, Poland, 2017.
- 53. Pruszyński, J.P. Dziedzictwo kultury Polski. Jego straty i ochrona prawna, Volume I; Kantor Wydawniczy Zakamycze: Kraków, Poland, 2001.
- 54. Zeidler, K. Prawo Ochrony Dziedzictwa Kultury; Wolters Kluwer: Kraków, Poland, 2007.
- 55. UNESCO. Convention Regarding the Safeguarding of Cultural and Natural Heritage of 16 November 1972, Dz.U.—Journal of Laws of 1976, No. 32, Item 190. Available online: https://legal.un.org/avl/ha/ccpwcnh/ccpwcnh.html (accessed on 1 April 2021).
- 56. UNESCO. Convention for the Safeguarding of the Non-Tangible Cultural Heritage, Executed in Paris on 17 October 2003, Dz.U.—Journal of Laws of 2011, No. 172, Item 1018. Available online: http://portal.unesco.org/en/ev.php-URL\_ID=17716& URL\_DO=DO\_TOPIC&URL\_SECTION=201.html (accessed on 1 April 2021).
- 57. Tsivolas, T. Law and Religious Cultural Heritage in Europe; Springer: Berlin/Heidelberg, Germany, 2014.
- 58. Purchla, J. Dziedzictwo kulturowe. In *Kultura a Rozwój;* Hausner, J., Karwińska, A., Purchla, J., Eds.; Narodowe Centrum Kultury: Warsaw, Poland, 2013.
- 59. Kenterelidou, C.; Galatsopoulou, F. Sustainable Biocultural Heritage Management and Communication: The Case of Digital Narrative for UNESCO Marine World Heritage of Outstanding Universal Value. *Sustainability* **2021**, *13*, 1449. [CrossRef]
- 60. Pardo Abad, C.J. Valuation of Industrial Heritage in Terms of Sustainability: Some Cases of Tourist Reference in Spain. *Sustainability* **2020**, *12*, 9216. [CrossRef]
- 61. Yttredal, E.R.; Homlong, N. Perception of Sustainable Development in a Local World Heritage Perspective. *Sustainability* **2020**, 12, 8825. [CrossRef]
- 62. Magliacani, M. Managing Cultural Heritage: Ecomuseum, Community Governance and Social Accountability; Palgrave Macmillan: New York, NY, USA, 2015.
- 63. Kozień, A. Contemporary challenges of cultural heritage management by public and private entities. In *Economic and Social Development (Book of Proceedings), 65th International Scientific Conference on Economic and Social Development;* Aleksic, A., Ruzic, V., Baracskai, Z., Eds.; Varazdin Development and Entrepreneurship Agency: Varazdin, Croatia, 2021.
- 64. Knysiak-Sudyka, H. Zasady postępowania administracyjnego jurysdykcyjnego. In *Postępowanie Administracyjne;* Woś, T., Ed.; Wolters Kluwer: Warsaw, Poland, 2017.
- 65. Bielecki, L. Publiczne prawa podmiotowe, interes publiczny, władztwo administracyjne, władza dyskrecjonalna i inne postacie luzu decyzyjnego. In *Prawo Administracyjne;* Zdyb, M., Stelmasiak, J., Eds.; Wolters Kluwer: Warsaw, Poland, 2016.

- 66. Wyrzykowski, M. *Pojęcie Interesu Społecznego w Prawie Administracyjnym*; Wydawnictwa Uniwersytetu Warszawskiego: Warsaw, Poland, 1986.
- 67. Zimmermann, J. Prawo Administracyjne; Wolters Kluwer: Warsaw, Poland, 2016.
- 68. Dworkin, R. Taking Rights Seriously; Harvard University Press: Harvard, MA, USA, 1977.





# Article Information Modelling Management and Green Public Procurement for Waste Management and Environmental Renovation of Brownfields

Laura Pellegrini <sup>1,\*</sup>, Mirko Locatelli <sup>1</sup>, Silvia Meschini <sup>1</sup>, Giulia Pattini <sup>1</sup>, Elena Seghezzi <sup>1</sup>, Lavinia Chiara Tagliabue <sup>2</sup> and Giuseppe Martino Di Giuda <sup>3</sup>

- <sup>1</sup> Department of Architecture, Built Environment and Construction Engineering, Politecnico di Milano, 20133 Milano, Italy; mirko.locatelli@polimi.it (M.L.); silvia.meschini@polimi.it (S.M.); giulia.pattini@polimi.it (G.P.); elena.seghezzi@polimi.it (E.S.)
- <sup>2</sup> Department of Computer Science, University of Turin, 10149 Turin, Italy; laviniachiara.tagliabue@unito.it
- <sup>3</sup> Department of Management, University of Turin, 10134 Turin, Italy; giuseppemartino.digiuda@unito.it
- \* Correspondence: laura1.pellegrini@polimi.it; Tel.: +39-02-23-995-749

Abstract: Information Modelling and Management (IMM) methods for Most Economically Advantageous Tender (MEAT) can promote the adoption of environmentally sustainable practices. Despite the wide regulatory framework and existing drivers, Construction and Demolition Waste (CDW) trends are still growing. The literature review analyzed IMM and CDW management implementation during design phases although few studies focused on Green Public Procurement (GPP) and CDW management integration from the Public Client's point of view. This research aims at investigating the integration and efficiency of MEAT and IMM to promote the application of sustainable strategies focused on waste reduction and resource valorization. The study investigates the Public Client's role in promoting sustainable practices, introducing digital material inventory and BIM during the design phases, and including environmental award criteria in the call for tender documents. A Design Build (DB) procurement model was considered in the case study of a brownfield renovation and the construction of a new school in northern Italy. The methodology provided the Public Client and included a method to evaluate the environmental impact of the bids, allowing for proper selective demolition planning, CDW decrease, and organization while promoting their integration in companies' expertise and procedures. The replicability of the methodology is demonstrated by positive results of present and previous case studies.

**Keywords:** building information modelling (BIM); construction and demolition waste (CDW); design build (DB); environmental assessment; public client; waste minimization

#### 1. Introduction

The construction sector is an industry with a high intensity of raw material consumption, corresponding to about a half of the Earth's raw materials [1,2]. In addition, over one half of annual material input returns as waste in the industrialized countries every year [3]. Construction and Demolition Waste (CDW) accounts for one third of the total amount of waste by volume in the EU, as reported by the EUROSTAT 2019 Waste Statistics [4]. Considering the high consumption of raw materials and the high level of discard production, the CDW management process must be reengineered to reduce scrap and to take advantage of the high potential for reusing and recycling materials when construction waste is correctly identified and separated through selective demolition procedures [5]. Waste management in the Architecture, Engineering, Construction, and Operations (AECO) industry requires adopting a 'cyclic' rather than a 'linear' approach [6]. The transition toward a circular economy approach in the AECO sector is aligned with the global framework to avoid dangerous climate change effects set by the Paris Agreement [7].

Citation: Pellegrini, L.; Locatelli, M.; Meschini, S.; Pattini, G.; Seghezzi, E.; Tagliabue, L.C.; Di Giuda, G.M. Information Modelling Management and Green Public Procurement for Waste Management and Environmental Renovation of Brownfields. *Sustainability* **2021**, *13*, 8585. https://doi.org/10.3390/ su13158585

Academic Editor: Antonio Caggiano

Received: 26 June 2021 Accepted: 28 July 2021 Published: 1 August 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Most of existing studies adopted Building Information Modelling (BIM) approaches to introduce environmentally sustainable practices, specifically to the AECO industry, in the building design phase from the designer and contractor's points of view [8]. Few studies have analyzed the integration of BIM and Green Public Procurement (GPP) in construction procurement to evaluate the environmental impact of the bids [9]. However, a comprehensive application of Information Modelling and Management (IMM) methods and GPP from the Public Client's point of view, aiming to evaluate and reduce the environmental impact of a building related to resource and waste management during design and call for tenders phases, represents a gap in the literature. This can be critical considering the high purchasing power of public organizations and institutions [10] and the poor competences in the Italian context in drafting calls for tender documents including environmentally sustainable criteria [11].

The goal of the presented research is to fill this gap by defining a replicable model that adopts IMM and GPP for a sustainable use of resources and CDW minimization and management in AECO sector. The methodology covers the design phases and the call for tenders phase of a public construction process.

During the preliminary design phase, digital techniques for field surveys and data management are applied by a defining waste audit and a related digital inventory of materials. This supports proper selective demolition planning and demolition waste management. During the design phase, the creation of BIM models for the designed facilities enables construction waste management and the definition of on-site reusing and off-site recycling strategies for demolition waste materials.

During the call for tenders phase, GPP is adopted to a Most Economically Advantageous Tender (MEAT) procedure by introducing award criteria to evaluate the bids in terms of their environmental impact and integration of sustainable practices. All of the data and information from the design stages performed via IMM approaches support the definition of the tender documents and award criteria. The BIM model allows the management of waste and the related recycled and reused quantities for construction site planning.

This research will provide the Public Client a method to integrate and evaluate CDW minimization and selective demolition criteria from the design phase to the call for tenders phase of a public construction process, thus encouraging construction companies to propose and apply sustainable practices. This will also help companies to embrace circular economy principles and processes by renewing and enhancing their own business models. The efficacy of the proposed methodology is evaluated through the analysis of the responses from construction companies during the call for tenders, and in particular, the responses from the winning one.

In previous studies, the described methodology was tested on a green field of a new construction project, namely a public school facility in Melzo, Italy [12]. In the present research, the model is applied to a brownfield renovation, including an extensive demolition phase and the final design and construction of a new public school complex in Inveruno, Italy. The selection of the case study aims to demonstrate the positive outcomes of the model application to support the proper renovation of brownfields, thus promoting renovation processes to recover the quality of community land. Furthermore, the application of the model on a case study that is more complex than the previous one mentioned above can demonstrate the replicability of the proposed methodology.

#### 2. Regulatory Framework and Guidelines

The main regulatory frameworks regarding GPP and waste management are hereafter investigated as the foundation of the proposed methodology. This section aims to analyze the topics describing global goals and directions while presenting the Italian position on the issues. In addition, soil protection and brownfield regulations are briefly analyzed since the case study is a brownfield revamp for new construction. The analyses are performed at the (Figure 1):

- 1. International/global level: identifying global strategic goals and actions in order to define the direction and guidance toward sustainable practices;
- 2. European level: describing the general framework and the main guidelines;
- 3. National level: analyzing Italian regulations and, if existing, regional specifications and requirements.



Figure 1. Regulatory levels.

Finally, at the end of each of the following sub-sections, the key aspects that the proposed methodology aims to address are presented.

### 2.1. Green Public Procurement

Green Public Procurement integrates requirements and criteria in order to achieve value for money in the whole lifecycle of a project while supporting resource protection [13] and reducing environmental impacts [14]. Since public organizations and institutions wield a purchasing power of 15 to 30% of the national gross domestic product [10], GPP is supported and promoted at the international level as a means of driving the market towards innovative and sustainable practices [15].

At the European level, the Directive 2014/24/EU introduced the possibility of including tender clauses related to environment protection, the minimization of negative impacts, and waste management by promoting and monitoring the implementation of GPP [16,17]. In addition to that, the Directive 2014/24/EU emphasizes the critical role of public procurement in reaching a smart, sustainable, and inclusive growth [17].

Concerning the Italian regulatory framework, the Legislative Decree 50/2016 was a turning point for national procurement regulations, making GPP mandatory in public tenders and introducing the Most Economically Advantageous Tender (MEAT) approach, which aims to achieve value for money on a basis of the lifecycle of a construction project [18]. However, only around 20% of the Italian Provinces have adopted GPP including sustainability criteria so far, while around 50% of municipalities have very poor knowledge of GPP. The main causes, especially when considering smaller municipalities, are the lack of staff training and competences regarding GPP, and issues in drafting tender documents that include sustainability criteria [11].

Concerning GPP implementation, the presented research aims to:

- provide Public Clients a method for GPP and environmentally sustainable criteria integration that can be further applied and adapted by municipalities;
- test the effectiveness of the proposed methodology through the presented case study.

### 2.2. Waste Management

Fundamental objectives to achieve a sustainable development, which are set at international level, are the efficient use of natural resources and the 3Rs approach, i.e., reduce, reuse, and recycle, for waste minimization and management [19,20]. A critical step for waste management regards the definition of the concept of Extended Producer Responsibility (EPR) [21]. The focus is shifted from the phase in which waste already exists to the entire product lifecycle to the responsibility of manufacturers in terms of waste generation and to the need for the sustainable use of resources and waste minimization [22].

In the European framework, attention to both the design phase supporting the EPR approach and the conscious choices of materials, i.e., recyclable, recycled, durable, non-hazardous, and local materials and products, and the concept of waste as a resource, is stressed [23,24]. The definition of selective demolition is also proposed, highlighting the need to integrate it in standard design procedures [23]. Furthermore, in 2018, the EU Commission released guidelines for the definition of waste audits: they have a central role to ensure the proper identification and separation of demolition materials, thus facilitating the application of selective demolition and reusing and recycling practices [5]. Waste audits exploit their major potential if performed during the design phase since they enable a detailed estimation of the costs of the selective demolition, separation, recycling, recovery, and landfilling of materials. Consequently, a correct estimation of costs can be encompassed in the total budget for the call for tenders, supporting a successful accomplishment of the process [25].

Concerning the Italian regulatory framework, Legislative Decree 50/2016 introduced the requirement for Public Clients to purchase products and services that are compliant with the national document called the Minimum Environmental Criteria (CAM—Criteri Ambientali Minimi) [18], which requires the reuse, recovery, or recycling rate of about 70% by weight of non-hazardous waste in cases such as renovation, maintenance, and demolition. Methodological hints and principles for demolition and material removal and requirements to conduct a pre-demolition audit were also defined in 2017 by a specific Ministerial Decree [26].

Concerning CDW management implementation, the presented research aims to:

- define a waste audit method for the preliminary design phase;
- integrate CDW management in the entire construction process with increased interest on the design phase, on sustainable resource use, and on waste minimization.

#### 2.3. Soil Protection and Brownfield Rehabilitation

Brownfields are defined as polluted and abandoned sites where urban transformation interventions combine remediation and reuse [27,28]. At international level, soil protection, reducing land degradation and consumption, and strategies for regreening city spaces are among the 2030 Agenda Goals for Sustainable Development [14,29].

The EU targets to achieve zero net land take by 2050 [30]. In fact, soil is fundamental for life on Earth and is a non-renewable resource [31]. Spatial planning strategies for urban expansion cause most of soil sealing. The EU Commission emphasized the need to implement good practices to reduce the negative effects of soil consumption and, in particular, to reduce soil sealing [32,33], identified as major threats to the environment [31].

Given the condition of fragility and criticality of the territory, soil protection, the sustainable management of natural resources, and arrested land consumption and desertification are crucial for the Italian context [34]. The Italian brownfield regulation falls under waste regulations, in which brownfield redevelopment is divided into site environmental remediation and subsequent urban redevelopment [35]. At the regional level, Lombardy Regional Law 16/2017 defined a digital catalogue of brownfields and introduced the con-

cept of 'ecological soil balance' and economic incentives for brownfield renovations and limitations on green fields use for new construction projects [36,37]. In addition, at project level, CAM introduced the need and indications to reduce land use and sealing, prioritizing the recovery of existing abandoned buildings and brownfields [26].

Concerning brownfield renovations, the presented research aims to:

- test the proposed methodology on a brownfield renovation;
- evaluate the advantages of proper waste audit definition and CDW management for the extensive demolition phase planning of an abandoned industrial area.

### 3. Background

As presented in the previous section, CDW management and minimization as well as GPP adoption in the construction sector are promoted and sustained both at the international, European, and national level. Nevertheless, the level of waste generated by the AECO sector is still growing. EUROSTAT 2019 Waste Statistics compared waste generation trends in European industrialized sectors in the period of 2004–2018 [4]. While other industrialized sectors had reduced their waste generation by over 20%, the construction sector had increased its waste production by about 20% since 2004 (Figure 2) [4].



**Figure 2.** Comparison of current waste generate trends in the construction, mining and quarrying and manufacturing sectors, excluding major mineral wastes, EU-28, 2004–2018. Data source: EUROSTAT 2019 (env\_wasgen).

The following sub-sections investigate the barriers that currently limit the introduction of sustainable practices in the AECO sector and drivers that can promote and guide the adoption of CDW management and minimization strategies.

#### 3.1. Barriers to Waste Management Adoption in Construction Sector

The main barriers to the reduction of waste generation are represented by socioeconomic factors [38,39], and these are described in Table 1.

Barriers	Effects on the Construction Process		
Incomplete design documents and information asymmetry between designers and contractors	Causes most errors and rework during the construction phase [40], consequently triggering waste generation [41,42] and increasing the risk of project time and cost overruns.		
Non-collaborative culture	One of the main causes of waste generation [43], producing an unclear definition of responsibilities and inconsistent tender documents [44]. The contractor is usually not involved in design choices, increasing the risk of waste generation during the construction phase [45].		
Blame culture	Actors of the different stages blame other specialists for generating waste [46,47] with a consequent shift in the responsibility for waste generation.		
Waste inevitability culture	All actors see waste generation as unavoidable [6]. In standard practice, owners and contractors have usually already paid for waste treatments, transportation, and landfilling; therefore, selective demolition and the adoption of the 3Rs approach are seen as burdens.		
Difficulty of innovating	Difficult replication of good practices and innovative applications in future projects due to the temporary relationships between the construction process parties [48].		
Lack of training in applying innovative methods and technologies	Difficult application of innovative methods and technologies by the contractor [6].		

Table 1. Barriers for the implementation of sustainable practices in AECO industry.

The proposed methodology aims to overcome the identified barriers by:

- promoting and demonstrating the concept of waste as a resource and as a possible source of income;
- involving the contractor in the design phase to enhance the collaboration and to enable the proper application of CDW management and minimization strategies by adopting the Design Build (DB) tender model;
- providing a replicable methodology to support the adoption of sustainable strategies and innovative methods by construction companies.

# 3.2. Drivers to Waste Management Adoption in Construction Sector

The main drivers for CDW management and minimization strategies and GPP implementation in the AECO sector are:

- Legislative drivers: as presented in the Regulatory framework section, the implementation of environmentally sustainable practices is promoted at all levels. Consequently, Public Clients can encourage the introduction of sustainable practices by implementing GPP tender processes and by using their purchasing power to opt for environmentally friendly goods, services, and works. In addition, positive public applications can highlight the advantages and possible income deriving from selective demolition, waste separation, and the recycling as secondary raw materials, such as, as an example, the high-quality recycling of aluminum.
- Business drivers: companies are facing the need to adopt sustainable strategies in order to maintain and enhance their competitiveness. In addition, companies are starting to focus not only on economic value but also on the company's social responsibilities and stakeholder engagement [6].
- Managerial and technological drivers: Information Modelling and Management (IMM) approaches can allow companies to overcome the barriers of sustainable practice integration by reducing design errors and information asymmetry, improving collaboration, and supporting waste quantity measurement and control [39].

### 4. Literature Review

This section aims to investigate previous applications of IMM approaches to support the integration of environmentally sustainable practices in the AECO sector. The literature review briefly analyzes:

- digital survey techniques for the preliminary analyses of existing areas;
- the integration of BIM and CDW management during the design phase;
- GPP implementation supported by IMM approaches.

#### 4.1. Digital Techniques for Field Surveys

One of the critical steps when dealing with existing buildings is collecting real-world data about the to-be-demolished structures. There are two different approaches for digital field surveys that are investigated: photogrammetry and laser scanning.

Photogrammetry is a well-established digital image-based survey technique that extracts data from 2D pictures, and places them in 3D spaces [49], providing 3D measurements and producing points clouds for several engineering fields [50]. Close-range photogrammetry considers measurement distances within 300 m between cameras and buildings [51]. Photogrammetry allows the safe and easy analysis of unsafe buildings [52] with a reduced number of on-site field surveys, and enables further analysis of the buildings in remote mode. In addition, it is less expensive than laser scanning techniques [53], and non-experts can conduct surveys using common, portable, and lightweight digital cameras [54]. However, photogrammetry cannot provide automatic modelling of BIM objects; the points cloud can be used as a basis for the creation of the BIM model.

Laser scanning can be considered as one of the most innovative approaches in the field, and aims to convert real objects into information-rich BIM objects. Modern laser scanning techniques are less expensive than traditional ones, such as terrestrial or aerial laser scanning [55]. However, the conversion process from the points cloud to a 3D plain model and then to an information-rich BIM model is still a time-consuming and expensive task [56]. Recent studies have tested Artificial Intelligence (AI) techniques to achieve an automatic or semi-automatic and therefore faster and less expensive conversion [57,58].

In summary, photogrammetry enables safe, low-cost field surveys but lacks automatic BIM object modelling capabilities. On the other hand, laser scanning enables the conversion from points clouds to BIM models, but it is more expensive and is still a developing technique.

#### 4.2. BIM Methods for Waste Management

Better data and information flow means improved decision-making processes and, in addition, enhanced capacity to manage and decrease CDW streams. IMM methods play the double role of managerial and technological drivers [6], supporting the implementation of sustainable strategies and waste management practices during all stages of the construction and demolition processes [39,42]. Digital methods and techniques are demonstrated to guarantee the consistency of the information flow during the entire construction and demolition process through several applications [59–62]. Regarding the integration of BIM and CDW minimization and management, most of the existing studies focused on CDW reduction and management during the design phase. Cheng and Ma [63] proposed a BIM-based system for estimating and planning demolition and reconstruction activity by reducing the waste stream. They proposed an automated and accurate waste estimation methodology using the quantity take-off enabled by a BIM-model with a fast and more cost-effective approach. The authors addressed the lack of data and information that afflicts the estimation task during the demolition planning and reconstruction activities [64] in order to support the decision-making process. The correct quantification of waste, which must provide waste typology and dimensions, is in fact critical for effective and proper waste management [65]. Cheng et. al. [66] investigated how BIM implemented during the design phase could support CDW reduction during the construction phase by eliminating the root causes of waste generation through clash detection, quantity take-off, phase planning, site utilization, and digital prefabrication. The study aimed to maximize the homogeneous fractions of mono-materials by using BIM, and they proposed a BIM-based waste management planning and execution system to monitor the minimized and disposed waste [66]. Akinade et al. [67] proposed a BIM-based Deconstructability Assessment Score system (BIM-DAS) to compare several design options by identifying the best technological solution for deconstructability purposes and the most influential design factors influencing deconstructability. Deconstruction is defined as the possibility of disassembling a whole or a single element of a building, enhancing the technical component of reuse and recycling [68], leading the sector to reach a zero-waste economy [69–72]. Liu et al. [73] investigated a BIM-aided construction waste minimization framework, aiming to support designers in addressing waste causes during the decision-making processes of the design stages. Guerra et al. [74] investigated an automated waste estimation system, aiming to streamline the waste estimation process and support more efficient waste management during the design phases. The authors implemented their research for the visual planning of construction waste, discretizing waste generation in quantities for on-site reuse and off-site recycling, and identifying specific activities in the construction schedule where reuse is possible [75].

The literature review highlighted the extensive implementation of BIM methodologies to support CDW management during the design phases through the parametric modeling, visualization, and simulation capabilities of BIM. Existing studies showed promising results, with possible waste reduction of 4–15% by using BIM during the design phases [76]. However, the application of BIM methodologies for CDW minimization and to manage the demolition or reconstruction of existing buildings that had not been designed following deconstructability principles is a less investigated topic [8].

#### 4.3. IMM Approaches for Green Public Procurement Implementation

Green Public Procurement aims to purchase goods at a good value for money during the whole product lifecycle, while supporting environmental protection and reducing negative environmental impacts [14]. For that reason, green purchasing criteria are fostered by global, European, and national regulations. Despite this, the topic of IMM approaches for GPP implementation is not often investigated, and few applications in the construction sector can be found in the scientific literature. Palmujoki et al. [77] stated that environmental requirements are considered as contract award criteria, but the majority of purchasers still tend to favor past practices to avoid violating the legal principles of free competition and transparency, and possible legal disputes. Wong et al. [78] highlighted the importance of the Public Client in taking a proactive role in pushing green procurement adoption in the AECO sector. Through questionnaire surveys and interviews, the study identified environmental government regulations and client requirements in tendering as main factors to enhance GPP in the construction process. In addition, the use of IMM approaches was identified as a facilitator for the adoption of green design approaches in the construction process [78]. Barbini et al. [9] proposed a system to integrate life-cycle data in a BIM library to support Green and Digital Pubic Procurement processes in the Italian context. The research emphasized the positive outcomes of using digital technologies to achieve environmental sustainability objectives in a virtuous circle between green and digital approaches.

According to the literature review, little investigation has been conducted on the integration of IMM approaches and GPP. However, the definition of a method for the Public Client can be critical considering the high purchasing power of Public Administrations [10] and the poor competence in the Italian context in drafting tender documents that include environmentally sustainable criteria [11].

# 5. Methodology

The proposed methodology is divided in the following sub-sections:

- 4. Discussion on the literature review findings and the resulting goals of the research project, and choices for the definition of the proposed methodology;
- 5. Methodology for the preliminary design phase focusing on the analyses needed to plan appropriate selective demolition activities and involving the definition of a waste audit with a digital material inventory;
- 6. Methodology for the design phase focusing on the use of BIM models for CDW management and sustainable resource use;
- 7. Methodology for the call for tender phase focusing on the integration of sustainable award criteria, including CDW management, resource management, waste minimization, and selective demolition criteria, in the GPP tender documents in order to promote their application during the final design and construction phases.

## 5.1. To BIM, or Not to BIM?

The literature review highlighted the successful applications of BIM for sustainability and during the design phase in particular. However, less investigated topics are the application of digital methods to optimize the planning of the demolition phase of existing buildings and to support the Public Client during the call for tenders phase, including environmentally sustainable criteria.

Regarding BIM adoption for the demolition phase planning, the creation of a BIM model exclusively for demolition purposes shall be carefully evaluated. The availability of a BIM model of the to-be-demolished building is an undoubted advantage for CDW management [67], supporting deconstruction, and reusing and recycling activities [68]. However, most of the existing buildings do not have a BIM model, and creating one is challenging [79]. In addition, when generating a BIM model that is exclusively for the demolition phase, many benefits and optimizations that justify the effort of producing a BIM model are invalidated [25]. In general, the IMM approach aims to support decisionmaking processes by focusing on organizing and digitizing data and information in a structured form. Data and information must be available, up-to-date, precise, and provided to the right person, at the proper time, and in the correct format [80,81], in a concurrence with process and information management [82]. Therefore, data should accomplish two fundamental pre-requisites: (I) information readiness and (II) information processability [83]. Databases are comparable to BIM models in fulfilling the two above-mentioned pre-requisites since BIM models can be assimilated to the relational databases that are required for 3D geometrical visualization [73]. In addition, the creation of a digital database is less time and effort consuming than producing a full BIM model.

Consequently, the presented methodology proposes the adoption of IMM approaches by using a digital database to manage CDW and for the proper planning of selective demolition activities. The creation of a BIM model, on the other hand, is proposed for the design phase, supporting sustainable resource use and waste minimization strategies. Concerning the call for tenders phase, the methodology proposes a framework for the introduction of award criteria regarding sustainability aspects in the tender documents, and for the evaluation of the environmental impact of the bids (Figure 3).

In summary, the research aims to provide the Public Client with a model to adopt sustainability practices focusing in particular on CDW management and minimization and selective demolition practices, that are supported and managed via IMM approaches during the design and call for tenders phases.



Figure 4. General scheme for waste audits during the preliminary design phase within a IMM approach.

In order to collect the information needed to define the digital material inventory, close-range photogrammetry was considered. The output of close-range photogrammetry are points clouds realized from the images collected on site. This technology enables 3D measurements and the remote analysis of the images to identify materials and elements.

Data collected through field surveys are the basis for the definition of the digital material inventory, i.e. the output of the waste audit. The digital material inventory allows the organization and management of a higher quantity of data by going beyond the basic level of information of a standard inventory of materials [5]. The proposed digital material inventory structure is shown in Table 2, and includes:

- Basic information, including material estimation and description;
- Detailed information, including material types and European Waste Codes (EWC);
- An improvement of the basic levels of reporting hazardous and non-hazardous materials by considering three levels of reporting: (a) hazardous, (b) non-hazardous and recyclable, and (c) non-hazardous and reusable on site. The three levels enable an easier definition of waste management strategies through the prior identification of recyclable and reusable materials;
- The identification of recyclable materials and authorized recycling plants in the proximity of the project site to minimize the carbon emissions for waste transportation, hence promoting a circular economy approach at the local scale [25].

EWC Code	Material Description	Quantity	% of the Total (in Weight)	Waste Management Strategy Hypothesis	Distance of Proposed Recycling Plant
		[m <sup>3</sup> /kg]	[%]	[Reuse; recycle; landfill]	[km]

The digital material inventory is then used to define proper selective demolition activities and their related costs. Consequently, a more reliable estimation of the costs for the demolition phase can be included in the total budget for the subsequent design and call for tenders phases. In addition, selective demolition plans can be defined considering actual waste quantities and components.

The evaluation of the methodology applied to the case study was performed by comparing the selective demolition costs calculated through the digital material inventory with the costs of the parametric non-selective demolition of the same buildings.

### 5.3. Design Phase: Waste Minimization and Sustainable Resource Use within a BIM Approach

During the design phase, the BIM approach was proposed for waste minimization and management and to foster the sustainable use of resources (Figure 5).

The BIM model for the design phase was generated by linking all construction materials to similar information as the ones collected for the demolition materials in the digital material inventory. Material quantities, types, related quantities of construction waste, and EWC codes are linked to each construction material.

The BIM model is used to:

- Store all graphic and non-graphic information, drawings, and documents of all the disciplines of the design phase;
- Perform design reviews, clash detection, quantity take-off, phase planning, and site utilization, thus supporting and facilitating the proper management and minimization of construction waste;
- Identify materials producing the highest quantities of construction waste.



Figure 5. General scheme for waste management during the design phase within a BIM approach.

The quantities of construction materials from the BIM-based quantity take-off and from the digital material inventory can be compared in order to identify possible reuse strategies on site. It is possible to select the construction materials that can be replaced by demolition waste materials as secondary raw materials, allowing for a reduced use of raw materials.

The effectiveness of the methodology applied to the case study is validated by analyzing the advantages enabled by the use of the BIM model, in particular, regarding the sustainable use of resources through the on-site reuse of demolition waste materials.

#### 5.4. Call for Tenders Phase: Environmental Award Criteria within IMM and BIM Approaches

GPP and IMM methodologies are considered as key aspects to introduce and integrate sustainable practices in AECO industry. This sub-section describes the innovative approach to the call for tenders phase, aiming to evaluate and compare the bids according to their environmental impact, and regarding the aspects of CDW minimization and management, and selective demolition procedures.

The adopted tender process type is the Design Build (DB) model, in which a single operator is selected to conduct the final design and construction. The presence of a unique actor for the two stages eases a more efficient information transfer with the Client and a more collaborative process, which also ensures an easier management of the procurement and construction processes for the Client [84]. As a consequence, the DB model represents a valuable framework for the application of IMM approaches [85]. In addition, if the construction company realizes the final design, company characteristics and standard procedures can be taken into account. As a result, the final design can be optimized according to the capabilities and characteristics of the company itself.

The call for tenders process based on the DB model implies the MEAT evaluation system introduced by D.Lgs. 50/2016, as recalled by Section 2.1. The evaluation of the bids is based on objective alpha-numerical criteria. The assessment of the quality of the bids is based on the rankings on the defined criteria (Table 3), thus promoting the automation of

the evaluation process. The environmental criteria considered in this research concerned the following sustainability aspects [86]:

- Selective Demolition Procedures Criteria: procedures that should be defined according to the waste audit and digital material inventory;
- Demolition Waste Management Criteria: considering demolition materials as defined by the waste audit. It was necessary to define recyclable demolition waste by at least 70%, a mandatory requirement introduced by CAM (Section 2.2). The digital material inventory represents a fundamental annex of the tender documents for the identification of demolition materials and waste;
- Recommended Authorized Recycling Plants: minimizing the distance from the construction site and the carbon emissions from waste transport according to the analyses of the waste audit and the digital material inventory;
- Construction Materials and Waste Management Criteria: their quantities, characteristics, level of recyclability or reusability to achieve recyclable construction materials by at least 70%, a mandatory requirement introduced by CAM (Section 2.2);
- Construction Technical Solutions and CDW Management: on-site reuse strategies for demolition waste materials according to the BIM-based analyses;
- Construction Technical Solutions: procedures to increase the levels of durability, maintainability, safety for construction, and to decrease the waste generation.

Category	Criterion	<b>Evaluation Sub-Criterion</b>	Sub-Criterion Value
	A.1—Material production site distance	A.1.1—Distance of the production site of materials	3
A—Passive element		A.3.1—Contractor certification according to UNI EN ISO 14001	1
requirements	A.3—Environmental requirements	A.3.2—Producer certification according to UNI EN ISO 14001	3
		A.3.3—Recyclability of materials at end-of-life	2
D—Construction and	D.1—Safety	D.1.1—Contractor certification according to OHSAS 18,001	1
demolition phase	D.2—Constructive solutions and site management	D.2.2—Demolition plan and CDW management	10
E—Maintenance	E.1—Building maintenance	E.1.1.—Maintainability and durability of construction materials	7

Table 3. Scheme of criteria and sub-criteria for the bid evaluation process.

Throughout the entire process, waste minimization paired with the definition of safe and efficient selective demolition procedures is promoted with the highest rankings. Respecting the environmental criteria when presenting the bids would have provided additional points to the ranking of the companies, thus increasing their possibility of winning the call for tenders, and, at the same time, promoting the application of sustainable practices among the participants. In particular, the call for tenders assigned 80 out of 100 points to the technical offer. Among them, 27 points regarded the environmental criteria (Table 3), covering more than 25% of total points. Moreover, environmental criteria were linked to safety aspects by the sub-criterion D.1.1, requesting the OHSAS 18001 certification of the construction company safety management processes (which has been replaced by International Standard ISO 45001 in 2018) [87,88].

In addition, the criteria and sub-criteria used to evaluate the bids in the tender are aligned with LEED certification criteria (Table 4) [89].

<b>Evaluation Sub-Criterion</b>	LEED Credit
A.1.1—Distance of the production site of materials	MR Credit: Sourcing of Raw Materials: products sourced (extracted, manufactured, and purchased) within 100 miles (160 km) of the project site are valued at twice their base contributing cost, up to a maximum of 200% of the cost.
A.3.1—Contractor certification according to UNI EN ISO 14001	ISO 14001 (evaluating environmental management system of companies) is not directly linked with LEED (evaluating building environmental performance). However, it has been demonstrated that companies implementing ISO 14000 standards are more likely to be able to provide
A.3.2—Producer certification according to UNI EN ISO 14001	buildings with higher energy and environmental performance, and consequently, higher LEED ratings [90,91].
D.2.2—Demolition plan and CDW management	MR Credit: Construction and Demolition Waste Management: develop and implement a construction and demolition waste management plan and achieve points through waste prevention and/or diversion.

Table 4. Parallelism between LEED credits and evaluation sub-criteria of the proposed tender.

The following paragraphs describe the sub-criterion D.2.2 (Table 3) and the related scoring system. Sub-criterion D.2.2 is related to CDW management and is aligned with the LEED "Material and Resource (MR) Credit: Construction and Demolition Waste Management". The intent of the LEED requirement is the reduction of "construction and demolition waste disposed of in landfills and incineration facilities through waste prevention and by reusing, recovering, and recycling materials, and conserving resources for future generations.", and specifically, the considered MR Credit aims "to divert at least 50% of the total construction and demolition materials from landfills and incineration facilities." [89].

Sub-criterion D.2.2 divides waste that the company expects to produce by:

- Re-usable waste on site, identifying the expected methods and, when necessary, the treatments to make waste reusable;
- Recyclable waste, identifying the related methods of separation of different types of waste on site;
- Landfilled waste, divided for hazardous and non-hazardous waste, including the disposal methods and, when necessary, treatments to be landfilled.
- Specifically, the following data shall be defined for each waste material (Table 5):
- EWC codes;
- Description of construction activities producing the waste materials;
- Total amount of m-th non-hazardous waste material  $Q^i_{Rif, m}$  and of p-th hazardous waste material  $Q^i_{Rif, p}$  (unit of measurement: kg);
- Percentage of non-hazardous waste material to be reused on site %<sup>i</sup><sub>Riu.m</sub>;
- Percentage of non-hazardous waste material, excluding excavated soil, to be recycled  $\sqrt[\%]{_{Ric, m}}^{i}$ ;
- Percentage of non-hazardous waste material, excluding excavated soil, to be landfilled %<sup>i</sup><sub>Sma.m</sub>;
- Percentage of hazardous waste material as defined by Directive 2008/98/CE to be landfilled %<sup>i</sup><sub>Sma. n</sub>.

FWC Code	Construction	Amount of	Non-Hazardous Materials			Hazardous Materials
	Activity Waste [kg]		% Reused	% Recycled	% Landfilled	% Landfilled
Non-hazardous material EWC code	Construction activity description	$Q^i_{Rif, m}$	% <sup>i</sup> <sub>Riu, m</sub>	% <sup>i</sup> Ric, m	% <sup>i</sup> Sma, m	-
Hazardous material EWC code	Construction activity description	$Q^i_{Rif, p}$	-	-	-	% <sup>i</sup> Sma, p

Table 5. Example of table for expected waste quantities, types, and EWC codes.

Considering each EWC non-hazardous waste code, the sum of the percentages  $\%^{i}_{Riu, m'}$  $\%^{i}_{Ric, m'}$   $\%^{i}_{Sma, m}$  shall be equal to 100% since it corresponds to the total amount of that specific non-hazardous waste material. On the other hand, for each EWC hazardous waste code, the single percentage  $\%^{i}_{Sma, p}$  shall be equal to 100% since the entire quantity of a hazardous waste material is represented by this percentage.

The quantities of each m-th non-hazardous waste material to be reused  $(Q_{Riu, m}^{i})$ , recycled  $(Q_{Ric, m}^{i})$ , and landfilled  $(Q_{Sma, m}^{i})$  shall be calculated as follows:

$$Q_{Riu, m}^{i} = Q_{Rif, m}^{i} \cdot \mathscr{N}_{Riu, m}^{i}; Q_{Ric, m}^{i} = Q_{Rif, m}^{i} \cdot \mathscr{N}_{Ric, m}^{i}; Q_{Sma, m}^{i} = Q_{Rif, m}^{i} \cdot \mathscr{N}_{Sma, m}^{i}$$

In addition, the quantities of each p-th hazardous waste material to be landfilled  $(Q_{Sma, p}^{i})$  shall be calculated as follows:

$$Q_{Sma, p}^{i} = Q_{Rif, p}^{i} \cdot \%_{Sma, p}^{i}$$

The total amount (in kg) of non-hazardous waste to be reused  $(Q_{Riu}^i)$ , recycled  $(Q_{Ric}^i)$ , and landfilled  $(Q_{Sma-Nper}^i; Nper; Subscript for non-hazardous waste) shall then be calculated as follows:$ 

$$Q_{Riu}^{i} = \sum_{m=1}^{N} Q_{Riu, m}^{i}$$
;  $Q_{Ric}^{i} = \sum_{m=1}^{N} Q_{Ric, m}^{i}$ ;  $Q_{Sma-Nper}^{i} = \sum_{m=1}^{N} Q_{Sma, m}^{i}$ 

The total amount (in kg) of hazardous waste materials to be landfilled ( $Q_{Sma-per}^{i}$ ; per: subscript for hazardous waste) shall also be calculated as follows:

$$Q^{i}_{Sma-per} = \sum_{p=1}^{N} Q^{i}_{Sma, p}$$

where N is the total number of materials from Table 5.

The total amount of waste ( $Q_{TOT}^i$ ) is the sum of recycled, reused, and landfilled waste, considering both non-hazardous (subscript <sub>Nper</sub>) and hazardous materials (subscript <sub>per</sub>).

$$Q_{TOT-Nper}^{i} = Q_{Riu}^{i} + Q_{Ric}^{i} + Q_{Sma-Nper}^{i}$$
$$Q_{TOT-per}^{i} = Q_{Sma-per}^{i}$$
$$Q_{TOT}^{i} = Q_{TOT-Nper}^{i} + Q_{TOT-per}^{i}$$

The total percentages of materials for reusing, recycling, and landfilling of non-hazardous (subscript <sub>Nper</sub>) and hazardous materials (subscript <sub>per</sub>) shall be defined as follows:

$$\mathscr{W}_{Riu}^{i} = \frac{Q_{Riu}^{i}}{Q_{TOT}^{i}}; \ \mathscr{W}_{Ric}^{i} = \frac{Q_{Ric}^{i}}{Q_{TOT}^{i}}; \ \mathscr{W}_{Sma-Nper}^{i} = \frac{Q_{Sma-Nper}^{i}}{Q_{TOT}^{i}}; \ \mathscr{W}_{Sma-per}^{i} = \frac{Q_{Sma-per}^{i}}{Q_{TOT}^{i}}$$

Finally, it is possible to calculate the coefficient  $D\%_{D,2,2}^{i}$ :

 $D\%^{i}_{D.2.2} = \%^{i}_{Riu} \cdot coef_{Riu} + \%^{i}_{Ric} \cdot coef_{Ric} + \%^{i}_{Sma-Nper} \cdot coef_{Sma-Nper} + \%^{i}_{Sma-per} \cdot coef_{Sma-per} + \%^{i}_{Sma-per} \cdot coef_{Sma-per} + \%^{i}_{Sma-per} \cdot coef_{Sma-per} + \%^{i}_{Sma-per} + \%^{i}_{Sma-per} \cdot coef_{Sma-per} + \%^{i}_{Sma-per} + \%^{i}_{Sma-per} \cdot coef_{Sma-per} + \%^{i}_{Sma-per} + \%^{i}_{Sma-per}$ 

where  $coef_{Riu}$ ,  $coef_{Ric}$ ,  $coef_{Sma-Nper}$ , and  $coef_{Sma-per}$  are defined in Table 6.

**Table 6.** Waste management coefficients for non-hazardous and hazardous waste materials according to waste treatments.

Classification of Waste	Waste Management Coefficient Code	Waste Management Coefficient Value
Reused non-hazardous waste materials	coef <sub>Riu</sub>	100%
Recycled non-hazardous waste materials	$coef_{Ric}$	80%
Landfilled hazardous waste materials	coef <sub>Sma-per</sub>	75%
Landfilled non-hazardous waste materials	coef <sub>Sma-Nper</sub>	50%

The final score for criterion D.2.2 of the i-th offer shall be calculated as follows:

$$P_{D.2.2}^{i} = D\%_{D.2.2}^{i} \cdot P_{D.2.2}$$

where:

- $P_{D.2.2}^{i}$  is the score of the i-th offer for criterion D.2.2;
- *P*<sub>D.2.2</sub> is the maximum score that sub-criterion D.2.2 can reach;
- D%<sup>i</sup><sub>D.2.2</sub> is the scoring percentage that multiplies the maximum score available for the criterion, obtaining the i-th offer's score as defined above.

The evaluation of the methodology applied to the case study was performed by analyzing the results of the call for tenders, and the responses of the participants, particularly the winning one, to the optional award criteria regarding environmentally sustainable aspects.

# 6. Case Study

The case study involved the decontamination and renovation of a brownfield in the Municipality of Inveruno in the Province of Milan, Italy, including an extensive demolition phase of the existing buildings in the industrial site and the subsequent construction of a new school complex for a total amount of EUR 15 M. The project area was classified as a brownfield in the open data set of the Lombardy Region.

#### 6.1. Case Study Selection: Importance of Soil Protection and Brownfield Rehabilitation

Brownfields are defined as polluted and abandoned sites where urban transformation interventions combine remediation and reuse [27,28]. As recalled in Section 2.3, the brownfield phenomenon and the importance of preserving green fields is widely acknowledged in Europe as well as in the Italian context. Despite that, land consumption in Italy maintained a rate of 2 square meters of land irreversibly lost every second in 2019 [28]. The most affected land type is green fields, i.e., areas with agricultural vocation. Green fields are three times more affected by land consumption than urban areas. In 15 out of 20 Italian regions, the consumed land exceeds 5% of the complete regional area. In particular, the highest percentage of consumed land, which is equal to 12,1%, belongs to the Lombardy Region [28]. Consequently, the remediation, reuse, and redesign of brownfields within a circular economy approach could enable the reduction of land consumption by avoiding taking advantage of green fields and helping achieving zero net land consumption by 2050 [30]. In addition, CDW management strategies can be even more effective when applied to brownfields [27,28]. Furthermore, disused former industrial areas inside urban contexts generate processes of environmental and social degradation, and their renovation can help in contrasting this combined phenomenon, representing, at the same time, an opportunity to redesign dense urban areas and to rethink the distribution of functions and services [37].

The presence of about 1,260 hectares of brownfields in the Province of Milan, corresponding to over twice the area of the Milan historical center [37], justifies the choice of the case study, since brownfield renovation is critical for the Province of Milan. The main goals of the case study application are: to integrate environmental sustainability by recycling demolition materials, and by reducing the use of raw materials and green fields during the construction phase; to integrate social sustainability through the transformation and rehabilitation of a brownfield into a new school complex area that will be open to the municipality.

A previous stage of the research project was discussed in an article by Pellegrini et al. [12], which also reported a brief preliminary description of the case study presented here, which was not yet fully developed. The tender has now been awarded, and this study describes the advantages, limitations, and effectiveness of the proposed method based on the analysis of the case study results.

### 6.2. Existing Area and Design of the New School in Inveruno

The project area has a total area of 18,229 square meters. There are several existing underground structures, portions of masonry, and a large surface covered by concrete floor. In addition, the area hosts a vegetable oil refinery building, which is 32 m tall with concrete walls and a composite steel-concrete deck (Figure 6). The project includes the decontamination of the brownfield and the selective demolition of the disused industrial buildings and concrete surfaces.



**Figure 6.** Project area in Inveruno showing the highest of the to-be-demolished existing buildings (on the left) and the existing underground structures (on the right).

Regarding the design of the school complex, this process involved the construction of secondary and primary school buildings and an auditorium. The school buildings have a courtyard shape, and in the courtyards, squares and gardens are located (Figure 7). All of the buildings and the external area surrounding the school facilities can be used separately from students, teachers and school staff and can be used by the citizens of the municipality. This represents a fundamental aspect concerning social sustainability for the transformation and rehabilitation of the brownfield.



Figure 7. Design configuration of the new school complex in Inveruno, Italy.

# 7. Results

# 7.1. Preliminary Design Phase: Waste Management and Selective Demolition Planning within a IMM Approach

During the preliminary design phase, the output of field surveys supported by closerange photogrammetry were image planes of the to-be-demolished facilities (Figure 8). The image planes were used for measuring, material recognition, and for conducting analysis in remote mode.



**Figure 8.** Image planes of two sides of the abandoned 32-m-high vegetable oil refinery building slated for demolition, obtained via close-range photogrammetry.

All the data and information regarding construction techniques, materials, and related data were gathered and digitized in the digital material inventory, some examples of which are shown in Table 7. The digital material inventory enabled the maximization of the quantity and quality of the recyclable materials by obtaining the fractions of mono-materials that were suitable for reuse or recycling as secondary raw materials.

EWC Code	Material	Quantity	% of the Total (Weight)	Waste Management Strategy Hypothesis	Distance of the Recycling Plant
17 09 04	Mixed construction and demolition waste, such as concrete and masonry	6716 m <sup>3</sup>	45% Possible reuse on site: Replenishment of underground volumes Formation of embankments Preparation of roadbeds for the path of construction vehicles Draining layer for lamination boxes Possible recycling by crushing on site or in authorized centers		30 km
17 04 05	Iron and steel	746 m <sup>3</sup>	51%	Transfer to ferrous materials recycling plant	4 km
17 03 02	Bituminous mixtures	345 m <sup>3</sup>	4%	Transfer to authorized disposal facility for non-hazardous waste	20 km

Table 7. Extract of material inventory data.

In addition, it supported the definition and adoption of selective demolition strategies (Table 7), by including the following data:

- Types of structures to be demolished;
- Safety measures to be adopted during the demolition activities;
- Material types and their EWC code classification;
- Quantity of materials and related management strategies, i.e., on-site reusing, recycling, or landfilling.

A central aspect that emerged from the field surveys and from the digital material inventory definition was the serious state of abandon of the entire vegetable oil refinery facility. None of the building elements could be dismantled and reused. In addition, the disassembly of the metallic components before the demolition would have been dangerous for the workers. Therefore, the sorting of the metal components was planned for after the demolition of the concrete and brick parts.

In addition to detailed demolition planning, it was possible to quantify the costs associated with the selective demolition activities by linking each activity to the related expected cost of execution. Consequently, it was possible to obtain a reliable prediction of the total costs of the selective demolition phase, which were compared to the costs associated with non-selective demolition procedures. Non-selective demolition procedures consider the demolition of an entire building without any interest in the materials involved and, consequently, does not allow the proper separation of the elements or the fractioning of mono-materials that are suitable for reuse or recycling as secondary raw materials.

For the present case study, selective demolition costs identified with the support of the digital material inventory accounted for EUR 514,381. On the other hand, non-selective demolition costs could be estimated at an amount of about EUR 405,020 by considering a unique parametric cost for the demolition of the total volumes. Consequently, the selective demolition costs represented 127% of the estimated non-selective demolition ones.

### 7.2. Design Phase: Waste Minimization and Sustainable Resource Use within a BIM Approach

During the design phase, the BIM model allowed the estimation of the quantities of the needed construction materials by means of the quantity take-off. Construction materials quantities were then compared with waste quantities stored in the digital material inventory, and finally, some applicable on-site reuse strategies were selected.

Among the selective demolition strategies proposed in the digital material inventory, the demolition of the concrete external floor surface and of the concrete parts of the vegetable oil refinery provided a considerable amount of concrete as a homogenous portion of waste. For both the great quantity of waste available and the various possible reuse strategies, the concrete waste was selected to be entirely reused on site (Table 8).

Table 8. Reuse activities of total quantity of concrete wastes as secondary raw materials.
--------------------------------------------------------------------------------------------

Reuse Activity	Amount of Surface or Volume Covered with Reused Materials	Quantity of Reused Material	% of the Total Concrete Wastes
Aggregate for the preparation of roadbeds for the paths of construction vehicles	2300 m <sup>2</sup> of construction site paths	470 m <sup>3</sup>	7%
Replenishment of underground volumes in the project area	5843 m <sup>3</sup> of underground volumes	5843 m <sup>3</sup>	87%
Aggregate for sidewalks realization and as draining layer for lamination boxes	2727 m <sup>2</sup> of sidewalks and remaining 144 m <sup>3</sup> as draining layer for lamination boxes	403 m <sup>3</sup>	6%

#### 7.3. Call for Tenders Phase: Environmental Award Criteria within IMM and BIM Approaches

This sub-section provides the results of the application of the presented DB call for tenders method to the case study analyzing the specific tender documents, followed by the responses and results of the bids, in particular, those of the winning company bid.

In regard to the tender documents, the digital material inventory supported the verification of the feasibility of the selective demolition strategies to be included in the tender documents. In addition, the digital material inventory supported the identification of selective demolition costs that could also be included in the tender documentation as a part of the overall budget for the proposed project.

Furthermore, starting from the digital material inventory and the BIM-based analyses during the design phase, the tender documents included the hypothesis of the on-site reusing of concrete wastes as secondary-raw-material aggregates. Consequently, a pulverizer was proposed as equipment to be used on site to reduce the dimensions of the inert materials and to make them uniform for further use.

Regarding the scoring of the bids in the environmental criteria described in Table 3, the average score of 10 out of 11 bids, excluding the lowest score, was equal to 23.4, corresponding to 87% of the total available score (27/100) in the environmental criteria (Figure 9). In particular, 10 out of 11 construction companies obtained good scores in sub-criterion D.2.2 (Table 3) by providing data and strategies for CDW management. Only one offer obtained a negative result, i.e., zero points in the sub-criterion, due to a lack of specifications and documentation on CDW management strategies.



Figure 9. Scores and rankings in environmental award criteria of the eleven bids, including the winning one (Bid 9).

Regarding the winning company (Bid 9, Figure 9), the bid scored a total of 25.7 out of 27 points on the environmental award criteria, which represented 26% of the total 100 points and 32% of the 80 points of the technical offer. Based on the waste audit and digital material inventory and on the suggestions from the tender documents regarding selective demolition as described above, the company proposed an entire plan and site layout for the demolition phase. The plan included:

- Selective demolition activities;
- Specific work areas inside the demolition site layout;
- Construction vehicles and equipment specific for different areas and activities.

In addition, the plan included safety requirements and indications for the demolition phase, e.g. areas of the site precluded to worker access near the areas where the total demolition of concrete structures would be performed with excavators equipped with demolition grabs. Furthermore, the company linked the analysis of the demolition phase and activities with information regarding the materials provided in the digital material inventory. Consequently, they could identify all waste quantities (Figure 10) and could select the specific authorized plants needed for recycling and landfilling, starting from the authorized plants suggested in the tender documentation. Figure 10 shows the types of materials and the related percentage of reused, recycled, and landfilled waste, aiming to minimize the landfilled waste.



**Figure 10.** Reused, recycled, and landfilled waste proposed by winning company and divided by material type.

Starting from the BIM-based suggestions for the on-site reuse strategies included in the tender documents, the winning company proposed to reuse most of the concrete, bricks, and ceramic demolition waste materials (EWC code equal to 170904) as secondary raw materials for the realization of the construction site roadbeds, as aggregate to realize sidewalks, and for the replenishment of underground volumes (Figure 11). The quantities of materials extracted from the BIM model of the designed buildings were compared to the quantities of waste materials recorded in the digital material inventory. Consequently, it was possible to define which quantities of the construction materials could be replaced with reused or recycled materials from the demolition of the existing facilities. In addition, in regard to soil and stones, the company planned for the majority of these materials to be reused on site for external works.



**Figure 11.** Comparison of design material quantities from the BIM model of the designed buildings, the selective demolition waste quantities from the digital material inventory, and the selection of reuse strategies.

Regarding the new school complex construction phase, the winning company used the BIM model in order to define and manage all quantities of waste and recyclable materials from the project. Figure 12 shows the total percentages of reused, recycled, and landfilled waste materials, accounting for around 45%, 46%, and 9% respectively. Consequently, over 90% of the waste was planned to be recycled and reused, minimizing the total quantity of landfilled waste accordingly.



Figure 12. Total percentages of reused, recycled, and landfilled waste.

### 8. Discussion

This section discusses the results of the application to the case study using IMM approaches during the design and call for tenders phases, which was presented in the previous section.

Regarding the preliminary design phase, the application of the proposed methodology allowed the definition of a precise and complete waste audit and digital material inventory, which supported the planning and optimization of the selective demolition phase. Closerange photogrammetry for field surveys was the easiest and safest survey technique considering the abandoned status and the height of existing facilities [52], allowing for a limited number of on-site surveys and adding the possibility of in-depth analysis in remote mode.

Figure 13 compares CDW management and demolition phase planning in a traditional process and considering the proposed methodology. The use of a digital database as a material inventory promoted always available and up-to-date data that were organized and structured in a machine-readable form, thus ensuring data readiness and processability and agile data management [80,81]. The case study confirmed that the definition of a waste audit allowed to [5]:

- Support the definition of reuse and recycling practices and the higher quality and easier traceability of waste through the proper identification and separation of materials;
- Plan selective demolition activities with their associated costs of execution;
- Ensure an unbiased competition amongst the participants in the call for tenders that could depend on reliable information regarding existing site and building conditions and demolition costs, allowing them to propose appropriate construction site plans.

# Waste audit and digital material inventory definition

- Quantification and classification of waste materials for waste collection and transport to authorized plants
- Precise planning of selective demolition
  activities
- Identification of reuse strategies for waste materials and elements
- Identification of authorized recycling plants and minimization of their distance from the construction site
- Identification of selective demolition costs to be included in tender documents

# Traditional process (without definition of waste audits)

- Definition of non-selective demolition costs, that are lower than selective demolition ones by about 25–30%
- Possible use of parametric demolition costs to define total non-selective demolition costs (but lower accuracy than selective demolition costs defined through digital material inventory)
- Not possible identification of waste quantities for reuse, recycle or reuse strategies
- Increasing uncertainties in the definition of demolition plans

**Figure 13.** Comparison between the proposed methodology with waste audit and digital material inventory definition, and a traditional process.

The selective demolition costs represented 127% of the estimated non-selective demolition ones, clearly highlighting the importance of applying proper waste audits and selective demolition planning. As a matter of fact, if lower non-selective demolition costs had been considered during the call for tenders phase, the actual increased selective demolition costs would have only been identified during the actual demolition phase. This would have caused the necessity of increasing the project budget or could have resulted in disputes between the Public Client and the company, causing time delays or increased costs. Therefore, the correct identification of selective demolition costs is paramount to ensure the proper development of the tender and construction phases.

Considering the design phase, the creation and use of a BIM model of the designed buildings allowed the reduction of construction waste and the promotion of waste minimization during design reviews and clash detections between design disciplines, designers, and specialists [39,42]. In addition, the possibility of defining on-site reuse strategies, in particular by using concrete wastes as secondary raw materials, resulted in savings for the purchase of a considerable amount of new raw materials, with positive results both regarding the project budget and the environmental aspects, fostering local circular economy practices, and demonstrating the concept of waste as a resource [23,24].

Concerning the call for tenders, the proposed methodology enabled to:

- Support of the Public Client in including sustainability criteria for the evaluation and scoring of the bids and in analyzing the environmental impact of the bids, while acquiring the necessary competences for proper GPP implementation [11];
- Promotion of the integration of CDW management and minimization, sustainable resource use, and selective demolition procedures among the participants.

The high scores of 10 out of 11 bids in the environmentally sustainable award criteria included in the call for tenders demonstrated that:

- All of the bids and the participants recognized the importance of the optional award criteria and decided to gain competences and know-how on the topic;
- The well-defined award criteria supported by the digital material inventory and by the BIM-based design data led the participants to more easily apply and integrate sustainable practices in their procedures.

Regarding the actual winning bid (Figure 14), the environmental factors represented a fundamental part of the offer since numerous points had been assigned to the company on

the environmentally sustainable award criteria, i.e., 25.7/100 total points and the 95% of the total available points regarding the above-mentioned award criteria. In particular, the detailed tender documents supported the selective demolition planning as a critical part of the project by assigning numerous points to correct demolition planning and proper CAM application. In addition, the precise identification of waste materials quantity and information in the waste audit supported the maximization of the quantities of recycled and reused waste materials and the proper selection of reuse strategies by the company. Furthermore, the company was able to select authorized recycling and landfilling plants, minimizing the distance from the construction site to reduce emissions.



Figure 14. Graph of relative weight of environmental sub-criteria and graph of the scores obtained by the winning bid.

The Design Build (DB) procurement model allowed the contractor to manage the process flow and information right from the final design stage [84], enabling the optimization of CDW management and selective demolition plans, and the decrease of CDW production, which in a traditional tender process would have been linked to design incompleteness or reworking [42–44]. In addition, the contractor's involvement in the final design phase avoided the increased risk of waste generation that typically occurs in the construction phase of a traditional construction process [45].

### 9. Conclusions

The construction sector produces over 30% of the total amount of waste by volume in the European Union, being one of the most polluting industrial sectors. In addition, since 2004, the amount of construction waste has increased of about 20%, while other industrialized sectors have reduced the waste generation of over 20% [4]. Consequently, the adoption of Construction and Demolition Waste (CDW) management and minimization strategies is paramount. The correct definition of selective demolition plans and strategies is also a fundamental step for proper CDW management when considering projects that involve extensive demolitions.

A review of the regulatory framework at the international, European, and national levels highlighted that environmentally sustainable processes, waste minimization and management, and the implementation of Green Public Procurement (GPP) are needed and are promoted by the United Nations, the European Union, and national governments.

The main drivers for the integration of environmentally sustainable strategies in the construction industry and to overcome the barriers are the following:

 Promoting sustainable strategies by regulations and by Public Clients through design requirements and call for tenders criteria;

- Integrating sustainable strategies into company business models and procedures as a way to maintain and increase their competitiveness;
- Introducing Information Modelling and Management (IMM) methodologies to support the implementation of sustainable strategies.

The literature review on the integration of IMM and BIM methodologies and waste management highlighted that existing studies have mostly investigated these aspects from the designer and constructor's points of view. Few studies have analyzed the integration of BIM and GPP in construction procurement to evaluate the environmental impact of the bids. However, a comprehensive application of IMM methods and GPP from the Public Client's point of view aiming to evaluate and reduce the building environmental impact during the design and call for tenders phases, represents a gap in the literature.

The research proposes a methodology for the preliminary design phase through the definition of a digital material inventory as a fundamental part of waste audits for selective demolition planning and CDW management by adopting close-range photogrammetry to safely perform field surveys of to-be-demolished abandoned buildings. During the design phase, a BIM model is created for the designed buildings to support CDW management and on-site reuse strategies selection. Regarding the call for tenders phase, the methodology involved the definition of environmental award criteria in a MEAT framework, enabling the introduction of GPP and CDW management in the tender process.

The methodology was tested on the case study of a brownfield renovation that included extensive demolitions and the construction of a new school complex in the Province of Milan.

The IMM approach, i.e. the digital material inventory for the preliminary design phase, and the BIM model for the design and tender activities, permitted to maintain consistency during the entire process. The selective demolition costs calculated through the digital material inventory and the demolition plans represented 127% of the estimated non-selective demolition ones. The correct identification of the demolition costs represents a fundamental aspect that ensures that the whole procedure, involving the tendering, project assignment, and the subsequent demolition and new construction phases, can be conducted without resulting in disputes, delays, and budget increases.

The demand of the call for tenders was highly focused and accurate, hence ensuring that the offers met the requirements and complied with environmental strategies, including improvements and considerations about on-site reuse strategies, CDW management, and selective demolition plans. In general, the combined use of IMM approaches for the selective demolition phase and demolition waste management, and of the BIM methods for the designed buildings, ensured effective CDW management and the precise identification of reuse and recycling strategies. The demand guided and pushed the application of sustainable practices by the construction companies that recognized the importance and value of introducing sustainable strategies in their own business models. In addition, the method allowed the Public Client to assess the environmental impact of the bids. Furthermore, the Design Build (DB) procurement model enabled an increased collaboration between the Client and the constructor and a less fragmented and more efficient information flow. Consequently, it allowed for the better implementation of CDW management and minimization strategies by almost all of the participants. Furthermore, the precise CDW management during the design, tender, and construction phases will support CDW management during the operational phase of the building, promoting proper reusing and recycling strategy application during the whole building lifecycle.

The research highlighted the critical role of the Public Client as the actor that can trigger a change in the construction sector by implementing GPP and applying best practices during all the phases, supporting a sustainable construction process. In addition, the study demonstrated the positive effects and results of the model application for brownfield recovery.

The proposed methodology was previously tested with positive results in another case study of the new primary school in Melzo, also using the DB procurement model,

but with a simpler design and without almost any demolition [12]. The positive results of the presented case study demonstrate the replicability of the methodology since this case study was a brownfield renovation, with a more complex design, an increased size of the new buildings, and included extensive demolitions, resulting in more complex CDW management, selective demolition planning and sustainable resource use.

# **10. Further Developments**

Further developments of this research will include the use of distributed ledger technology; in particular, the implementation of Smart Contracts based on blockchain technology can be proposed. Thanks to its main features, the use of a distributed ledger can improve information management, and the execution of a Smart Contract can guarantee the immutable and reliable recording of all of the information. The potential integration of a blockchain and IMM methodology can boost the information production, validation, monitoring, and management, offering a transparent, permanent, and shared archiving of the information on a distributed ledger.

For this research case, the proposal of this innovative system permits the proper traceability of all the information related to the materials and the waste cycles pursued in the process. Consequently, this application enables better communication and collaboration among the participants acting in each phase of the construction process, from the planning to the disassembly of the building. The use of a distributed ledger and Smart Contracts in the management of the information related to waste production and management enables the recording and tracking of all the of the relevant data, assuring their access at any time in a reliable way. This means that an IMM based on a blockchain can be considered as a trusty base for the decision-making process, allowing the proper planning and management of construction waste and the environmental impact of the entire process. Based on real and reliable information, more sustainable decision and planning can be conducted in the entire construction industry.

As stated before, despite the fact that some environmental sustainability protocols are mandatory (i.e. CAM in Italy) and that others often provide additional scores in MEAT approaches (e.g. LEED-Leadership in Energy and Environmental Design, ITACA, etc.), an objective and reliable assessment of their compliance is still struggling to be obtained. This is mainly due to the intrinsic complexity and the lack of adequate assessing digital tools. Thus, another development of the outlined IMM approach will concern the introduction of automated sustainability criteria evaluation methods through the integrated simulation of Information Models and automated score attribution systems. This could provide the dual objective of increasing GPP adoption and enhancing digitalization in the tender evaluation processes as required by European Directives. The achievable goals consist of the shortening of the tendering phase, cost reduction as well as avoidance of controversies and corruption. In addition, this will ensure the objective and trustworthy evaluation of sustainability criteria compliance based on the actual use of the building rather than on rough saving assessments. On the contrary, bids will be evaluated through well-defined, clear, and machine-readable criteria in order to digitalize and automate the evaluation process.

Author Contributions: Conceptualization, L.P. and E.S.; methodology, L.P. and M.L.; software, M.L.; validation, L.P., L.C.T., and G.M.D.G.; formal analysis, M.L.; investigation, L.P.; resources, G.P.; data curation, M.L.; writing—original draft preparation, L.P. and M.L.; writing—review and editing, L.C.T.; visualization, S.M.; supervision, L.C.T.; project administration, G.M.D.G.; funding acquisition, G.M.D.G. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

**Institutional Review Board Statement:** Ethical review and approval were waived for this study because they were not applicable.

**Informed Consent Statement:** Informed consent was obtained from all subjects involved in the study.
Data Availability Statement: The data are not publicly available due to privacy reasons.

Acknowledgments: This work was supported by the ABCLab research unit of the BIMGroup for the Digital Transition in AECO sector of the Department of Architecture, Built Environment, and Construction Engineering, Politecnico di Milano. In particular, the authors want to thank Francesco Paleari and Marco Schievano for the development of the case study and for their support of this research project.

Conflicts of Interest: The authors declare no conflict of interest.

#### Glossary

AECO: Architecture, Engineering, Construction, and Operations industry. BIM: Building Information Modelling. A BIM method refers to digital techniques used for the modelling and management of information for a built asset and specifically refers to the creation of a digital model of a building. CDW: Construction and Demolition Waste. DB: Design Build. The DB procurement model merges the actors responsible for final design and the construction into a single operator. The Client dialogues with a unique actor, increasing the efficiency of information transfer [84]. EWC: European Waste Code. GPP: Green Public Procurement. GPP aims to purchase goods with good value for money during the whole product lifecycle while supporting environmental protection and reducing negative environmental impacts [14]. IMM: Information Modelling and Management. The IMM approach refers to digital techniques to create and manage information throughout an entire process, allowing for optimized management, organization, querying, computing, and analyzing capabilities. LEED: Leadership in Energy and Environmental Design. LEED is a green building rating system that is used worldwide. It provides a framework for healthy, highly efficient, and cost-saving green and sustainable buildings [89]. MEAT: Most Economically Advantageous Tender. The MEAT is assessed by organizations on the basis of the best price quality ratio, aiming to achieve value for money on a building lifecycle basis. Contracts are awarded based on both quality and price by defining award criteria linked to the subject matter of the contract [18].

#### References

- 1. Herczeg, M.; McKinnon, D.; Milios, L.; Bakas, I.; Klaassens, E.; Svatikova, K.; Widerberg, O. *Resource Efficiency in the Building Sector*; ECORYS: Rotterdam, The Netherlands, 2014.
- 2. Council of The European Union. *Circular Economy in the Construction Sector;* Council of The European Union: Bruxelles, Belgium, 2019.
- Matthews, E.; Amann, C.; Bringezu, S.; Fischer-Kowalski, M.; Huttler, W.; Kleijn, R.; Moriguchi, Y.; Ottke, C.; Rodenburg, E.; Rogich, D.; et al. *The Weight of Nations. Material Outflows from Industrial Economies*; Hutter, C., Ed.; World Resources Institute: Washington, DC, USA, 2000; ISBN 1569734399.
- 4. Eurostat Waste Statistics. Available online: https://ec.europa.eu/eurostat/statistics-explained/index.php?title=Waste\_statistics# Waste\_generation\_excluding\_major\_mineral\_waste (accessed on 14 May 2021).
- European Commission. Guidelines for the Waste Audits before Demolition and Renovation Works of Buildings. UE Construction and Demolition Waste Management; European Commission: Bruxelles, Belgium, 2018.
- Osmani, M. Construction Waste. In Waste: A Handbook for Management; Academic Press: Cambridge, MA, USA, 2011; ISBN 9780123814753.
- 7. United Nations. Paris Agreement; United Nations: New York, NY, USA, 2015.
- 8. Chong, H.Y.; Lee, C.Y.; Wang, X. A mixed review of the adoption of Building Information Modelling (BIM) for sustainability. *J. Clean. Prod.* 2017, 142, 4114–4126. [CrossRef]
- 9. Barbini, A.; Malacarne, G.; Romagnoli, K.; Massari, G.A.; Matt, D.T. Integration of life cycle data in a BIM object library to support green and digital public procurements. *Int. J. Sustain. Dev. Plan.* **2020**, *15*, 983–990. [CrossRef]
- 10. United Nations Environment Programme. *Sustainable Public Procurement Implementation Guidelines*; United Nations Environment: New York, NY, USA, 2012.
- 11. Osservatorio Appalti Verdi. I numeri del Green Public Procurement in Italia; Legambiente: Rome, Italy, 2020.
- 12. Pellegrini, L.; Campi, S.; Locatelli, M.; Pattini, G.; Di Giuda, G.M.; Tagliabue, L.C. Digital Transition and Waste Management in Architecture, Engineering, Construction, and Operations Industry. *Front. Energy Res.* **2020**, *8*, 1–21. [CrossRef]
- 13. International Organization for Standardization. International Standard ISO 20400—Sustainable Procurement—Guidance; ISO: Geneva, Switzerland, 2017.
- 14. United Nations Environment Programme. *ABC of SCP: Clarifying Concepts on Sustainable Consumption and Production;* United Nations Environment: New York, NY, USA, 2010.

- 15. United Nations Environment Programme. *Global Environment Outlook (GEO-6): Healthy Planet, Healthy People;* United Nations Environment: New York, NY, USA, 2019.
- 16. European Commission. The European Green Deal; European Commission: Bruxelles, Belgium, 2019.
- 17. The European Parliament; The Council of the European Union. *Directive 2014/24/EU of The European Parliament and of The Council of 26 February 2014 on Public Procurement and Repealing Directive 2004/18/EC (Text with EEA Relevance)*; Council of The European Union: Bruxelles, Belgium, 2014; p. 178.
- 18. Italian Parliament and Government. Decreto Legislativo 18 Aprile 2016, n. 50 Codice dei Contratti Pubblici; Italian Parliament and Government: Rome, Italy, 2016.
- 19. Organization for Economic Co-Operation and Development. *Resource Productivity in the G8 and the OECD. A Report in the Framework of the Kobe 3R Action Plan;* OECD: Paris, France, 2011.
- 20. United Nations. Transforming Our World: The 2030 Agenda for Sustainable Development; United Nations: New York, NY, USA, 2015.
- 21. Lindhqvist, T.; Lidgren, K. "Modeller för förlängt producentansvar" ("Models for Extended Producer Responsibility," in Swedish). In Från Vaggan Till Graven—Sex Studier av Varors Miljöpåverken" ("From the Cradle to the Grave—Six Studies of the Environmental Impact of Products," in Swedish); Ministry of the Environment: Stockholm, Sweden, 1991.
- 22. United Nations Environment Programme. *Global Waste Management Outlook;* United Nations Environment: New York, NY, USA, 2016.
- 23. The European Parliament; The Council of the European Union. *Directive 2018/851 Amending Directive 2008/98/EC on Waste Framework*; Council of The European Union: Bruxelles, Belgium, 2018.
- 24. European Commission. *COM*(2015) 614 Final. Closing the Loop—An EU Action Plan for the Circular Economy; European Commission: Bruxelles, Belgium, 2015.
- 25. Big Buyers Initiatives. *Public Procurement of Circular Construction Materials. Key takeaways from the Big Buyers Initiative Working Group;* Eurocities: Bruxelles, Belgium, 2020.
- 26. Italian Parliament and Government. Decreto Ministeriale 11 Ottobre 2017 Criteri Ambientali Minimi per L'affidamento di Servizi di Progettazione e Lavori per la Nuova Costruzione, Ristrutturazione e Manutenzione di Edifici Pubblici; Italian Parliament: Rome, Italy, 2017.
- 27. Ferber, U.; Grimski, D. Brownfields and Redevelopment of Urban Areas; Grimski, D., Lowe, J., Smith, S., Ferber, U., Eds.; Umweltbundesamt GmbH (Federal Environment Agency Ltd.): Wien, Austria, 2002.
- Sistema Nazionale per la Protezione dell'Ambiente (SNPA). Consumo di suolo, dinamiche territoriali e servizi ecosistemici. In Report SNPA 08/19; Munafò, M., Ed.; Digital Print Store s.r.l.: Roma, Italy, 2019; pp. 1–224. ISBN 9788844809645.
- 29. United Nations. United Nations Conference on Sustainable Development: "The Future We Want"; United Nations: Rio de Janeiro, Brazil, 2012.
- 30. European Commission. *Living Well, within the Limits of Our Planet 7th EAP—The New General Union Environment Action Programme to 2020;* European Commission: Bruxelles, Belgium, 2013.
- 31. European Commission. Communication from the Commission to the Council, the European Parliament, the Economic and Social Committee and the Committee of the Regions: "Towards a Thematic Strategy for Soil Protection"; European Commission: Bruxelles, Belgium, 2002.
- 32. European Commission. *Proposal for a Directive of the European Parliament and of the Council Establishing a Framework for the Protection of Soil and Amending Directive 2004/35/EC;* European Commission: Bruxelles, Belgium, 2006.
- 33. European Commission. *Guidelines on Best Practice to Limit, Mitigate or Compensate Soil Sealing;* European Commission: Bruxelles, Belgium, 2012.
- 34. Italian Parliament and Government. Strategia Nazionale per lo Sviluppo Sostenibile; Italian Parliament: Rome, Italy, 2017.
- Passalacqua, M.; Favaro, T. Rigenerare siti industriali dismessi attraverso un «sistema» giuridico incentivante. In *Ri-Conoscere la Rigenerazione. Strumenti Giuridici e Tecniche Urbanistiche*; Passalacqua, M., Fioritto, A., Rusci, S., Eds.; Maggioli: Santarcangelo di Romagna, Italy, 2018; pp. 61–90. ISBN 9788891628633.
- 36. Regione Lombardia. Legge Regionale 28 Novembre 2014, n. 31 "Disposizioni per la Riduzione del Consumo di Suolo e per la Riqualificazione del Suolo Degradato"; Regione Lombardia: Milan, Italy, 2014; p. 9.
- Pietra, S. Il fenomeno dei brownfields. In Proposta di Linee Guida per il Recupero Ambientale e la Valorizzazione Economica dei Brownfields; APAT—Agenzia per la protezione dell'ambiente e per i servizi, Ed.; I.G.E.R. srl: Rome, Italy, 2006; pp. 20–27. ISBN 8844802198.
- 38. Teo, M.M.M.; Loosemore, M. A theory of waste behaviour in the construction industry. *Constr. Manag. Econ.* **2001**, *19*, 741–751. [CrossRef]
- 39. Ajayi, S.O.; Oyedele, L.O.; Akinade, O.O.; Bilal, M.; Owolabi, H.A.; Alaka, H.A.; Kadiri, K.O. Reducing waste to landfill: A need for cultural change in the UK construction industry. *J. Build. Eng.* **2016**, *5*, 185–193. [CrossRef]
- 40. Dainty, A.; Green, S.; Bagilhole, B. People and Culture in Construction: A Reader; Routledge: Abingdon, UK, 2007; ISBN 1134274653.
- 41. Koskela, L. Making-do—The eighth category of waste. In Proceedings of the 12th Annual Conference of the International Group for Lean Construction, Helsingor, Denmark, 3–5 August 2004; Volume 10.
- Adjei, S.D.; Ankrah, N.A.; Ndekugri, I.; Searle, D. Sustainable construction and demolition waste management: Comparison of corporate and project level drivers. In Proceedings of the 34th Annual ARCOM Conference ARCOM 2018, Belfast, UK, 3–5 September 2018; pp. 99–108.

- 43. Deborah, H.; Trefor, W.; Zhaomin, R. Differing perspectives on collaboration in construction. *Constr. Innov.* **2012**, *12*, 355–368. [CrossRef]
- 44. Osmani, M. Construction Waste Minimization in the UK: Current Pressures for Change and Approaches. *Procedia Soc. Behav. Sci.* **2012**, *40*, 37–40. [CrossRef]
- 45. Arain, F.M.; Assaf, S.; Pheng, L.S. Causes of Discrepancies between Design and Construction. *Archit. Sci. Rev.* 2004, 47, 237–249. [CrossRef]
- 46. Osmani, M.; Glass, J.; Price, A.D.F. Architects' perspectives on construction waste reduction by design. *Waste Manag.* 2008, 28, 1147–1158. [CrossRef]
- 47. Fewings, P.; Henjewele, C. Construction Project Management: An Integrated Approach; Routledge: London, UK, 2019; ISBN 9781351122030.
- 48. Fairclough, J. Rethinking Construction Innovation and Research—A Review of the Government's R&D Policies and Practices; Department of Trade and Industry: London, UK, 2002.
- 49. Dai, F.; Lu, M. Assessing the Accuracy of Applying Photogrammetry to Take Geometric Measurements on Building Products. J. Constr. Eng. Manag. 2010, 136, 242–250. [CrossRef]
- 50. Nebiker, S.; Bleisch, S.; Christen, M. Rich point clouds in virtual globes—A new paradigm in city modeling? *Comput. Environ. Urban Syst.* **2010**, *34*, 508–517. [CrossRef]
- 51. Luhmann, T.; Robson, S.; Kyle, S.; Harley, I. *Close Range Photogrammetry. Principles, Techniques and Applications*; Whittles: Dunbeath, UK, 2006; ISBN 9780470106334.
- Chalioris, C.E.; Tsioukas, V.E.; Favvata, M.J.; Karayannis, C.G. Recording Historic Masonry Buildings Using Photogrammetry-Two Case Studies. In Proceedings of the ECCOMAS Thematic Conference on Computational Methods in Structural Dynamics and Earthquake Engineering, Kos Island, Greece, 12–14 June 2013; pp. 1401–1409.
- 53. Aguilera, D.G.; Lahoz, J.G. Laser scanning or image-based modeling? A comparative through the modelization of San Nicolas church. *Int. Arch. Programm. Remote Sens.* **2006**, *36*, B5.
- Kolecka, N. Photo-Based 3D Scanning Vs. Laser Scanning—Competitive Data Acquisition Methods for Digital Terrain Modelling of Steep Mountain Slopes. ISPRS Int. Arch. Photogramm. Remote Sens. Spat. Inf. Sci. 2012, 38, 203–208. [CrossRef]
- 55. Kersten, T.P. 3D Point Clouds through Image-Based Low-Cost Systems; HCU: Hamburg, Germany, 2012; 41p.
- 56. Macher, H.; Landes, T.; Grussenmeyer, P. From point clouds to building information models: 3D semi-automatic reconstruction of indoors of existing buildings. *Appl. Sci.* 2017, 7, 1030. [CrossRef]
- 57. Chen, L.; Lu, Q.; Zhao, X. A semi-automatic image-based object recognition system for constructing as-is IFC BIM objects based on fuzzy-MAUT. *Int. J. Constr. Manag.* **2019**, 1–15. [CrossRef]
- Kokorus, M.; Pizarro, F.; Eyrich, W.; Heuser, S. From Optical Symbol Recognition (OSR) of Point Clouds to the Substation Information Model. In Proceedings of the 2018 IEEE/PES Transmission and Distribution Conference and Exposition (T&D), Denver, CO, USA, 16–19 April 2018; pp. 6–9. [CrossRef]
- Ahankoob, A.; Abbasnejad, B.; Wong, P.S.P. The Support of Continuous Information Flow Through Building Information Modeling (BIM). In *Proceedings of the The 10th International Conference on Engineering, Project, and Production Management*; Panuwatwanich, K., Ko, C., Eds.; Springer Nature: Singapore, 2020; pp. 125–138.
- 60. Russell, A.; Staub-French, S.; Tran, N.; Wong, W. Visualizing high-rise building construction strategies using linear scheduling and 4D CAD. *Autom. Constr.* 2009, *18*, 219–236. [CrossRef]
- Wu, Y.; Xu, N. BIM information collaborative framework based on supply chain management. In Proceedings of the International Conference on Construction and Real Estate Management; American Society of Civil Engineers (ASCE): Kunming, China, 2014; pp. 199–2017.
- 62. Zhang, S.; Teizer, J.; Lee, J.K.; Eastman, C.M.; Venugopal, M. Building Information Modeling (BIM) and Safety: Automatic Safety Checking of Construction Models and Schedules. *Autom. Constr.* **2013**, *29*, 183–195. [CrossRef]
- 63. Cheng, J.C.P.; Ma, L.Y.H. A BIM-based system for demolition and renovation waste estimation and planning. *Waste Manag.* 2013, 33, 1539–1551. [CrossRef] [PubMed]
- 64. Yuan, H.; Shen, L. Trend of the research on construction and demolition waste management. *Waste Manag.* **2011**, *31*, 670–679. [CrossRef] [PubMed]
- 65. Jalali, S. Quantification of Construction Waste Amount. In Proceedings of the 6th International Technical Conference of Waste, Viseu, Portugal, October 2007.
- Cheng, J.C.P.; Won, J.; Das, M. Construction and demolition waste management using bim technology. In Proceedings of the IGLC 23—23rd Annual Conference of the International Group for Lean Construction: Global Knowledge—Global Solutions, Perth, Australia, 29–31 July 2015; pp. 381–390.
- Akinade, O.O.; Oyedele, L.O.; Bilal, M.; Ajayi, S.O.; Owolabi, H.A.; Alaka, H.A.; Bello, S.A. Waste minimisation through deconstruction: A BIM based Deconstructability Assessment Score (BIM-DAS). *Resour. Conserv. Recycl.* 2015, 105, 167–176. [CrossRef]
- 68. Kibert, C.J. Sustainable Construction: Green Building Design and Delivery, 4th ed.; John Wiley & Sons: Hoboken, NJ, USA, 2016; ISBN 1119055172.
- 69. Guy, B.; Shell, S.; Homsey, E. Design for Deconstruction and Materials Reuse. In *Design for Deconstruction and Materials Reuse;* Inhouse Publishing: Rotterdam, The Netherlands, 2002; p. 21.

- 70. Tingley, D.D.; Davison, B. Developing an LCA methodology to account for the environmental benefits of design for deconstruction. *Build. Environ.* **2012**. [CrossRef]
- 71. Akbarnezhad, A.; Ong, K.C.G.; Chandra, L.R. Economic and environmental assessment of deconstruction strategies using building information modeling. *Autom. Constr.* **2014**, *37*, 131–144. [CrossRef]
- 72. Addis, B.; Jenkins, O. Briefing: Design for deconstruction. In Proceedings of the Institution of Civil Engineers: Waste and Resource Management; ICE Publishing: London, UK, 2008; Volume 161, pp. 9–12.
- 73. Liu, Z.; Osmani, M.; Demian, P.; Baldwin, A. A BIM-aided construction waste minimisation framework. *Autom. Constr.* 2015, 59, 1–23. [CrossRef]
- 74. Guerra, B.C.; Bakchan, A.; Leite, F.; Faust, K.M. BIM-based automated construction waste estimation algorithms: The case of concrete and drywall waste streams. *Waste Manag.* **2019**, *87*, 825–832. [CrossRef] [PubMed]
- 75. Guerra, B.C.; Leite, F.; Faust, K.M. 4D-BIM to enhance construction waste reuse and recycle planning: Case studies on concrete and drywall waste streams. *Waste Manag.* 2020, *116*, 79–90. [CrossRef] [PubMed]
- 76. Won, J.; Cheng, J.C.P.; Lee, G. Quantification of construction waste prevented by BIM-based design validation: Case studies in South Korea. *Waste Manag.* 2016, 49, 170–180. [CrossRef]
- 77. Palmujoki, A.; Parikka-Alhola, K.; Ekroos, A. Green public procurement: Analysis on the use of environmental criteria in contracts. *Rev. Eur. Community Int. Environ. Law* **2010**, *19*, 250–262. [CrossRef]
- 78. Wong, J.K.W.; Chan, J.K.S.; Wadu, M.J. Facilitating effective green procurement in construction projects: An empirical study of the enablers. *J. Clean. Prod.* **2016**, *135*, 859–871. [CrossRef]
- 79. Hossain, M.A.; Yeoh, J.K.W. BIM for Existing Buildings: Potential Opportunities and Barriers. In *IOP Conference Series: Materials Science and Engineering*; IOP Publishing: Bristol, UK, 2018; Volume 371, pp. 1–9. [CrossRef]
- 80. Chen, K.; Lu, W.; Peng, Y.; Rowlinson, S.; Huang, G.Q. Bridging BIM and building: From a literature review to an integrated conceptual framework. *Int. J. Proj. Manag.* 2015, *33*, 1405–1416. [CrossRef]
- 81. Flanagan, R.; Lu, W. Making informed decisions in product-service systems. In *Proceedings of the IMechE Conference, Knowledge and Information Management Through-Life*; Institute of Mechanical Engineers: London, UK, 2008.
- 82. Niu, Y.; Lu, W.; Liu, D.; Chen, K.; Anumba, C.; Huang, G.G. An SCO-enabled logistics and supply chain management system in construction. *J. Constr. Eng. Manag.* 2016, 143, 04016103. [CrossRef]
- 83. Lu, W.; Webster, C.; Chen, K.; Zhang, X.; Chen, X. Computational Building Information Modelling for construction waste management: Moving from rhetoric to reality. *Renew. Sustain. Energy Rev.* 2017, 68, 587–595. [CrossRef]
- 84. Beard, J.L.; Loulakis, M.C.; Wundram, E.C. *Design-Build: Planning through Development*; McGraw-Hill Education: New York, NY, USA, 2001; ISBN 978-0070063112.
- 85. Eastman, C.M.; Teicholz, P.M.; Sacks, R.; Liston, K. BIM Handbook. A Guide to Building Information Modeling for Owners, Designers, Engineers, Contractors, and Facility Managers; John Wiley & Sons: New York, NY, USA, 2011; ISBN 978-0470541371.
- 86. Comune di Inveruno. Linee Guida alla Compilazione Dell'offerta Tecnica; Comune di Inveruno: Milano, Italy, 2021.
- 87. International Organization for Standardization. International Standard ISO 45001; ISO: Geneva, Switzerland, 2018.
- 88. British Standards Institution. BS OHSAS 18001; BSI: London, UK, 1999.
- 89. U.S. Green Building Council. *LEED v4.1 Building Design and Construction*; U.S. Green Building Council: Washington, DC, USA, 2021.
- 90. Brem, A.; Cusack, D.Ó.; Adrita, M.M.; O'Sullivan, D.T.J.; Bruton, K. How do companies certified to ISO 50001 and ISO 14001 perform in LEED and BREEAM assessments? *Energy Effic.* **2020**, *13*, 751–766. [CrossRef]
- 91. Ongpeng, J. Environmental Performance Assessment on High-Rise Building Projects in Taguig and Makati Using LEED and ISO 14001; ResearchGate: Berlin, Germany, 2015. [CrossRef]





# Article Limits of the Environmental Impact Assessment (EIA) in Malaysia: Dam Politics, Rent-Seeking, and Conflict

Peter Ho<sup>1</sup>, Bin Md Saman Nor-Hisham<sup>2,\*</sup> and Heng Zhao<sup>1</sup>

- <sup>1</sup> Zijingang Campus, School of Public Affairs, Zhejiang University, Hangzhou 310058, China; peter\_ho@zju.edu.cn (P.H.); zhaoz5755@gmail.com (H.Z.)
- <sup>2</sup> Department of Town and Regional Planning, Universiti Teknologi Mara, Cawangan Perak 32610, Malaysia
- \* Correspondence: norhi559@uitm.edu.my

Received: 6 November 2020; Accepted: 7 December 2020; Published: 14 December 2020

Abstract: Environmental Impact Assessment (EIA) is often portrayed as a policy measure that can mitigate the environmental influence of corporate and government projects through objective, systematic, and value-free assessment. Simultaneously, however, research has also shown that the larger political context in which the EIA is embedded is crucial in determining its influence on decision-making. Moreover, particularly in the case of mega-projects, vested economic interests, rent-seeking, and politics may provide them with a momentum in which the EIA risks becoming a mere formality. To substantiate this point, the article examines the EIA of what is reportedly Asia's largest dam outside China: the Bakun Hydro-electric Project (BHP) in Malaysia. The study is based on mixed methods, particularly, qualitative research (semi-structured interviews, participatory observation, and archival study) coupled to a survey conducted in 10 resource-poor, indigenous communities in the resettlement area. It is found that close to 90% of the respondents are dissatisfied with their participation in the EIA, while another 80% stated that the authorities had conducted the EIA without complying to the procedures. The findings do not only shed light on the manner in which the EIA was used to legitimize a project that should ultimately have been halted, but are also testimony to the way that the BHP has disenfranchised the rights of indigenous people to meaningfully participate in the EIA.

**Keywords:** environmental and social impact assessment; dam and mega-projects; forced displacement and resettlement; land eviction and expropriation; Borneo and Orang Ulu; first nations and ethnic minorities

# 1. Introduction

Incorporating the Environmental Impact Assessment (EIA) into existing planning and decision-making processes is generally put forward as a means to identify potentially adverse effects of proposed (mega-) projects. The EIA can provide the information for more grounded decisions on how to proceed, and ensure the project's environmental sustainability, economic viability, and social acceptability [1–4]. At the same time, however, research has ascertained a divergence between what is aimed for by the EIA and what is practiced and enforced [5].

Various studies have shown that the EIA can become a political tool whereby the decision to approve projects is outweighed by other reasons than technical or environmental considerations [6]. In this context, the EIA may merely serve to legitimatize decisions which, in fact, have already been taken [7,8]. In effect, the EIA exists as a symbolic token decoupled from realities on the ground, and with limited influence on the decision-making to approve, alter, or even reject projects [9–12].

This article acknowledges that improvement on certain shortcomings of the EIA could be achieved through procedural, methodological, and technical ways [13–16]. At the same time, however, it also

moots that a decisive factor in the success or failure of EIA is constituted by the larger socio-political context in which it takes place. In fact, the politics and rent-seeking over EIA, by which public policies and/or economic conditions are manipulated as strategies to raise profits or personal gains, are prime factors that need reflection prior to considering its execution. Put differently, when the environment in which the EIA is embedded cannot safeguard its independence, it will be extremely difficult to conduct in an impartial manner. It needs emphasis that this presupposes that the EIA practitioner is not part and parcel of the politics and rent-seeking itself, which may not be necessarily the case [17]. To substantiate the argument, this article presents the case of the EIA of a dam, which at 205 m high, is reportedly Asia's second largest dam: the Bakun Hydro-electric Project (BHP) in Malaysia.

The bulk of the research on the BHP has focused on its impact on energy supply [18,19], its policies on compensation [20], and its social sustainability [21]. With regard to the latter, Andre's in-depth study [22] duly noted that social sustainability for mega-projects, such as the BHP, need to move beyond formal lists of social indicators. Instead, these should be assessed through evolutionary, qualitative, and hermeneutic approaches that enable the identification of social issues of concern. To date, however, few studies have specifically examined the EIA of the BHP, which is a lost opportunity, as the EIA was the main tool through which the Malaysian government tried to legitimize it.

The study by Memon [23] is one of the few exceptions that squarely looks into the EIA of the BHP. Although done with great care and depth, his study was carried out well over a decade before the dam went into operation. In this context, there is a need to follow-up and update Memon's seminal study, which we aim to do in various ways: (1) by complementing an ex ante analysis of the EIA, with an ex post analysis conducted after the dam went into operation; (2) by not only shedding light on the way that the EIA was conducted, but also how it was experienced by the affected population—the indigenous Orang Ulu; (3) by not only achieving this in a qualitative way (through interviews, participatory observation, and archival study) as done by Memon, but also in a quantitative way (through a survey conducted in 10 rural communities in the resettlement area).

In this respect and to our best knowledge, the article is one of the first studies of this nature on the Bakun Dam. In addition, through the comprehensive, mixed methodology adopted here, this case might also enhance our general understanding of the dynamics and politics of the EIA around dams and other large infrastructural projects. The article is structured around a dual research question: (1) How and to which extent did politics and rent-seeking around the EIA influence project implementation? (2) How did these politics and rent-seeking, in turn, influence the views and experiences of the affected population with regard to the EIA?

Apart from the introduction, this article is divided into four main sections. In the second section, we will provide an in-depth description of: (1) the BHP's basic features; (2) the history of the EIA in Malaysia; (3) the vested economic interests that propelled the initiation of the BHP, and; (4) the politics and rent-seeking that surrounded this mega-project's EIA. In the third section, we will introduce the methodology, features of the survey sample and interviews, and describe the research sites in terms of their socio-economic, geographical, and demographic conditions. This is followed by the results section, where we analyze the survey data and qualitative fieldwork in terms of respondents' views on the EIA, with particular reference to public participation, satisfaction, conflict management, and trust. In the fourth and concluding section, we discuss the empirical and theoretical implications of the case for the practice of EIA.

# 2. EIA of the Bakun Dam: Controversy, Conflict, and Colluding Interests

# 2.1. Project Features

The BHP is situated in Sarawak State, also known as "Land of the Hornbills," the biggest state within the Malaysian federation of a total of 13 states (see Figure 1). The Malaysian government initially approved the construction of the BHP in 1986, which was Asia's first dam project of this size and level of electricity-generation capacity. At full capacity, the BHP can generate 2400 mW,

while its artificially formed reservoir is the largest lake in Malaysia with a surface area of approximately 70,000 ha (roughly equal to the size of Singapore) and a storage volume of 43.8 billion m<sup>3</sup>.



Figure 1. Location of the Bakun Hydro-electric Project. Source: Drawn by Nor-Hisham.

The dam has had a major environmental and social impact. First, it necessitated clear-cutting of around 700 km<sup>2</sup> of virgin tropical rainforest with the consequent loss of rare and endangered plant and animal species [24–26]. Moreover, it has caused the forced displacement and marginalization of an estimated 10,000 people, mostly indigenous Orang Ulu. This has raised serious concerns over social sustainability [17,21]. The Orang Ulu originally inhabited the lands along the Balui River in the Belaga District, but have now been resettled at the Resettlement Scheme of Sungai Asap (hereafter: RSSA). The Orang Ulu are considered a socially vulnerable and economically disadvantaged group in Malaysia. They are mainly engaged in agriculture and the exploitation of forest-based resources including hunting and gathering, small-scale trade and retail, and traditional fishery.

From an economic perspective, the BHP has also proven problematic, more specifically with regard to the economics of energy supply [17,18]. Initially, the government of Sarawak state planned that 90% of the generated electricity would be sent via undersea cables to peninsular Malaysia, as well as neighbouring Brunei, Indonesia, and the Philippines. However, this initiative was aborted due to concerns over costs and feasibility. Today, Sarawak state is still looking for ways to sell its surplus electricity and is operating the dam below its capacity [27].

In March 2012, shortly after the federal government began a corruption investigation into the BHP, the transnational mining corporation Rio Tinto cancelled a plan for the construction of a US\$2 billion aluminium smelter that would have used electricity from the BHP. Since the dam went into operation in 2011, a quarter of a century after it was approved, the total costs of the project had ballooned to RM 7.3. billion (as compared to the originally estimated costs of RM 2 billion).

# 2.2. History of EIA in Malaysia

Prior to 1970, the Malaysian economy was predominantly agricultural-based. There were virtually no environmental concerns due to a strong focus on economic development. Moreover,

environmentalism was minimal, if not absent [28,29]. The agenda for development in Malaysia could be argued to have started through the introduction of the New Economic Policy (NEP) in 1970. During the implementation of the NEP, many large-scale development activities were carried out. Inevitably, while this effort brought about positive economic effects, it imposed pressure on environmental quality and human well-being. As asserted by Aiken [28] the rapid exploitation of natural resources and the expansion of resource-based industries resulted in environmental problems.

Parallel and contradictory to this development, Malaysia attempted to align itself closer to global environmental policies, and more particularly, the principles of the Stockholm Conference on the Human Environment held in 1972 [30]. This was also due to the successful lobbying of Malaysian Non-Governmental Organizations (NGOs), such as the Consumers' Association of Penang (CAP) and the Malaysian Nature Society (MNS). These cumulative factors led to the adoption of the 1974 Environmental Quality Act [31,32].

The need for an EIA in Malaysia was not felt until several natural disasters occurred, more specifically, the flood in Kuala Lumpur in 1970 and in Bota Perak in 1976, respectively. The Department of Environment (DOE) of Malaysia was asked by the World Bank to assess environmental impacts, as flooding had become a major concern [33]. The World Bank visited Malaysia in 1975 and consequently drafted a report recommending the design and formulation of EIA as policy for any major development project [33,34].

EIA has only legally been implemented in Malaysia since 1988. Prior to this, it was in operation since 1979, but without statutory provision. In the Fourth Malaysia Plan (1981–1985), the emphasis on EIA was reinforced through three-pronged strategies: pollution control, comprehensive land use planning and integrated project planning [35]. This further strengthened the role of EIA in Malaysia. During the administration of Prime Minister Mahathir, the industrialisation and urbanisation intensified. This era was marked by economic liberalisation and privatisation causing rapid expansion of the business sector.

Taking cognizance of the fact that prevention measures had failed to address the environmental problems, the Malaysian government introduced EIA through an amendment of the Environmental Quality Act in 1985 (the inclusion of "Section 34A"). The introduction of EIA has—at least in principle—brought about a new approach in environmental planning, where anyone intending to carry out a certain "prescribed activity" has to undertake a study of the environmental impacts before approval can be granted. A major litmus test of the robustness of the EIA was presented the following year when the Malaysian government formally approved the construction of the BHP.

# 2.3. Collusion of Interests: A Line of Events

Mohamad Mahathir, former Prime Minister of Malaysia, was a staunch proponent of the BHP. He opined that the BHP was crucial for a reliable electricity supply for the nation's industrialisation and urbanisation. Furthermore, he argued that the socio-economic position of indigenous peoples, such as the Orang Ulu, could be improved through the resettlement scheme. In his view, the project was a catalyst for development. Within Sarawak state, the BHP was fervently championed by its Chief Minister, Taib Mahmud, who maintained:

Bakun is a huge gift from the federal government—proof of the lie of the outdated idea that the Federal Government wants to rob us of our resources and colonise us. With Bakun, Sarawak will be the powerhouse of Malaysia [24].

Under the banners of "public interest" and development, both Mahathir and Taib criticized those who opposed the project as foreign agents, unpatriotic, or anti-modernist [36]. However, politicians' interest in the BHP was not entirely out of public or national concerns. There was a tight collusion between political leaders and the corporate sector. A poignant example is the manner in which the project was granted to the main contractor. In 1994 the concession was awarded—without open tender—to Ekran Berhad, a company engaged in timber and property development, yet,

inexperienced in dam construction. The company's owner, Ting Pek Khiing, was widely known to have close personal connections to Mahathir, the federal Minister of Finance, and the Chief Minister of Sarawak State [24,37,38].

Ekran Berhad enjoyed substantial benefits from the BHP construction, including an estimated 4 billion tonnes of timber valued at RM one billion. In addition, the company was awarded 11,578 acres of plantation in the resettlement area. Lastly, it won a contract with a value of RM 300 million to construct the 165 km Bakun-Bintulu road connecting the BHP and the resettlement area (Figure 2).



**Figure 2.** Bakun-Bintulu road (connecting the Bakun Hydro-electric Project (BHP) and Resettlement Scheme of Sungai Asap (RSSA)). The road is often busy with logging trucks. Source: Photograph by Nor-Hisham.

Yet, in 1997 a dispute erupted between Ekran Berhad and the Swiss-Swedish engineering contractor, Asia Brown Boveri Limited (ABB). The latter objected to a RM 9 billion contract being awarded without open tender to four companies under Ting Pek Khiing's control [37,38]. In the wake of the 1997 Asian Economic Crisis, the federal government deferred further construction and Ekran Berhad withdrew from the project two years later. For this, it was compensated at RM one billion [24,37,39] (In 2010, Ting Pek Khiing was declared bankrupt by the Kuala Lumpur High Court because he defaulted on a loan amounting to RM 60.79 million that he had borrowed from the Bank of Commence 6 years earlier [40]).

The BHP was also strongly supported by other corporations. For one, there is the sole and largest cement producer in Sarawak: the Cahaya Mata Sarawak Berhad (CMSB). Onn Mahmud, the brother of Sarawak's Chief Minister, Taib Mahmud, is generally assumed to direct the operations of CMSB [37,41]. Moreover, Taib Mahmud's two sons, Sulaiman and Mahmud, had considerable interests in Pacific Chemicals, the company that was granted logging operations in the BHP project area [24]. In 2002, the entire operation over the BHP was taken over by a joint venture between Sime JV and Sinohydro of China: the Malaysia–China Hydro Joint Venture (MCHJV). Sime JV is a subsidiary of Sime Darby (a Malaysian flag conglomerate, with business roots in plantations), a so-called 'Government Linked Company' of which chairman and board of directors are traditionally appointed from former political leaders of Malaysia's major political party, the United Malays National Organization (UMNO) or other parties from the National Front, the coalition of which UMNO is a part. The chairman of Sime Darby is without exception appointed from previous UMNO leaders who have retired or failed to win general elections. For example, the previous Sime Darby Chairman was Musa Hitam (2007–2012), a former Deputy Prime Minister and ex-UMNO Deputy President. He was followed by Abdul-Ghani Othman (2013–2008), the former Chief Minister of Johore.

Sinohydro, Sime JV's counterpart, is a Chinese state-owned hydroelectric and construction corporation founded in 1950. In the wake of China's Belt and Road Initiative, the corporation has entered the global market, and is now active across Asia, Africa, the Americas, and Europe. Its current portfolio includes the construction of the Rio Blanco Dam in Honduras, the Hambantota Port in Sri Lanka, the Hamad Port in Qatar, and the Kochav-Hayarden Hydro-Electric Station in Israel. Under the new joint venture, Sime JV obtained 70% and Sinohydro 30% of the shares. Having described the network of interests that is tied to the BHP, we will proceed to examine the procedures, rhetoric, and actual practice of the BHP's EIA below.

# 2.4. The EIA between Rhetoric and Reality

The BHP was legally subject to federal EIA requirements as it involved at least one or more of the following, so-called "prescribed activities": (i) the conversion of hill forest land to other land use covering an area of 50 ha or more; (ii) logging activities which cover an area of 500 ha or more; (iii) dam and hydroelectric power scheme projects with dams over 15 m high; and, (iv) ancillary structure installations covering a total area in excess of 40 ha and/or reservoirs with a surface area in excess of 400 ha. As such, Malaysia's Environmental Quality act (Order 1987) ordains that:

Any person intending to carry out any of these prescribed activities shall appoint a qualified person to conduct an environmental impact assessment and to submit a report thereof to the Director-General (DG) (Section 34A (2), [42]).

In April 1994, Ekran Berhad designated the Centre for Technology Transfer and Consultancy at UNIMAS, a local Sarawakian university, as the main consultant for the BHP's EIA. Its CEO, Ting Pek Khiing, claimed that the EIA would be completed by June 1994. This tight timeframe raised questions of whether it would be rushed to completion [43]. One month later, the socio-economic impact study was carried out separately by the Sarawak State Planning Unit, a unit under the Sarawak Chief Minister's Department, causing concerns over the study's independency.

While the EIA was still being conducted, Ekran Berhad had already carried out preliminary construction work, and by September 1994, an earth-breaking ceremony for the BHP was officiated by its greatest proponent, Prime Minister Mahathir [24]. Markedly, the ceremony took place before the project had received Cabinet approval, and even before the EIA itself had been approved. Other groundwork followed, including land clearance, construction of the access road, and river diversion. Only by March 1995, the first EIA report (for the reservoir) obtained approval, a mere six months after submission [24]. The decision on this part of the EIA had been reached without public review and without the full EIA report being completed, despite clear legal requirements [44,45]. The three remaining parts of the EIA were then still pending.

On 20 April 1995, three indigenous representatives from the longhouses (large rural communal dwellings) of Long Bulan, Uma Daro, and Bato Kalo (now all inundated by the reservoir) filed an originating summons at the High Court in Kuala Lumpur. They claimed that the 1974 federal Environmental Quality Act had been violated, and asked to be given the right to make representation as provided for under the law [46]. However, the case did not materialize, as the project was by then delayed due to the Asian Economic Crisis and the fact that Ekran Berhad had pulled out.

According to federal regulations, the EIA process requires public participation at two distinct time points:

- The stage of conducting the EIA and the preparation of its report, when the project proponent ought to consult affected people via: (a) citizens' committees; (b) public meetings and workshops; and, (c) soliciting public opinion [47];
- The review of the report and its approval. At this point, the EIA report compiled by the project proponent must be displayed for public inspection including by the affected people [47].

In reality, however, the affected population and civic groups who demanded access to information received no response from authorities. For example, in February 1994, a coalition of NGOs and the

Democratic Action Party (DAP) urged the government to be more transparent in releasing information on the BHP. Yet, when a public forum was planned in Kuching, which was to be attended by the Finance Minister of Sarawak State, the Chair of the Bakun Development Committee, representatives of the Orang Ulu Communities and civic organizations, the event was cancelled two days before [48].

As controversy over the BHP was mounting, Prime Minister Mahathir criticised interest groups and NGOs, and maintained:

Some parties, like NGOs, oppose the EIA report merely because they want to reject it without listening to clarifications [49].

In addition, he asserted that it was

Normal practice for the government to approve an EIA before displaying it to the public [49].

To this, two NGOs, Sahabat Alam Malaysian and the Consumers' Association of Penang, retorted that there was no public copy for review, even four months after the EIA had been approved [49]. After strong protests, the reports were made available another month later, and only at the DOE Headquarters in Kuala Lumpur over 1300 km away from where the affected population lived. Moreover, despite the fact that two months is the standard term for public review, the EIA reports were extremely shortly displayed: in Belaga and Kapit the report was up for inspection during a single day [24].

Under the background above, we conducted our fieldwork—including a survey, participatory observation, and semi-structured interviews—in the project area and the resettlement area of the BHP.

# 3. Materials and Methods: Survey and Interviews

Our analysis of the Bakun Dam's EIA was accomplished through "mixed methods" [50,51], i.e., the use of multiple sources that include quantitative information gathered through a survey and the analysis of government and corporate statistics, as well as qualitative data gathered through interviews, participatory observation, and literature research. The survey and fieldwork have been conducted at the Resettlement Scheme of Sungai Asap (or RSSA), a new settlement area built for the displaced Orang Ulu. Preliminary data collection was done shortly after the dam started operation, in May 2011, and coupled with 21 semi-structured interviews and participatory observation. A second round of fieldwork was carried out from September until November 2012. This time, a pilot survey of fifteen households was carried out, followed by a full survey among 220 respondents from 10 "longhouses," large communal dwellings built on stilts and generally hold up to 100 families in separate living quarters. More specifically, these were: Uma Kulit, Uma Belor, Uma Daro, Uma Nyaving, Uma Kelep, Uma Lahanan, Uma Bawang, Uma Batu Liko, Uma Balui Ukap, Uma Bakah, Uma Badeng, Uma Penan, Uma Lasong, and Uma Juman. These 10 longhouses comprised five different ethnic sub-groups, namely the Kenyah, Kayan, Lahanan, Ukit, and Penan. The total population in the RSSA was 11,616 in 2012 (accounting for 2219 households) [52].

Of the total sample (Table A1, Appendix A), the majority (74.5%) was working as farmer, while logging workers accounted for 8.2%, transport workers for 5.5%, professional and clerical workers, respectively, for 2.7% and 1.4%, and other employment for 7.7%. In terms of monthly income, the majority (44.1%) earned a monthly income below RM 450.00 During the fieldwork period, the currency exchange between Malaysian Ringgit to US Dollar was around RM 0.32 to 1 US Dollar. whereas the second monthly income bracket (25.5%) earned RM 451.00–RM 700.00. Respondents who earned RM 700.00–RM1000.00 were recorded at 18.6%. Other high-income bracket categories contained few respondents. For example, those who earned a monthly income bracket of RM 1601.00–RM 1900.00 constituted 3.7% of respondents. As a comparison to the poverty line at the national level, this monthly income finding gives us an alarming trend whereby, according to Malaysia's poverty standard for the state of Sarawak, households with a total monthly income less than RM 830.00 and RM 520.00 are considered poor and extreme (hardcore) poor respectively [53]. The monthly income also exposed that the objective of the resettlement programme for the BHP which focuses on job-oriented activities

to generate higher income for the resettlers on a sustainable basis through the restructuring of the existing socio-economic activities as aimed by the Economic Planning Unit [54] is a long way from being achieved.

To achieve a higher degree of representativeness, the study employed a multi-stage cluster sampling [55,56] Clustering was made according to the ethnic sub-groups whereby about 10% of the total number of family heads of the Orang Ulu at the RSSA. In this way, the survey aimed to ensure that all the heads of the Orang Ulu households received equal opportunity to be selected as respondents. In addition, theoretical saturation was used to determine the sample size, up to the point where additional data provided no new insights into the research questions [57,58]. All surveys were executed by the main researcher without assistance from students, interpreters, or else. To prevent bias in answering the questionnaires, a household-by-household approach (one-on-one household visits) was utilized while group meetings or group discussions were intentionally avoided.

To complement and strengthen the quantitative part, a qualitative approach through semi-structured interviews was carried out with local community leaders at the RSSA. There are four categories of local leaders among the Orang Ulu: the highest is "Temenggung" followed by "Pemancar," then "Penghulu" and lastly "Tuai Umah" (or headman). Sixteen interviews were conducted with the local village leaders of the Orang Ulu of the Village Development and Security Committee. Semi-structured interviews were also conducted with a variety of stakeholders involved in the BHP (see: List of Interviewees, Appendix A). Each of the interview sessions lasted between sixty minutes to ninety minutes.

Lastly, the research also included literature and archival research on the larger Malaysian context of dam-building, EIA, and customary land rights. For this part of the research, we relied on federal and local state reports, as well as studies by NGOs and (semi)corporate organizations, such as the dam contractors and funding agencies.

Having discussed the materials and methods of this study, the following section continues to examine how the affected indigenous people, local leaders and NGOs' viewed the EIA. The section is structured around the following main themes: (1) the level of respondents' participation in the EIA; (2) the extent to which procedures of the EIA were followed in a fair and open manner, and (3) the management of conflict and consent to the project.

# 4. Results: The Fieldwork

# 4.1. Lacking Participation in EIA

Of the respondents, the greater majority (87.9%) were dissatisfied with the participation in the EIA. In contrast, only 5.4% were satisfied. Probing into this, respondents provided the following reasons (Figure 3):

- 1. 41.7% felt that the authorities did not consider their requests (such as concerning compensation or relocation);
- 2. 21.1% felt unable to effectively review the EIA reports due to the limited period granted to them;
- 3. 20.7% stated that the use of the English language in the reports posed a major barrier;
- 4. 15.1% maintained that the locations where the EIA reports were displayed were too far away from where they lived.

When we examined whether and to what extent respondents felt if the EIA had empowered their public participation, three quarters disagreed, of which 54.5% disagreed strongly. The survey findings were corroborated through the interviews with key informants. As, for instance, one of the plaintiffs in the court case against the government said:

*The government does not want to consult us. In this case, the government did not follow the* [EIA] *regulations (Interview, Kajing Tubek, villager, RSSA 24-9-2012).* 



**Figure 3.** Reasons for respondents' dissatisfaction with the Environmental Impact Assessment (EIA). Source: This survey.

Similarly, a representative from the NGO Sahabat Alam Malaysia, asserted:

Public participation in Malaysia is conducted more as a procedural requirement. Project developers do not encourage it. They tend to be silent rather than to facilitate it (Interview, Jau-Evong Jok, coordinator, Marudi, Sarawak, 19-9-2012).

# 4.2. Procedures: Manipulation, Bias, and Distrust

Due to the limited possibilities to participate, the majority (80.0%) of the respondents felt that the authorities had conducted the EIA without complying to procedures. In fact, 36.7% was of the opinion that there were elements of manipulation, while 21.1% felt that the EIA was biased towards project proponents, and 19.5% stated that it focused on local leaders rather than the general populace (Figure 4).



Figure 4. What are the non-compliance aspects of the EIA? Source: This survey.

The respondents' negative views on the role of the EIA were reflected by the NGOs. The Executive Director of the Borneo Resources Institute, an indigenous civil organization in Sarawak stated:

In many cases, the EIA process is used to legitimize the project proponent's action without genuine intention to empower the public. Many do it (sic) just for the sake of procedure or formality, particularly when the project is strongly backed by the government. This is clearly the case in the BHP (Interview, Mark Bujang, Miri Sarawak, 21 September 2011).

In the next part of the survey, we probed respondents' view of the government's role in the EIA. It was found that the majority (80.5%) disagreed that the BHP had increased their trust towards the government. Three-quarters or 75.5% of the respondents disagreed (of which 40% strongly disagreed) that the way in which the EIA had been conducted by the government was acceptable. Lastly, 85.9% agreed (of which 36.8% strongly) that politicians were the ones who reaped the economic benefits from the dam project.

# 4.3. Conflict Management and Consent

One of the stated objectives of the Malaysian EIA is its role as a platform where social discontent can be managed, facilitated, and channelled through public participation. Consequently, participation is assumed to mitigate social conflict towards the proposed projects. Unfortunately, the survey revealed that 75% of the respondents felt that the public participation had failed to reduce conflict, while 13.6% even felt that it had increased conflict.

In order to have a better understanding of the nature and frequency of conflicts, we asked the respondents about how they protested and how often they were engaged in them. Over half (53.9%) expressed to have resisted against the project through non-violent approaches, such as writing of petitions and letters, amongst them, 37.0% had written between 2–4 times (since the moment they were informed about the project and the time of resettlement) (see Table 1). The second type of protest was through road blocks, to which 18.7% resorted, mostly only one time (by 10.2%). This was closely followed by demonstrations undertaken by 17.4%, with over 12% equally divided over having engaged in it either once or between 2–4 times. Open fights with workers was the form of protestation that was opted for only by 9.4% of the respondents, varying between once (4.1%) to 2–4 times (also 4.1%) before they were resettled.

Trues of Desistence		Freque	ncy (% )		% A co of Total
Type of Resistance	<b>1</b> ×	2–4×	5–7×	>8×	% Age of Total
Petitioning	12.8	37.0	3.5	0.6	53.9
Road blocks	10.2	5.3	2.0	1.2	18.7
Demonstration	6.4	5.8	3.7	1.5	17.4
Fights with project workers	4.1	4.1	1.2	0	9.4
Others	0.6	0	0	0	0.6
Total	34.1	52.2	10.4	3.3	100.0

Table 1. Type and frequency of protest prior to resettlement.

Source: This survey.

A final indication of the failure of the EIA to rally greater social support was with regard to the issue of consent (albeit not legally required for projects to go ahead). An overwhelming majority (84.5%) stated that they had not consented to the project, while a similar percentage (86.0%) indicated they had also not agreed with the decision to be relocated.

Other reports have corroborated that local people were forced into relocation through intimidation, tricks, and threats [59]. For instance, the government withheld monetary compensation to force people to accept resettlement at the RSSA [60–62]. In this regard, a female leader of one of the communal longhouses stated:

If we have an opportunity, of course we don't want to move out from Ulu Balui to here. However, the government withheld our compensation money. ( ... ) So, we have to agree to resettle. No option (Interview, Devong Anjie, headwoman of Uma Nyaving, RSSA, 1 October 2012).

# 5. Discussion: Limits and Opportunities of the EIA

This article has presented and analysed the EIA of the Bakun Hydro-electric Project. It aimed to demonstrate how the politics and rent-seeking that surround the EIA have been decisive in its failure to safeguard the overall sustainability of the project. By analysing the EIA's execution itself, coupled to fieldwork in the resettlement area of the indigenous population displaced by the dam, several conclusions can be drawn.

One, both at the federal and the regional state level, there were significant vested interests in an expedient, hasty approval of the EIA. Scholarly discussion has also focused on the question whether rent-seeking in resource-rich economies fosters development, although the evidence on this is inconclusive [63,64]. Through a collusion between large companies that stand to benefit from the dam and the leadership of political parties, such as UMNO, the EIA was diluted into a mere formality. The disregard of a legally required EIA was most glaring in the fact that commencement of the dam construction took place before having been accorded by the Cabinet, and even prior to the EIA's approval. Whereas the EIA was not submitted for review until September 1994, construction had already started 22 days later. The EIA was approved just 6 months later, in March 1995.

Two, the government's rush with the EIA has caused widespread resentment among the affected indigenous population. Most respondents (close to 90%) were not satisfied with their level of participation in the EIA. Reasons for this were enumerated as (in order of importance): (1) lacking government attention to information requests; (2) limited time allowed for reviewing the EIA reports; (3) linguistic barriers making it impossible to understand the reports, and; (4) the remote location where the reports were displayed, i.e., in the nation's capital over 1300 km away from the indigenous settlements. Thus, it is not surprising that four-fifths of the respondents maintained that the government paid scant regard to the EIA procedures. Of these, well over one-third felt that the EIA had been manipulated, whereas one fifth stated that it was biased towards supporters of the dam, and another one fifth maintained it was focused on local leaders and not the general populace.

Three, the EIA's objective to rally social support for the project through public participation has utterly failed, with almost 9 out of 10 respondents stating that they had not consented to the dam project, nor with the decision to be resettled. Alternatively, one might also ascertain that the objective to push through the BHP with a minimum effort at mitigating environmental impact and social grievances has succeeded. Demonstrating this bitter success is the fact that despite evident, widespread resentment over the dam amongst the affected population, it did go into operation nevertheless.

It is in the light of these three reasons that one may read others' condemnatory typification of the BHP:

# *If for no other reason, then, Bakun is an excellent case study for policymakers because it intimately sketches the anatomy of failure, a failure of government planning, implementation, and oversight, no matter how technically sound the dam's concrete face, spillway, or powerhouse become* [18].

When zooming out from our case-study, it can be seen that the BHP is no exception. Regardless of whether we look at large dam projects in the Philippines [5,65], Cambodia [66], or Laos [67], the EIA has been seriously impaired by the larger political context of which it is a part.

In this regard, other studies have called attention to the dynamics of rent-seeking surrounding mega-projects. It has, for instance, been observed that in Thailand, Indonesia, and Malaysia, the support for the EIA by political and corporate decision-makers was low [68]. Others duly noted:

Policy failures on environmental grounds need to be grasped for what it is—not as an overnight, nor as a faulty judgement. The decision of public policies in these countries is too often shaped both

*directly and indirectly, by those with a vested interest in the continued mismanagement of natural resources* [69].

Is this study testimony to the inefficacy and futility of the EIA? Not quite. There might be a dual lesson to be learnt from mega-dam projects. First, there is a tiny, yet limited window of opportunity through which the EIA practitioner (or opponents of the project) could hope to influence decision-making. Projects with the magnitude of the BHP tend to acquire an own momentum, due to the substantive economic interests and political prestige vested in them, leaving two ways out: to divert it, or if all fails, to divide it. Diverting a mega-project implies the search for alternatives that may achieve similar objectives, but in ecologically and socially less disruptive ways. For instance, in the case of hydro-electricity, one may consider managing demand through energy saving technologies or behavioural change towards "greener" life-styles [20]. However, when such proposals run counter to political realities, one could also try lobbying for the division of a mega-project into smaller, more manageable components (which could be politically more palatable than scrapping the project altogether). In both cases, diversion or division, the EIA might have an important role to play.

There is a second lesson from this case-study: if none of the options above are accepted, and the project is still given the green light, the EIA practitioner is to tread with extreme care not to legitimize something that should have been halted. With the legacy of the BHP by and large unresolved, there are well-founded concerns that such a scenario might unfold once more, not in the least given Sarawak State's plans to build another ten to twelve dams.

**Author Contributions:** P.H. (conceptualization; writing; funding acquisition; oversight); B.M.S.N.-H. (writing; methodology; formal analysis; data collection; visualization); H.Z. (methodology; validation; formal Analysis; corrections). All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the China National Science Foundation, grant number 71573231.

Conflicts of Interest: The authors declare no known conflicts of interest associated with this publication.

# Appendix A

Gender	Number	% of Total
Male	215	97.7
Female	5	2.3
Total	220	100
Age Cohort	Number	% of total
18–30 years	7	3.18
31–40 years	11	5.00
41–50 years	53	24.09
51-60	87	39.55
61 and above	62	28.18
Total	220	100
Education	Number	% of total
University	3	1.4
College	8	3.6
High school	58	26.4
Junior high school	54	24.5
No formal education	97	44.1
Total	220	100

Table A1. The respondents' profile.

Current Occupation	Number	% of total
Professional and administrator	6	2.7
Clerical	3	1.4
Farmer	164	74.5
Logging worker	18	8.2
Transport worker	12	5.5
Others	17	7.7
Total	220	100
Household Size	Number	% of total
<3	30	13.64
4–5	113	51.36
> 6	77	35.00
Total	220	100
Monthly Income	Number	% of total
RM 451.00-RM 700.00	153	69.6
RM 701.00-RM 1450.00	50	22.7
RM 1451.00-RM 2450.00	14	6.4
RM 2451.00 and above	3	1.4
Total	220	100

Table A1. Cont.

# List of interviewees

Anjie, D, a Headwoman of Uma Nyaving, at the RSSA on 1 October 2012

Bit, S., a Penghulu, at the RSSA on 30 September 2012

Bujang, M. an Executive Director of BRIMAS, in Miri Sarawak on 21 September 2011

Igang, D., a villager at the RSSA, by telephone on 22 July 2013

Igang, N., a Teacher at the RSSA on 11 October 2012

Imu, S., a villager and NGO Activist, at the RSSA on 12 October 2012

Jok, J-E, a Coordinator from SAM Sarawak, in Marudi, Sarawak on 19 September 2012

Kulleh, T., a Pemancar, at RSSA on 2 October 2012

Ligue, M., an Assistant of Tuai Umah from Uma Balui Liko, on 16 October 2012

Lihan, M.M., Chairman of the Village Development and Security Committee of the RSSA, at RSSA Bakun Sarawak on 25 May 2011

Lusat, K., Assistance District Office (ADO) of Sub-district Office of Sungai Asap, at the RSSA on 25 September 2015

Magui, M., a Tuai Umah of Uma Penan, at the RSSA on 11 October 2012

Nyipa, L., a Tuai Umah of Uma Lahanan, at the RSSA on 7 October 2012

Sanggul, A., a resettler, at the RSLB on 25 September 2012

Tubek, K., a villager cum a plaintiff, at the RSSA on 24 September 2012

Tungau, L., a villager, at the RSSA on 23 May 2011

Umek, J., a Pemancar, at the RSSA on 29 September 2012

Urun, A, a villager at RSSA on 1 October 2012)

# List of interviewed organizations

Association of the Orang Asal Network Peninsular Malaysia (JOAS)

Borneo Research Institute (BRIMAS)

Consumers' Association of Penang (CAP), Malaysia

Department of Environment (DOE) of Kuala Lumpur

Department of Environment (DOE) of Putrajaya (HQ)

Economic Planning Unit (EPU), Prime Minister's Department Ministry of Energy, Green Technology and Water (MEGTW) Natural Resources and Environmental Board (NREB) Sahabat Alam Malaysia (SAM), Penang Sahabat Alam Malaysia (SAM), Sarawak Sarawak Hydro Berhad Sarawak Economic Planning Unit, Kuching Sarawak Sub-district Office of Sungai Asap, Sarawak Traditional administrators at the Resettlement Scheme of Sungai Asap, Sarawak Village Development and Security Committee of the RSSA, Sarawak

# References

- 1. Clark, B.D.; Turnbull, R.G.H. Proposals for environmental impact assessment procedures in the UK. In *Planning and Ecology*; Roberts, R.D., Ed.; Chapman and Hall: London, UK, 1984.
- 2. Wathern, P. An introductory guide to EIA. In *Environmental Impact Assessment: Theory and Practice;* Wathern, P., Ed.; Unwin Hyman Ltd.: London, UK, 1988.
- 3. Canter, L.W. Environmental Impact Assessment, 2nd ed.; Irwin McGraw-Hill: New York, NY, USA, 1996.
- 4. Glasson, J. The first 10 years of the UK EIA system: Strengths, weakness, opportunities and threats. *Plan. Pract. Res.* **1999**, *14*, 363–375. [CrossRef]
- 5. Bravante, M.A.; Holden, W.N. Going through the motions: The environmental impact assessment of nonferrous metals mining projects in the Philippines. *Pac. Rev.* **2009**, *22*, 523–547. [CrossRef]
- 6. Formby, J. The politics of environmental impact assessment. *Impact Assess.* **1990**, *8*, 191–196. [CrossRef]
- 7. Ortolano, L. Environmental Regulation and Impact Assessment; John Wiley & Sons: Hoboken, NJ, USA, 1997.
- 8. Jay, S.; Jones, C.; Slinn, P.; Wood, C. Environmental impact assessment: Retrospect and prospect. *Environ. Impact Assess. Rev.* 2007, 27, 287–300. [CrossRef]
- 9. Amy, J.D. Decision techniques for environmental policy: A critique. In *Managing Leviathan: Environmental Politics and the Administrative Statel*; Paehlke, R., Torgerson, D., Eds.; Broadview Press Ltd.: Peterborough, ON, Canada, 1990.
- Bartlett, R.V. Ecological reason in administration: Environmental impact assessment and administrative theory. In *Managing Leviathan: Environmental Politics and the Administrative State*; Paehlke, R., Torgerson, D., Eds.; Broadview Press Ltd.: Peterborough, ON, Canada, 1990.
- 11. Bryant, R.L.; Parnwell, M.J.G. Introduction: Politics, sustainable development and environmental change in South-East Asia. In *Environmental Change in South-East Asia: People, Politics and Sustainable Development*; Bryant, R.L., Parnwell, M.J.G., Eds.; Routledge: Abingdon-on-Thames, UK, 1996.
- 12. Buhrs, T. *Environmental Integration: Our Common Challenge;* State University of New York Press: Albany, NY, USA, 2009.
- 13. Tenney, A.; Kvaerner, J.; Gjerstad, K.I. Uncertainty in environmental impact assessment predictions: The need for better communication and more transparency. *Impact Assess. Proj. Apprais.* **2006**, *24*, 45–56. [CrossRef]
- 14. Dipper, B.; Jones, C.; Wood, C. Monitoring and post-auditing in environmental impact assessment: A review. *J. Environ. Plan. Manag.* **1998**, *41*, 731–747. [CrossRef]
- 15. Weston, J. EIA in a risk society. Environ. Plan. Manag. 2004, 47, 313–325. [CrossRef]
- 16. Lawrence, D.P. Impact significance determination-Pushing the boundaries. *Environ. Impact Assess. Rev.* **2007**, *27*, 770–788. [CrossRef]
- 17. Ijabadeniyi, A.; Vanclay, F. Socially-tolerated practices in environmental and social impact assessment reporting: Discourses, displacement, and impoverishment. *Land* **2020**, *9*, 33. [CrossRef]
- 18. Choy, Y.K. Sustainable Development—An institutional enclave (with special reference to the Bakun Dam-Induced development strategy in Malaysia). *J. Econ. Issues* **2005**, *39*, 951–971.
- 19. Sovacool, B.K.; Bulan, L.C. Behind an ambitious megaproject in Asia: The history and implications of the Bakun hydroelectric dam in Borneo. *Energy Policy* **2011**, *39*, 4842–4859. [CrossRef]
- 20. Lee, W.C.; Viswanathan, K.K.; Ali, J. Compensation policy in a large development project: The case of the Bakun hydroelectric dam. *Int. J. Water Resour. Dev.* **2015**, *31*, 64–72. [CrossRef]

- 21. Choy, Y.K. Sustainable development and the social and cultural impact of a dam-induced development strategy—The Bakun experience. *Pac. Aff.* **2004**, *77*, 50–68.
- 22. Andre, E. Beyond hydrology in the sustainability assessment of dams: A Planners perspective—The Sarawak experience. *J. Hydrol.* **2012**, *412*, 246–255. [CrossRef]
- 23. Memon, P.A. Devolution of environmental regulation: Environmental impact assessment in Malaysia. *Impact Assess. Proj. Apprais.* **2000**, *18*, 283–293. [CrossRef]
- 24. INSAN. Power Play: Why We Condemn the Bakun Hydroelectric Project; INSAN: Kuala Lumpur, Malaysia, 1996.
- 25. Allison, T. Malaysia's Bakun Project: Build and be Damned. 2000. Available online: http://www.atimes.com/ reports/BJ28Ai01.html (accessed on 20 August 2010).
- Beck, M.W.; Claassen, A.H.; Hundt, P.J. Environmental and livelihood impacts of dams: Common lessons across development gradients that challenge sustainability. *Int. J. River Basin Manag.* 2012, 10, 73–92. [CrossRef]
- 27. Ng, E. Malaysia Shelves Plans for Undersea Power Cable. 2010. Available online: http://www.businessweek. com/ap/financialnews/D9FKJ6GG0.htm (accessed on 20 August 2013).
- 28. Aiken, S.R. Environment and the federal government in Malaysia. Appl. Geogr. 1998, 8, 291–314. [CrossRef]
- 29. Sani, S. Economic development and environmental management in Malaysia. *New Zealand Geogr.* **1993**, *49*, 64–68. [CrossRef]
- 30. Thanarajasingam, S. *Background Paper on Law, Policy and the Implementation of a Conservation Strategy;* Economic Planning Unit (EPU) Prime Minister's Department of Malaysia: Kuala Lumpur, Malaysia, 1992.
- 31. Ramakrishna, S. The environmental movement in Malaysia. In *Social Movement in Malaysia: From Moral Communities to NGOs;* Weiss, M.L., Hassan, S., Eds.; Routledge Curzon: Abingdon-on-Thames, UK, 2003.
- 32. Aiken, S.R.; Leigh, C.H. Land use conflicts and rainforest conservation in Malaysia and Australia. *Land Use Policy* **1986**, *3*, 161–179. [CrossRef]
- 33. Staerdahl, J.; Zakaria, Z.; Dewar, N.; Panich, N. Environmental impact assessment in Malaysia, South Africa, Thailand, and Denmark: Background, layout, context, public participation and environmental scope. *J. Transdiscipl. Environ. Stud.* **2004**, *3*, 1–19.
- 34. Ho, P.Y.C. Background Paper on Development and Implementation of the Environmental Impact Assessment (EIA) System in Malaysia; Economic Planning Unit (EPU) Prime Minister's Department of Malaysia: Kuala Lumpur, Malaysia, 1992.
- 35. GOM (Government of Malaysia). *Fourth Malaysia Plan.* (1981–1985); Government Printers: Kuala Lumpur, Malaysia, 1979.
- 36. Pura, R. Court Decision Poses Hurdle for Malaysia's Bakun Dam. 1996. Available online: http://www.wsj. com/articles/SB835214167654346500 (accessed on 10 August 2010).
- 37. Gomez, E.T. Chinese Business in Malaysia: Accumulation, Accommodation and Ascendance; Curzon Press: Richmond, UK, 1999.
- 38. Wain, B. Malaysian Maverick: Mahathir Mohamad in Turbulent Times; Palgrave Macmillan: London, UK, 2009.
- 39. Gomez, E.T.; Jomo, K.S. Malaysia's Political Economy: Politics, Patronage and Profits; University Press: Cambridge, UK, 1999.
- 40. The Borneo Post. Govt to Ensure Runway and Flyover Projects Completed—CM. 2010. Available online: http://www.theborneopost.com/2010/11/23/govt-to-ensure-runway-and-flyover-projects-completed-%E2%80%93-cm/ (accessed on 20 August 2011).
- 41. Utusan Konsumer. *Big Money for Big Boys: All in the Family—Cahaya Mata Sarawak Berhad (CMSB);* Utusan Konsumer: Pulau Pinang, Malaysia, 2001; p. 17.
- 42. GOM (Government of Malaysia). Environmental Quality Act. 1974: Regulations, Rules & Orders (as at 20st November 2013); International Law Book Services: Kuala Lumpur, Malaysia, 2013.
- 43. Utusan Konsumer. *Law Gives Bakun EIA Assurance: UNIMAS Pressured to Complete EIA Study Early?* Utusan Konsumer: Pulau Pinang, Malaysia, 1994; p. 4.
- 44. Utusan Konsumer. Sheer Mockery in EIA Approval! Utusan Konsumer: Pulau Pinang, Malaysia, 1995; p. 1.
- 45. Utusan Konsumer. Cabinet Accepts EIA in Stages; Utusan Konsumer: Pulau Pinang, Malaysia, 1995; p. 6.
- 46. Nijar, G.S. The Bakun dam case: A critique. Malay. Law J. 1997, 3, 229–230.
- 47. DOE (Department of Environment). *A Handbook of Environmental Impact Assessment Guidelines*, 5th ed.; Department of Environment, Ministry of Natural Resource & Environment: Putrajaya, Malaysia, 2009.
- 48. ALIRAN. Cancellation of forum on Bakun: Violation of human right. ALIRAN, 14 April 1994; p. 7.

- 49. Utusan Konsumer. *PM Grossly Misled on Bakun: Govt Should Welcome Fair Comment;* Utusan Konsumer: Pulau Pinang, Malaysia, 1995; p. 6.
- 50. Creswell, J.W.; Creswell, J.D. *Research Design: Qualitative, Quantitative and Mixed Methods Approaches*, 5th ed.; SAGE Publications, Inc.: Thousand Oaks, CA, USA, 2018.
- 51. Brannen, J. Mixing Methods: The entry of qualitative and quantitative approaches into the research process. *Int. J. Soc. Res. Methodol.* **2005**, *8*, 173–184. [CrossRef]
- 52. Sub-District Office of Sungai Asap. Basic Profile of Sungai Asap Population, Unpublished, 2012. Available online: http://www.ictu.tmp.sarawak.gov.my/seg.php?recordID=M0055&contentID=SM0124 (accessed on 8 December 2020).
- 53. ICU (Implementation and Coordination Unit). Press Release: Information System on National Poverty (e-kasih). 2011. Available online: http://www.icu.gov.my/pdf/kenyataan/kenyataan\_media\_ekasih.pdf (accessed on 24 January 2011).
- 54. EPU (Economic Planning Unit). *Bakun Hydroelectric Project: Green Energy for the Future;* Economic Planning Unit (EPU), Prime Minister's Department of Malaysia: Kuala Lumpur, Malaysia, 1996.
- 55. Babbie, E. *The Practice of Social Research*, 14th ed.; Cengage Learning: Boston, MA, USA, 2016.
- 56. Neuman, W.L. *Social Research Methods: Qualitative and Quantitative Approach*, 7th ed.; Pearson Education Inc.: Hudson, NY, USA, 2011.
- 57. Small, M.L. How many cases do I need? On science and the logic of case selection in field-based research. *Ethnography* **2009**, *10*, 5–38. [CrossRef]
- 58. Morse, J.M. Editorial: Qualitative significance. Qual. Health Res. 2004, 14, 151–152. [CrossRef] [PubMed]
- 59. Hutnyk, J. Resettling Bakun: Consultancy, anthropologists and development. *Left Curve* **1999**, *23*, 82–90.
- 60. Utusan Konsumer. *Sweet on the Outside, Sour in the Inside;* Utusan Konsumer: Pulau Pinang, Malaysia, 2000; p. 1.
- 61. Gabungan (The Coalition of Concerned NGOs on Bakun of Malaysia). The Resettlement of Indigenous People Affected by the Bakun Hydro-Electric Project, Sarawak, Malaysia. 1999. Available online: http://www.internationalrivers.org/files/attached-files/resettlement\_of\_indigenous\_people\_at\_bakun.pdf (accessed on 10 August 2010).
- 62. Gabungan (The Coalition of Concerned NGOs on Bakun of Malaysia). *The Mother of Bakun: Fact. Finding Mission on Bakun Dam;* Gabungan: Petaling Jaya, Malaysia, 1999.
- 63. Sadik-Zada, E.R.; Loewenstein, W. A note on revenue distribution patterns and rent-seeking incentive. *Int. J. Energy Econ. Policy* **2018**, *8*, 196–204.
- 64. Sadik-Zada, E.R. Distributional bargaining and the speed of structural change in the petroleum exporting labor surplus economies. *Eur. J. Dev. Res.* **2020**, *32*, 51–98. [CrossRef]
- 65. Kim, S. Greening the dam: The case of the San Roque multi-purpose project in the Philippines. *Geoforum* **2010**, *41*, 627–637. [CrossRef]
- 66. Hensengerth, O. Hydropower planning in institutional settings: Chinese institutions and the failures of environmental and social regulation in Cambodia. In *Evolution of Dam Policies: Evidence from the Big Hydropower States;* Scheumann, W., Hensengerth, O., Eds.; Springer: Heidelberg, Germany, 2014.
- 67. Goldman, M. Imperial Nature: The World Bank and Struggles for Social Justice in the Age of Globalization; Yale University Press: New Haven, CT, USA, 2005.
- 68. Boyle, J. Cultural influences on implementing environmental impact assessment: Insights from Thailand, Indonesia, and Malaysia. *Environ. Impact Assess. Rev.* **1998**, *18*, 95–116. [CrossRef]
- 69. Broad, R. The political economy of natural resources: Case studies of the Indonesian and Philippines forest sector. *Dev. Areas* **1995**, *29*, 317–340.

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





# Article Understanding Social Impact and Value Creation in Hybrid Organizations: The Case of Italian Civil Service

Paolo Esposito<sup>1</sup>, Valerio Brescia<sup>2,\*</sup>, Chiara Fantauzzi<sup>3</sup> and Rocco Frondizi<sup>3</sup>

- <sup>1</sup> Department of Law Economics Management and Quantitative Methods, University of Sannio di Benevento, 82100 Benevento, Italy; pesposito@unisannio.it
- <sup>2</sup> Department of Management, University of Turin, 10134 Turin, Italy
- <sup>3</sup> Department of Management and Law, University of Rome Tor Vergata, 00133 Rome, Italy;

chiara.fantauzzi@uniroma2.it (C.F.); rocco.frondizi@uniroma2.it (R.F.)

Correspondence: valerio.brescia@unito.it

**Abstract**: The aim of this paper is twofold: first, it aims to analyze what kind of value is generated by hybrid organizations and how; second, it aims to understand the role of social impact assessment (SIA) in the measurement of added value, especially in terms of social and economic change generated by hybrids. Hybrid organizations are a debated topic in literature and have different strengths in responding to needs, mainly in the public interest. Nevertheless, there are not many studies that identify the impact and change generated by these organizations. After highlighting the gap in the literature, the study proposes an innovative approach that combines SIA, interview, interventionist approach and documental analysis. The breakdown of SIA through the five elements of the value chain (inputs, activities, outputs, outcomes, and impact) guarantees a linear definition of the value generated through change with procedural objectivity capable of grasping hybrid organizations' complexity. The value generated or absorbed is the change generated by the impact measured based on the incidence of public resources allocated. Through the SIA and counterfactual approach, the civil service case study analysis highlights how the value generated by public resources can be measured or more clearly displayed in the measurement process itself.

Keywords: hybrid organization; social impact; value creation; civil service; social impact assessment

# 1. Introduction

Academics have been debating the definition and outcome of hybrid organizations [1]. In the literature, there are three classic forms of hybrid organizations: hybrid public companies formed by subsidiaries and investee companies, social cooperatives to which the public provides a mandate for achieving specific objectives and social enterprises [2,3]. However, the classical forms of hybrid organization are added to other more complex forms and have not yet been studied in the literature. An example is given by those third sector organizations that receive public funds, where in fact, the public sector is a partner in specific activities and in some cases also involves the private sector to achieve shared social objectives. In these cases, the interest in the project's impact and the value created become recurring themes of interest for the new organizational forms created. The subjects involved in the hybrid organization pursue the common interest and support the achievement of social, environmental, or economic needs by overcoming potential obstacles thanks to the characteristics of each one who in such organizations unite to form a new organizational form [4]. The reform linked to New Public Management has led to different types of outsourcing of public interest services, some typically oriented towards new corporate forms under public control, others towards the public-private partnership, others towards new hybrid solutions in which different organizational forms are united to answer the need [5–8]. Furthermore, since 2017 in Italy, the country has

Citation: Esposito, P.; Brescia, V.; Fantauzzi, C.; Frondizi, R. Understanding Social Impact and Value Creation in Hybrid Organizations: The Case of Italian Civil Service. *Sustainability* **2021**, *13*, 4058. https://doi.org/10.3390/ su13074058

Academic Editor: Christian Vandenberghe

Received: 3 March 2021 Accepted: 3 April 2021 Published: 6 April 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). witnessed the reorganization of legislation on the third sector, including associations (social promotion association, voluntary associations), social cooperatives, social enterprises, ecclesiastical bodies, and foundations [9]. The new legislation increasingly directs existing organizations towards general interests of public interest, with objectives of social and environmental sustainability as already highlighted for non-profits and the possibility of carrying out commercial activities with a type of taxation different from those profit companies community-oriented shared public purposes [10,11]. In several cases, the reform has changed the third and non-profit, making it part of the public fabric through new forms of hybrid organization to respond to a lack of available resources and economic difficulties [10,12,13]. The recent global financial crisis and the pandemic triggered by COVID-19 have had various effects on evaluating public performances that have led to budget cuts and the search for new organizational forms to meet the need with fewer resources available [14,15]. To decrease the available public budget associated with better systems of efficiency and use of public resources, the parallel stream of research on sustainable business models capable of responding to the economic, social, and environmental system's developmental complexity has been added [16–18]. The approach provided by hybrid organizations could support the correct use of resources, the limited use of the same response to a paradigm already studied by O'Flynn [19], who questioned himself with the adoption of new public management in the twentieth century of managerial approaches borrowed from the private sector and applied to the public sector to increase public value. The public value generated by the adoption of new managerial approaches, as clarified by Stoker [20], includes the involvement of stakeholders in the collaborative and production process as a way of overcoming the previous model. From this perspective, the analysis conducted sees the subjects traditionally involved as stakeholders on equal components of hybrid organizations.

O'Flynn [19] identifies in the concept of public value a new approach that adheres to that defined by hybrid organizations which, as theorized by the scholar, have a greater capacity for collecting preferences, a multi-accountability approach that involves all subjects, the ability to pursue multiple objectives including service results, satisfaction, results, trust, and legitimacy. The public value generated by hybrid organizations can be determined through the social and economic impact of the projects and services of public interest carried out [16,21]. Public value offers a broader way of measuring government performance and guiding policy decisions, according to Kelly, Mulgan, and Muers [22], where public value could measure the impact of public interest projects. The concept of performance recalled in the paper therefore refers to the holistic conception of public value. The literature on hybrid organizations has not yet defined the public value it can generate either through a meaningful approach or case studies. The analysis conducted focuses on an example of a hybrid organization generated by a large project shared between the different types of organizations to merge them into a single organization with the same expectations and interests. This is the Italian national civil service. The theoretical approach proposed to define public value in hybrid organizations can be generalized as it responds to the verifications and requirements identified by Ruddin [23], who also highlights how an approach that can be generalized to other case studies has practical relevance in social studies. In the analysis conducted, the same hybrid organization approach applied to civil service projects can be adopted in different European contexts. The analysis conducted after a brief description of the existing literature seeks to identify a possible approach to determine the hybrid organization's social impact and value [24]. The approach to facilitate understanding of the logical passage divides the project into the five phases of the social impact assessment approach [25,26]. Interviews, collection of results, interventionist approach, and application proposal for evaluating the added value allow the inductive passage [27] from practice to theory and vice versa, highlighting and understanding the main aspects. The study initiates a process of reflection on complex hybrid organizations and the possibility of measuring the impact generated through an innovative approach from a logical point of view. Following the research gap and the observation of empirical

phenomena not explained by existing studies, our analysis will provide answers to the following research question (RQ):

RQ1: How do hybrid organizations generate value and what kind of value is it?

- RQ2: How does the social impact assessment measure the phenomenon of value creation linked to change?
- RQ3: Is social impact assessment a useful tool to understand the use of public resources generated by hybrid organizations?

The study consists first of an analysis of the literature and the second section presents the methodology and the inductive approach on the specific case divided into phases and recalls through the social impact analysis. The findings guide the reader in the subsequent discussion and conclusions, providing practical elements to demonstrate the approach's capacity and generated value. The discussion guides the reader to understanding social impact. The conclusion highlights the international debate theme and the possible application of the results in an international context.

# 2. Theoretical Framework

This section is dedicated to the theoretical framework of the paper and it is composed of two main pillars. First of all, the purpose is to analyze several definitions of hybrid organizations. Then, the aim is to analyze the academic debate on impact measurement in hybrid organizations, individuating a literature gap on value creation evaluating to fill. The debate on the issue of social impact is generated by the consideration of multiple criteria of different nature (economic, environmental, and social), as well as the transparency and engagement of the different stakeholders, such as organizations, government, and communities oriented towards mapping resource sustainability in a complex environment.

# 2.1. Framework

The term "hybrid organization" is used to combine elements from both for-profit and non-profit sectors, to maintain a mixture of market and mission oriented practices, and to address economic, social, and ecological issues [28,29]. Nevertheless, there are different definitions of hybrid organizations, due to their nature and complexity [30].

By definition, hybrids are the offspring of different species [31] and, for what concerns the organization and management literature, the term has been employed to describe organizations that draw on at least two different paradigms, logics, and value systems, allowing the rise of a new conception of economic organizing [32]. In this sense, hybrid organizations can be seen as a new form of organization, able to compete not only on the quality of goods, but also on the capacity to effect positive social and environmental change [33]. Boyd et al. [34] defined hybrid organizations as entities that are market-oriented and mission-centered, which can be studied on the basis of two specific criteria [35]:

- they have a business model aimed to create social value;
- they are able to generate income to sustain their operations.

Hybridity represents a mixture of several heterogeneous components and does not refer to something new, but to a new combination of existing elements. Besharov and Smith [36] used this term to indicate the coexistence of two or more distinct forms of organization. Jay [37] provided a similar definition, according to which hybrid organizations are seen as entities able to combine multiple institutional logics to solve complex problems. Indeed, a hybrid organization is driven by two forces, represented by social change and the sustainability of organization [38], offering a blended value proposition, composed of economic, social, and environmental components. It is in this sense that, according to Santos [39], this type of organization is mainly focused on creating value rather than capturing it, while Haigh and Hoffman [40] underline its ability to provide high quality differentiated goods and to pursue both a social and environmental mission. Furthermore, Boyd et al. [34] state that hybrid organizations are generally characterized

by a long-term perspective on profit and a very close and personal relationship with their crucial stakeholders (suppliers, producers, and customers).

According to Grossi et al. [41], in the field of public administration and management, hybridity refers to organizations composed of structural elements deriving from other types of organizations belonging to different sectors (private, for profit, and third sector). It is within such triangulation that they identify specific hybrids:

- public/private for profit hybrids, like state-owned enterprises;
- private for profit/third sector hybrids, like social enterprises or co-operatives;
- public/third sector hybrids, like welfare associations or other organizations sponsored by government.

The concept of hybrid organizations has been adopted to describe various configurations of cross-sectoral collaboration, such as network and hierarchy [42], government and business [43], academic and market [44], healthcare and science [45]. According to Doherty et al. [46], by spanning the boundaries of private, public, and non-profit sectors, hybrid organizations are able to bridge institutional fields, facing conflicting institutional logics. In this sense, they are able to earn trust and the authority to establish connection and dialogue between several categories of actors, also by including former opponents [47].

Given these considerations, for the purpose of this paper, we adopt the definition provided by Jolik and Niesten [48], according to which hybrids represent collaborations between independent organizations that exchange and develop goods and services to create value, reduce agency and transaction costs and allocate residual claims, by combining resources, organizing information, and safeguarding contractual hazards and property rights. Despite the recent increased interest in hybrid organizations, the literature appears fragmented across many academic disciplines over several decades.

According to one of the most popular approach, hybrid organizations are seen as a continuum between sectors [49,50], while other authors preferred to follow a "single sector emphasis" [51–54], studying hybrid organizations from the perspectives of one specific sector, the public or the private one.

Other writers have gone further in a separate sector approach, looking at hybridization and hybrid organizations as the permanent features in the welfare system [55,56].

Furthermore, numerous studies were carried out with the aim to individuate the factors on which depends the likelihood to develop a hybrid organization [57]. They focused on the motivation of traditional entrepreneurs. Some showed the desirability for self-employment, tolerance for risk and self-efficacy at the center of their interests [48]. The literature on hybrid organizations presents hybridity as an enabling condition to achieve legitimacy from different institutional logics, in order to survive [28,58]. On the other hand, organizational stakeholders and policy-makers can become "institutionally confused" if an hybrid's behavior does not match the description of the ideal-typical form of organization in contemporary society [59]. In this sense, even if hybrid organizations have to solve tensions regarding their identity, according to Battilana and Lee [60] and Santos et al. [39], the conflicting logics they respond to and the practices they implement can be seen as the essence of their constitution, the condition to handle specific and complex situations [61].

The specific concept of hybridity in public administrations still appears undeveloped [62]. Several studies have analyzed hybridization within public organizations [63], considering structural and cultural complexity as its main feature [64,65]. Structural complexity in public organizations can be measured in terms of vertical specialization and horizontal specialization, both characterized by intra and inter-organizational elements [58–60]. Vertical intra-organizational specialization indicates how formal authority is distributed among different levels of hierarchy. Vertical inter-organizational specialization instead focuses on specialization among public organizations (ministries with many subordinate agencies). On the other hand, horizontal intra-organizational specialization indicates the internal specialization of public organizations, while horizontal inter-organizational focuses on the level of specialization among public organizations at the same levels. Cultural complexity, instead, shows the variety of informal, cultural norms and value within and among public organizations. A weak level of cultural complexity means cultural homogeneity and integration, with members who are all committed to the same norms and values. On the basis of these considerations, Christensen and Laegreid [63] affirmed that hybridization in public organizations can assume different meanings, which can be explained through complexity (in all its configurations). More deeply it reflects potential inconsistency between diverse structural and cultural elements in a public entity. In this conception, hybrid structures that follow different organizational principles can link new means of coordination and traditional sectoral agreements and be an effective way to manage the "coordination paradox" (i.e., vertical coordination measures can counteract horizontal coordination). However, the performance and effects of these practices are often conflicting and uncertain, and there is a trade-off between potential gains through flexibility and disadvantages through ambiguities, tensions and conflicts that fall within the sphere of cultural elements. Coordination is a crucial structural element for governance and quality capability because it shapes program design and influences efficiency gains, which in turn affect government legitimacy.

# 2.2. Hybrid Organizations and the Determination of the Generated Value

Academics and scholars are now debating the fallout that different organizational forms have in terms of impact measurement [66] based on environment, workers, community, and governance and managerial approaches and frameworks adopted for measuring the previous elements. The economic aspect actually has an impact on the reference community and internal elements of the organization in the hybrid organization model [66]. If different approaches have been identified to map and measure the impact of organizations whose boundaries are well defined, the literature cannot define numerous application cases and possible approaches of objective measurement within hybrid organizations [1,67]. The definitions provided also identify social cooperatives and hybrid organizations in the literature as typical examples of hybrid organizations in which, at the same time, interests oriented to generate profit from the private sector, respond to community needs as required by the public sector, and achieve social objectives or environmental as required mainly by non-profit organizations [46,68]. The two forms were the first to be fully configured as hybrid organizations. The third sector is increasingly configured as a hybrid organization withouthaving identified appropriate tools for measuring and evaluating the value generated by the impact of the activities often organized in collaboration with the public and private sectors. Several articles analyze how collaboration or partnership with the private sector can generate value [69,70] or destroy value [71]. Several articles suggest that the partnership between public and third sectors can generate value in the community context [72]. To the studies, there are also methodological proposals for mapping the social impact in the third sector [73,74]. However, there are no studies on how to determine the impact of hybrid organizations objectively [28,46]. The two studies highlighted the absence in the existing literature reviews of a common approach that considers mission drift and challenges to legitimacy in hybrid organizations. They recalled particular social enterprises as a classic example of hybrid organization [41] and the difficulty in defining the value generated by hybrid organizations—public and third sector in the definition of value for the territory and the reference community [23]. Hybrid organizations can generate excellent value, especially when welfare is transforming into a welfare mix in which the public needs the private sector and the third sector to achieve efficient and effective results in the social field under the principles of New Public Governance and New Public Management [75]. Moreover, if the public defines the objectives of general interest, approves the activities, finances them, or co-finances them and requires the collaboration or prevalent implementation by the third sector and the private sector, how can this generated value be measured? Esposito and Dicorato [76] highlight how the public sector needs to determine the impact of the resources used through the partners' ability to generate value, legitimize social activities, and guarantee operational and financial sustainability. The literature gap linked to both hybrid organizations and the evaluation of the value

generated is repeated and coincides with a mutual interest in enhancing the impact on the social and economic system.

For what concerns the partners' ability to generate value, according to Rhodes [77] and Osborne [78], this depends on the ability to build partnerships and inter-actions between different sectors (generally the public and the private ones), promoting cooperation as well as equity and democracy. Nevertheless, different approaches for what concerns the measurement of public value can be retraced overtime [79]. The public value generated by hybrid organizations can be determined through the social and economic impact [11,16,74]. Indeed, if in line with the Bureaucratic Public Administration (BPA) public value was evaluated in terms of legitimacy and formal correctness [80], the New Public Management focused on its economic dimension [81]. On the other hand, the theories of network governance stressed the role of interdependencies for multi-actor collaboration and how meta-governance can manage and provide direction to such networks [82–84].

In this sense, co-production assumes a significant relevance for public value production and engages citizens not only in voting [85], but also implies a redistribution of power between professionals and citizens, pointing out important issues related to accountability [86]. In order to co-produce public value, different actors have to share the same perception of public value and then they have to develop specific capacities such as being able to overcome significant gaps, divergent resourcing, and differing time frames. Emerging literature has identified different mechanisms to manage potential conflicts between actors at different levels [87–89] and, among these, hybridization appears as the crucial one by sustaining distinct policies and practices that pursue competing values.

Kelly et al. [22] studied public value on the basis of its three "building blocks," represented by services, outcomes and trust. According to such an approach, the aim is to generate outcomes in society by providing public value through services. Furthermore, other authors have started to focus on new and intangible aspects, such as citizen participation and the need to satisfy their demand and overcoming the previous public sector paradigms [90–92]. This new attention can be seen as the main contribution provided by the Public Value theory to evaluate the value generated by hybrid organizations.

Lastly, even with the aim to measure performance, Faulkner and Kaufman [93] carried out a model based on four different dimensions which can be adapted to several contexts, including hybrid organizations. The first dimension is related to social, economic, cultural, and environmental outcomes [4], while the second one is about trust and legitimacy, since public activity must be legitimized by all stakeholders involved, stimulating trust [91,92,94]. The third dimension takes into consideration the quality of public service, which is a crucial aspect in the provision of public services and involves satisfying users' needs [95–99]. The latter is dedicated to efficiency, with the aim to minimize costs [100], bureaucracy [101], and value for money [102].

As anticipated, a second current of theories tends to evaluate the impact of social activity through social legitimacy [76], highlighting the importance of specific factors such as autonomy, trust, steering, and ability to influence performance [103].

Performance represents a complex concept to define, especially for what concerns hybrid organizations. While for-profit organizations measure it in terms of outputs—taking into account their effectiveness, efficiency, and productiveness—for non-profits it is harder to quantify goals, since they operate in less competitive environments which makes benchmarking more difficult [104,105].

From a literary point of view, there are different opinions for what concerns the effect of an organizations' autonomy on performance. I On the one hand, autonomy can create transaction costs which negatively affect performance [106,107]. On the other hand, autonomy can be beneficial, valuing citizens' satisfaction rather than efficiency. According to Cambini et al. [108] and Swarts and Warner [109], autonomization ensures cost savings, but it depends on the sector where it takes place [110].

Generally, strategic steering is associated with positive effects on performance, since it can help communicate what level of performance is expected, to assess whether objectives

are achieved and to align management's interest with that of governance establishing sanctions and rewards [111,112]. But, according to Benabou and Tirole [113], extrinsic motivation through incentives can replace intrinsic motivation, generating distrust.

The last approach that can help in assessing hybrid organizations' performance suggests looking at the concept of sustainability, considering not only the social and environmental impact related to activities but also wider issues such as climate change, social stability, job creation, and the protection of human life [114,115]. According to Mahadi and Sino [116], understanding the notion of public value represents the first step to achieve sustainable development and deliver services satisfactorily. Sustainability represents a fundamental driver for public–private partnership activities, allowing the pursuit of public and private partners' goals without compromising the capacity to meet the needs of their reference communities. The same could be said for hybrid organizations, which overcome the risk of conflicting logics and ensure a positive social impact.

#### 3. Method Approach

In Italy, the civil service represents a typical example of the transfer of public funds to cover general interest activities that involve users present throughout the national territory. The civil service has similar characteristics in other European countries. In Italy, civil service is voluntary and can be carried out as an alternative to military service as in France, Germany, Spain, Holland, Poland, Portugal, United Kingdom, the Czech Republic. In other countries such as Austria, Cyprus, Denmark, Finland, Greece, Latvia, Lithuania, Slovakia, Sweden, Switzerland, Norway, it is mandatory [117]. On 3 April 2017, the Civil Service in Italy was established with the characteristics currently in force according to the Italian Legislative decree 6 March 2017, n. 40. The Universal Civil Service, "is aimed at the non-violent and unarmed defense of the homeland and peace between peoples" and, "constitutes an institution of integration, inclusion and social cohesion, aimed at strengthening the relationship between the citizen and the institutions of Italian Republic, contributing to the stability of democratic institutions as well as to the construction of a participatory democracy and new forms of citizenship." This experience exclusively dedicated to young people between 18 and 28 years can be carried out in Italy or abroad, both in member countries of the European Union and others. Participation is bound to adhesion to a project, presented by a third sector, which can last from 8 to 12 months. The organization which organized the universal civil service is an excellent example of a hybrid organization as the selected projects have been taken advantage of and are co-financed by the Department for Youth Policies of the Italian State. The Department is partner and investor of the organization. The accredited body evaluates the impact obtained by the various projects in response to the Department's general objectives, and the offices that welcome the subjects involved are part of the third sector and non-profit. Therefore, there is an extension, control, and public interest that involves different organizational forms with non-uniform types of financing. The case study is subject to public contributions and public control. Having to respond to reporting purposes of a public nature and public accountability, it is itself a hybrid organization due to the principle of the prevalence of economic substance over legal form [118]. The case study considered is significant for the analysis that we intend to conduct because the Vol.To volunteer service center is one of the third sector bodies that welcomes and coordinates the largest number of people involved in the civil service. Vol.To was accredited as a National Civil Service Body in 2005 by enrolling in the Register of the Piedmont Region (Italy). Over time, the number of its reception centers increased to become he only private social body registered in the first class in the Regional Register (with several offices greater than 100) in 2017. With the reform of the third sector that establishes the Universal Civil Service, Vol.To obtained accreditation for 140 reception centers in collaboration with 72 organizations (non-profit and ecclesiastical centers) defined by the "Reception Bodies" standard.

The civil service project represents in its configuration a significant example of a hybrid organization where public interests, objectives, and projects are shared, approved, and

financed or co-financed by the public sector to generate social value. The bodies accredited to host the civil service subjects are either public bodies or third sector. The projects involve third sector partners, public or private, for the performance of specific activities falling within the objectives of general public interest. The empirical investigation and analysis of the generated value of the hybrid organization of the civic service analyzed, as well as the project realized by Vol.To, was supported by the Bank Foundation and established by law as a point of regional reference [119].

Analysis of the empirical data followed a systematic combining approach based on an abductive process [27] which is characterized by the interplay between rich longitudinal empirical data and literature. The longitudinal analysis of a case study allows, through different methodologies and approaches, identification and explanation of the same phenomenon with different sources and visions and responding to the gaps that a single approach provides in literature. Examples of longitudinal analysis of a case study are defined equally in the literature [120,121]. The project is defined in four phases:

- (1) Analysis of the case study through Social Impact Assessment (SIA). SIA is the approach identified as a possible solution to mapping the process of hybrid organizations [25,26]. To analyze the approach, we refer to the phases of the value chain shown in the results section. The study recalls in the results the five phases for the impact assessment and the elements associated with each. These are input (human, financial and material resources), activities (transformation of inputs), outputs (products and services), outcomes (results and effects on beneficiaries) and phase impact (change in the reference community), which define the generated value [122,123].
- (2) Creation of a chronological overview of the interview and interventions in the examined public value based on empirical data from documentation and interventionist workshops to structure a new approach to measure the impact and social change. The interventionist approach was conducted through five meetings which took place from three months before the start of the project up to one month after the end of the project. The interviews were carried out on subjects who participated in the civil service at the end of 2020 and onmanagers and employees of Vol.To between 2019 and 2020 to identify the best approach to be adopted for measurement. The interventionist approach involved the participation of one of the authors defining the tools, elements, and methodology to be adopted to measure social impact and value generated through an agreement between the University of Turin (Italy), Vol.To and the Presidency of the Italian Council of Ministers. The activity was developed in 2020. Interviews with volunteers, were developed by semi-structured questions aimed at defining the actual change provided by the project through the counterfactual method [124].
- (3) Use of key concepts of the theoretical framework (SIA method) to link the empirical findings with measurement of impact in the hybrid organization. The data collected are available online. References were provided in open access in order to guarantee the replication of the result by the other researchers [125–127]. The determination of the social impact of the project described is part of an impact certification process by a European certifying third party. The approach is the derivation of a practical and theoretical training project which providedcertification of the skills on the social impact of the author who dealt with the interventionist analysis by the University of Turin (Italy) and a European skills certification body called CEPAS [128,129]. The approach on social impact assessment is one of those teachers and certified to one of the authors present in the SCH120 register held by CEPAS.
- (4) Analysis of the overall results to answer the research question.

The impact is the actual assessment of the change and the value generated. Therefore, it will be described and commented on in the Discussion as a logical element of the approach's final determination.

The case study analyzes the fallout of the 2019 project of Vol.To completed in the year 2020.

Ninety-four people had access to the project, 36 of whom were males and 58 females, with 44 young people employed in Turin (Italy) and 48 in the Province of Turin. The selections rewarded graduates who have risen to 59, 6% (56 young people) and young people with low schooling were penalized, which fell to 13.8% (13 people). Graduates which slightly decreased, are 26.6% of the total number of operators who participated in Vol.To. Vol.To's civil service projects support the activities carried out by the third sector, which can strengthen their support for the reference communities with the help of volunteer operators involved. The needs and the projects activated identify 25.7% cultural animation activities for minors and young people, followed by initiatives aimed at users with disabilities (20%) and support for the elderly (17.1%). The other activities aim to support adults in distress, including women in difficult or poor health conditions, school tutoring, peace education, civil rights, and environmental protection activities. In Table 1 and Figure 1, all partners and direct stakeholders of the project interested in increasing their workforce or covering voluntary activities were already active with a transfer of values and skills.

**Table 1.** Hosting third sector organizations according to the classification defined by the Legislative Decree Italian 117/2017.

Type of Organization	Percentage Distribution
APS (associations for social promotion)	8.6%
ASD (amateur sports associations)	2.9%
Association	8.6%
Social cooperatives	11.4%
Foundations	8.6%
Odv (volunteer organizations)	57.1%
Ecclesial bodies	2.9%



Figure 1. The host organizations of civil service Vol.To.

The representation of the impact of activities and stakeholders is made even more complicated as each third sector entity has one or more public, private or third sector entities as a partner (Appendix A).

# 4. Empirical Evidence

This section presents the interviews, the proposed approach for evaluation and enhancement of personnel, and the impact as conclusive elements of the SIA analysis in support of the study's empirical and theoretical tips.

# 4.1. Input

The project involved an overall investment of  $\pounds$ 42,911 against a ministerial loan of  $\pounds$ 7740. Three trainers were employed for general training: Rosanna Lopez: responsible for managing volunteers, Vol.To. Maida Caria: accredited trainer, Vol.To. Sandro Prandi: accredited trainer, Vol.To. Experts in general training were also employed concerning specific necessary skills not possessed among Vol.To employees: Nice Law Firm Stefano Lergo: civil protection expert Enrico Bussolino: third sector expert. They were involved after general training of trainers for specific projects and on average each project required six trainers, for a total of 185 trainers, of which 87 were male and 98 were female, with an average age of 51 years.

#### 4.2. Activities

The activity carried out included acquiring generic skills on preordained topics and specific training provided within the host organizations with work activities in the project context defined for each. The generic training considered four macro-areas with a total duration of 42 h for each class (consisting of a maximum of 25 people each) delivered in the first 180 days to provide useful elements for understanding and elaborating the experience of the Civil Service. The macro-areas treated customized based on the project of interest concern the values and identity of the national civil service, active citizenship, and the young people within the civil service context. The frontal lesson (40%) is the classic technique for teaching, where the trainer deals with a specific topic using his/her studies and experiences in the field of civil service and related issues. Support of diversified instruments includes documentary screening, PowerPoint presentations, reading of texts, and testimonials from external experts. Non-formal dynamics (60%) includes the most interactive techniques between the trainer and the group and between the members of the group itself such as role-play, simulations, plenary discussions, group work, expressiveness workshops, and sharing of personal experiences. Generic training supports the experience, although specific training and experience in the civil service are more characterizing and have a bigger impact. Each project required a period of specific training on project activities for a minimum of 50 h with teachers who are experts in the subject with many years of experience and/or a degree relevant to the activities envisaged. All the organizations involved chose to administer more hours than the minimum required. On average, the specific training lasted 75 h with a minimum of 72 h and a maximum of 108 h, delivered within the first 90 days of the project. On average, each project required six trainers, for a total of 185 trainers, of which 87 were male and 98 were female, with an average age of 51 years. It is sufficient to refer to the project activities indicated in the table below. In addition, each volunteer was accompanied by a tutor, who monitored and supported people (a minimum of 10 h per week). The need to which Vol.To responds through the project activities includes an activity equal to 30 h a week for 12 months. The areas of action concern minors and young people's cultural animation, organization and management of workshops for children, and participation in events for children and families (Figure 2).

The individual specific activities for each project falling within the defined activities of general interest can be found in Appendix B.

To be able to map the change in skills and knowledge of each subject, the civil service project envisaged participation in a skills certification path in collaboration with the University of Turin. The path started in parallel with training and field activities involving the subjects in five steps. Certification of skills is part of a project funded by the European Union and adopted for Erasmus+ courses' certification. The reference model used was born from a European LEVER UP project that was created in line with the Non-formal/Informal Learning Validation (NFILV), and Validation of Prior Learning (VPL) approaches. Creation of the reference model was created by partners between 2014 and 2016 in Italy, the Netherlands, Denmark, Spain, and Poland [130]. LEVER UP was created to help individuals enhance and make visible transversal skills and competencies acquired through non-formal and informal learning experiences, for example, by volunteering. This allows them to

increase their awareness, social responsibility, employability, and mobility. The first step is to create awareness of the value of the experience carried out in the volunteer/civil lawyer from a human and educational point of view. The key factor in this step recognizes that skills can be learned in informal settings, and volunteering is one of them. Each experience allows a person to learn specific knowledge and skills, which allows for skill development. The first step was conducted through an interview.



Figure 2. Percentage of Civial Project Activities.

In the second step, the subjects were helped by their tutors and had to reflect on what they have learned and the skills they developed during the year of Universal Civil Service. To understand which skills were acquired, the Lever Up tools were used, which allowed for a comparison between their level of competence and that which is described in the European standards. In the second step, the subjects were helped by their tutors. They had to reflect on what they learned and the skills they developed during their year of Universal Civil Service. The Lever Up tools were used to define the skills that were acquired, which allowed for comparison of their competence level with that described in the European standards. The third step involved documentation and collection of evidence on the skills acquired through various types of materials (photos, audio, video, reports, presentations, letters, storytelling). There was also an optional part in which it was possible to declare how the chosen skills related to school, work, volunteer work, or other experiences. The documentation and evidence collected were evaluated by an external commission, composed of Vol.To and the Faculty of Psychology, which analyzed the documents and evidence presented based on pre-established criteria and based on the level of complexity experience. In the last step, the subject acquired a greater awareness of his/her skills. The growth of awareness about these abilities will be useful to the volunteer to strengthen their self-esteem and for personal development. Finally, a certificate of validation of Lever Up skills will be issued, which can be used in various areas, such as work.

#### 4.3. Output

The output included the 94 trained volunteers, the 42 h of generic training and the 75 h of specific training, the number of subjects who participated in the certification of skills and the project paths carried out.

# 4.4. Outcomes

The outcome achieved refers to participation in continuing training. The outcome is defined by the percentage of people who followed continuing vocational training after graduation. The outcome is related to the change in the skills and behavior of the subjects. Moreover, among the outcomes is also the impact of the number of hours of activities carried out thanks to the civil service in addition to those that the host subjects already carried out. Of the subjects in this study, 32.4% managed to validate at least three competences with appropriate evidence as required, while 44.1% preferred to focus exclusively on one of the 15 competences provided by the standards. The remainder did not complete the validation process.

# 4.5. Social Impact Assessment

The self-assessment associated with the certifications acquired highlights a first approach linked to personal change. At the end of the course, participants were provided with a questionnaire to map this aspect. The participants found a behavior change in almost all cases thanks to the skills and the path taken.

Furthermore, for 93.8% of participants, the change was good (21.5% good, 27.7% excellent, 44.6% excellent) while only 4.6% were considered sufficient and 1.5% insufficient. Of all the participants, 96.9% believe they increased their skills, are oriented to work on the project, and were able to identifythe soft skills that the project intended to instill in participants. The indicator is above the average as the 96.9% achieved by the Vol.To project exceeds the national average for the same parameter in the previous year by 15.8 percentage points. The certification of skills certifies the subject's ability. The second impact is linked to the activities carried out and generated 1440 h for each subject within the host institution, impacting the general population through activities of general interest.

The real novelty is being able to map the impact of the activities on the territory, which, through specific actions, respond to the public interest (or general interest) de facto in integrating state public activities.

#### 4.6. Interventionist Journey, Interview Results and Evaluating Volunteer Activity Approach

The development and definition of the approach adopted for measuring the added value and the consequent social impact generated saw the collaboration of one of the authors with four key employees of the voluntary service center Vol.To who deal with deciding and developing the project and an official of the Italian council of Italian Council Presidency responsible for the project.

Based on previous experiences, the president of Vol.To had the need to identify and map the impact of civil service project. In particular, he declared, "I am sure the civil service has potential repercussions that must require an objective evaluation of the value generated given that the center co-finances the project." The need to map the impact and determine the added value is also stated by the official of the Italian Council Presidency who said, "We need to know if the objective of the project approved and shared by our offices has been achieved and what the results are based on to the public resources transferred." Duringvarious meetings, the two employees who deal with the definition of objectives, activities, and results highlighted two main aspects: "We need to identify the impact of training towards objectives of general interest and real change" and, "We understood that the activities in collaboration with the host third sector organizations have an impact on the whole system that allows to carry out activities that otherwise would not have been provided." No one has highlighted the difference between partner organizations participating in the project, different interests, or internal conflicts. With respect to the specific question concerning the legitimacy of the hybrid organization configured in the project during the design phase of the approach to be adopted to evaluate the impact and relative value generated, the president of Vol.To states, "There are no problems with the legitimacy of collaboration between organizations with different objectives because they all converge towards a common goal that leads to greater welfare without creating real differences between all those involved." Indeed, all the subjects have taken for granted the sharing of the project's formal and informal objectives within a hybrid organization that sees the project presence of different subjects with opposite legal forms. All highlighted the importance of mapping the added value generated and determining the impact of the funding from a perspective of efficiency and analysis of the change generated, in this case by the subjects selected by the civil service.

The cost effectiveness and reliability of the measure are two fundamental requirements for measuring volunteering activities and confirm the national need for an international framework to measure volunteering activities [131]. The need to effectively identify and add value can be achieved through interviews with civil service volunteers. In agreement with all the subjects, the activities and changes, skills and abilities detected by each subject were evaluated. Some examples of interviews can be viewed at [125] but will be summarized in the outcome section. However, the method is subjective and not objective and cannot determine the enhancement of the volunteer activity's impact during the civil service on the system. The Italian Ministry of Labor and Social Policies already proposed a methodology to determine the value of volunteering in the third sector. The proposed scheme defines an indirect replacement method that allows identifying the hourly value of volunteering carried out to allow an objective quantification. During the periodic annual meetings, the interventionist approach made it possible to identify that the indirect method was the best one for determining the impact on context. According to the SIA analysis, all elements led to the quantification of the impact. The replacement cost can be determined through the following method:

UWIFR = ∑t HiVi Wi

UWIFR = replacement cost for single function

Hi = average hours worked by volunteers in function i

Vi = number of volunteers who performed the function i

Wj = average salary applicable to function i

The approach identifies the real value of the volunteers' hourly activities based on the type of service carried out at the host organizations or at private or public companies that are partners of the initiatives. The determined value is shared within Vol.To and accepted by the Ministry which approved it as an objective value for determining the hours of volunteering. The determined value is shared within Vol.To and accepted by the Ministry which approved it as an objective value for determining the Ministry which approved it as an objective value for determining.

# 4.7. Understanding Social Impact and Measuring New Value Creation

This section proposes the desired approach determining the added value linked to the specific activities carried out and the related social impact. The project's economic value starts from the evaluation of income and expenses related to the project. The revenues include public transfers linked to the recognition of expenses and revenues for consultancy to third sector entities that received the subjects in charge.

For the development of its social activity, Vol.To has generated economic value created which is distributed to human resources through the payment of wages and all related charges. Suppliers were remunerated due to the purchase of products and services necessary for the production of the services and management costs related to the structure. The value relating to personnel costs includes 68.40% of the activities carried out by the head of the Civil Service project in the figure of Dr. Maida Caria, for 17.63% of the activity carried out for the selection of participants in the course, for 12.02% linked to the activities of the internal trainer and a residual part for contracted external training. The head of the Civil Service's activities concerns the entertainment activity of institutional relations

with the region, the Department of Youth Policies, the university, the high schools and the Foreign Office of the City of Turin. The manager also managed the project sites' accreditation activities, planning, monitoring, research, selecting volunteers, maintaining relations with 35 tutors, and providing general training for 111 subjects (42 h for four classes). The activities of the head of the Civil Service lasted eight months.

The person in charge of the selection activities is configured with the management of institutional communications relating to the projects and participated in the selection interviews for a duration of two months. Simultaneously, the training phase that saw the specific design for the training course and delivery can be qualified with one month of activity.

Training activities related to 1.95% of the performance carried out by a safety expert (4 h module), activities of a communication expert (5 h module), and assistance for the legal part carried out by two lawyers (5 h).

The remuneration of suppliers, on the other hand, is divided by 96.94% into printing services for material used for the training course (roll-up printing, postcards, A3 posters, gift vouchers, plasticized signs for each organization, riders for each organization, A5 flyers, and printing personalized T-shirts for each volunteer) and the residual part for the hospitality offered to the Civil Service party. The management costs are completely related to expenses for the general training room used (42 h for four courses) and expenses for using the workstation equipped with PC (Office) for eight months.

The activities represented in the reclassification of financial items can also be assessed through a questionnaire administered to civil service volunteers at the end of the course. From the data obtained, another degree of satisfaction is highlighted concerning all the aspects evaluated. In analyzing the financial statement, there is no real distributed added value as the costs are higher than the project's revenue. Vol.To has covered the difference of €25,711 with its own operating income. In proportion, the project absorbed 1.32% of the total fund. Table 2 represents the financial statement of income and expenses.

Element	Values in € Year 2019
Revenues from consultancy institutions	9400.00
Transfer for reimbursement of training expenses from the Department of Youth Policy–Italy Government	7740.00
Economic value	17,140.00
Remuneration of suppliers	655.00
Remuneration of personnel and utilities	40,936.00
Management fees	1390.00
Economic value distributed	41,911.00
Economic value absorbed	25,711.00

Table 2. Prospect of income and expenses.

From a first analysis, the civil service project absorbs rather than generates value, but this is wrong if one evaluates the impact and not the single reality. However, the volunteer activity carried out on the territory can be defined as the real distributed impact that did not represent financial terms. A total of 94 volunteers were employed, of which 36 were males and 58 females. In general, the qualification possessed by the majority of participants is in the Higher Middle School License (56 volunteers), followed in order by the University Diploma (14 volunteers), the Lower Middle School License (13 volunteers), and Degree (11 volunteers). Of 94 volunteers, 44 were employed in the city of Turin, and 48 in the Province of Turin.

Therefore, the impact of the civil service in economic terms includes the activity of 94 volunteers located throughout the region with a hypothetical remuneration of €17.11 per hour recognized in terms of enhancement of voluntary activities in the project. Volunteering

hours are recognized through the tabular criteria previous described (Section 4.2) associated with the fifth level of contributions (educator, hours without title, head worker, head cook, home assistant and tutelary services operator, hours of social assistance involved in basic assistance or otherwise defined, coordinator hours, teacher or manual and expressive activities, guidance with programming tasks, masseuse, entertainer, general nurse or childcare assistant with educational functions).

The same criteria were adopted by Vol.To on the occasion of the 2011 Single Deadline Call included in the financing activities of the projects of volunteer associations. The recognition and enhancement of voluntary work is a practice aimed at enhancing the third sector's contribution, methods, and push towards European recognition which took place by CSVnet in the national association of service centers for volunteering (CSV), which includes 62 of the 63 Service Centers for Volunteering.

Each volunteer worked 1440 h in one year. Therefore, the volunteers added value in the unrecognized territory for €24,638.40. Net economic contribution due monthly to each volunteer was equal to €439.50 per month and €5274 per year, which is equal to €19,364.40 for each volunteer. If the enhancement considers the number of active volunteers involved in the project, the value generated should be equal to €1,820,253.60. If we consider the value absorbed by Vol.To for public transfers and recognition by third sector entities, the net value is equal to €1,803,113.60. This value is further decreased by the economic value (net of transfers and contributions) absorbed for realizing the project (equal to  $\xi 25,711$ ), which brings the quantification of the activities carried out to €1,777,402.60. This is a value that does not find correspondence in financial result, but which is distributed throughout the territory. Therefore, out of 2500 Third Sector Entities accredited by Vol.To, the national civil service contributes €710.96 to the workforce for each. With 35 organizations involve in the project, the value for each organization of the hours worked would be equal to €50,782.93. According to the data, this activity has an impact on the provinces of Turin, Vercelli, Cuneo, and Asti. In 24 cases, there is expectation of a collaboration with public bodies such as civic libraries, hospitals, municipalities, schools, universities which would increase the overall value.

# 5. Discussion

The realization of the civil service through new organizational forms responds to the highlighted need to reduce available resources and obtaining a better impact in terms of performance [22]. The evidence provided by this case study analysis highlights how the social impact assessment and the breakdown into five phases of the analysis guarantees a real ability to determine and map the change up to the evaluation of the value generated through the quantification and qualification of social impact [25,26]. The process is innovative because SIA is applied together with interviews, interventionist analysis, and methodological determination of a hybrid organization's value. Vol.To is configured as hybrid organizations that directly carry out a coordination activity based on public funding and collaboration and by the host subjects (third sector, private and public partners) that involve the adhering subjects. The observation interventionist and the interviews confirm how hybrid organizations on specific welfare and social activities are able to overcome institutional limits with a capacity for dialogue between the different forms that strengthen the achievement of objectives [22,117,118]. Public funding that necessarily requires the involvement of other subjects in order to effectively respond to the need of the context for the allocation of voluntary resources of civil service on the territory on specific activities accompanies a vision of prolongation of interests and needs towards subjects with different forms, where business models and organizations share purposes and methods. The interviews highlight the shared need of the two financing organizations (Presidency of the Italian Council of Ministers and Vol.To) to define the territorial impact and value generated by highlighting an approach capable of mapping changes. In the literature, analyses of sustainability approaches and models focus on a single organization, proposing valid models exported to other types of participating organizations. In the case analyzed, the
approach analyzes cross-cutting elements such as social change, welfare, and educational activities carried out, and the economic context's impact, providing a broader and more generalizable vision. The analysis also highlights the elimination of the potential conflict between different interests usually identifiable in the network. The minimization of costs is evident for the public sector given the co-financing of Vol.To although this approach highlights from the accounting documents a greater control of inputs and outputs planned upstream and shared by all subjects with less bureaucratization of access to resources and the effective generation of value [11,14,16]. It is not possible from the analysis to state that the hybrid organization is better than direct management by the public sector. However, it is indisputable to affirm that the third sector's reform and the need to map the resources provided are due to an awareness of the limits already highlighted by the new public governance on the management and evaluation of performance between public and private entities. At the same time, the public sector alone would not be able to trace the market need, would not be able to identify the right placement of the subjects adhering to the civil service, and would not be able to provide proper training and assistance in the path of individual growth without stealing resources from a sector where austerity has already led to a reduction in staff in search of efficiency [30].

The orientation towards a common interest of public service leads to the determination of objective indicators capable of supporting and supporting the financing decisions and determining shared projects within hybrid organizations.

#### 6. Conclusions

Through theory and case study, the study highlights that hybrid organizations generate social and economic value thanks to the change generated by the activities carried out. This value is determined in social-educational and economic well-being that would not have been successful without hybrid organizations. Therefore, the study confirms the first definition and configuration of generated value that is attributable to organizations oriented to the public interest [16,21]. The value generated is often an intangible value such as the change in skills or the economic fallout that does not present reporting by the organizations themselves. The case study represents for practitioners and academics an excellent starting case study to highlight all the cases in which public funding between subjects generates hybrid organizations for the principle of the prevalence of economic substance over legal form [118]. The social impact assessment adopted within the study provides a useful tool to map and determine the change given by the impact and the real value, which can be objectively quantified. The approach adopted can be generalized to any business and contributes at the literature on hybrid organization and social impact. Social impact assessment is a necessary tool to determine the value creation by hybrid organizations. In the case examined without the analysis, the hybrid organization had the determination of the outcome without being able to assess the real value creation that was determined on the change in individual skills subjects (social change) and the economic value distributed on the territory. Therefore, social impact assessment is one of the tools that public administrations and hybrid organizations must use to allocate resources and objectively enhance public interest activities inside the new complex organizations. The breakdown of impact assessment through the five elements of the value chain (inputs, activities, outputs, outcomes, and impact) guarantees a linear definition of the value generated through change with procedural objectivity capable of grasping hybrid organizations' complexity. The value generated or absorbed is the change generated by the impact measured based on the incidence of public resources allocated. The case study analyzed does not allow to generalize the metrics adopted for inputs, activities, outputs, results and impact. The impact that personalization requires is based on the context and objectives of interest and quantifying each country's added value [122,123]. The impossibility of generalizing the precise metrics but only the approach generates several research questions aimed at identifying indicators and measures that can be shared for each sector or social, environmental, and educational project. The study is part of the debate on evaluating

efficiency, effectiveness, and external context need of hybrid organization, providing new stimuli and reflections through the literature and case studies on value creation. Evaluating the performance, impact, and generated value of hybrid organizations initiates a debate within the area that is interested in post-modern public management. The analysis could begin a debate on the relationship between generated value and sustainability in the light of reducing expenditure and the response to epidemic crises.

Civil service impact assessment is common to many nations and requires a reproducible methodology for assessing change and generated value. The two approaches defined the impact in terms of skill change and individual approach with a certified change verification, and the real impact and economic value of the project concerning what was transferred and absorbed, which highlight real sustainability of the activity realized by the particular form of hybrid organization in the welfare system. The hybrid organization has an impact in social terms (for the change in the behavior of volunteers) and economic terms (for the value generated and the workforce made available to the third sector and private entities). It also increases the ability to respond to the questions of the market of social, welfare, and educational services envisaged by the project with the sharing of expectations, objectives, and cost-effectiveness required by the public sector, which is the project partner itself.

#### Limitations and Future Research

The study carried out presents only one case that should be analyzed in similar international contexts, although civil service projects highlighted characteristics and organizational change between adhering subjects are common. The approach adopted is suitable for analyzing the civil service's impact and value but could change metrics according to each hybrid organization's specificity. Future studies should highlight the impact of hybrid organizations and their generated value. When the literature is rich enough, it will be possible to determine common approaches to organizational clusters. The empirical evidence suggests that the approach could be guided by the common purpose of all the subjects that come together to achieve a common goal. This hypothesis should be refuted by new evidence.

**Author Contributions:** Conceptualization, P.E. and V.B.; methodology, P.E. and V.B.; validation, P.E.; formal analysis, V.B., C.F. and R.F.; investigation, V.B.; data curation, V.B.; writing—original draft preparation, V.B., C.F. and R.F.; writing—review and editing, V.B. and P.E.; visualization, V.B.; supervision, P.E.; All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: The data presented in this study are available on request from the authors.

Conflicts of Interest: The authors declare no conflict of interest.

### Appendix A

Organization	Partner
A.I.B. Volontari Antincendi Boschivi-Sez. Bussoleno	A.S.D. ANIMA LIBERA: teaching of laboratories and courses A.C.F. MUSIC-ALL:singing lessons, show assembly.
A.S.D. L'isola che c'è	City of Druento: consortium of social assistance services of Pianezza-Day Center Management
Amici di Lazzaro	/
Amici di M.A.I.S.	M.A.I.S. Ong-CIFA Ong-RE.TE. Ong-Collaboration during workshops and in communication and promotion activities. Cascina Roccafranca e Associazione Variante Bunker: collaboration in the promotion and dissemination of materials, making available spaces for activities.
Area G Volontari	/
Associazione Don Bernanrdino Reinero	Associazione di Volontariato "Libro Aperto": collaboration for the creation and organization of events and workshops.
Auser Volontariato	Municipality of Turin and the police: transport network at affiliated hospitals and healthcare facilities. Cooperativa "La Valdocco": delivery of water to users reported by the cooperative and also organization of recreational activities in air-conditioned environments. City of Carmagnola-Transport and accompaniment of children with disabilities CISA 31-Transport to health facilities Municipal Pharmacy: free delivery of medicines for people over 75 or with walking problems. School complexes transport and accompaniment of pupils with handicap for playful support activities and psychomotor activities.
Banco Alimentare	/
Cascina Macondo	Associazione C.P: conducting the folk dance workshop-pedagogy and history of folk dance. Gruppo musicale Lumayna: use of the voice-popular song-rhythms-musical games. Municipality of Torino:search engine project-accreditation at the Ingenio Shop for the sale of artifacts and books; logistical support for the organization of events and dissemination of the project;contribution. VI circumscription di Torino: loan for the use of a small gym for the dance; theater workshop in Turin; contribution to the ceramics and storytelling workshops at the Via delle Querce facility where disabled people are accompanied by educators and OS of the Municipality
Contact	/
Cooperativa Quadrifoglio	Associazione Culturale Balancé: specific consultancy with its experts for the creation and monitoring of laboratories.
Cooperativa Quadrifoglio	Associazione Culturale Balancé: specific consultancy with its experts for the creation and monitoring of laboratories.
Cooperativa Quadrifoglio DUE	Associazione Culturale Balancé: specific consultancy with its experts for the creation and monitoring of laboratories.
Croce Rossa Susa	Cities: Piedmont Region: hospitals and reports; conventions and agreements as needed.
DI.A.PSI Difesa Ammalati Psichici-PIEMONTE	/
Don Bosco 2000	/
Fondazione Circolo dei Lettori	Civic library of Torino: loan of books and reading aloud in hospitals, making instrumental and human resources available. Associazione La Brezza Onlus: involvement in the preparatory phase, through the provision of human resources. Host organizations: Humanitas Gradenigo hospital, Mauri-ziano Umberto I hospital, Molinette hospital, Univoc Institute, "Lorusso e Cutugno" prison.
Fondazione Piazza dei Mestieri Marco Andreoni	Associazione Piazza dei Mestieri: organization and management of project activities. Immaginazione e Lavoro soc.coop: promotion of activities proposed by the civil service. Cooperative La Piazza: graphic design and printing Vanni Editore-promotion and distribution, training of civil service volunteers.

Organization	Partner
Jeshua	Image Capture: supply of materials. La Piccola Casa della Divina Provvidenza, Cottolengo Torino: support at the project ü F.I.S.M. Torino, Centro di Formazione Francesco Faa di Bruno-Formazione Scuola dell'Infanzia: provision of material and human resources and spaces.
L'Altra Idea	Associazione Sportiva Dilettantistica HIDALGO Onlus: implementation of equestrian rehabilitation projects, collaboration with qualified therapists.
L'Isola che non c'è	/
La vita al centro	Associazione Biodanza Italia-Biodanza Teacher Training Baby Fox Association:provided professional entertainers. La Casa di Riposo San Vincenzo-The Cooperative Terra Terra, Azienda Agri-cola Apenocciola: provided an animator available for the creation and management of the G.A.S.
Legambiente Metropolitano	Istituto per l'Ambiente, Educazione Scholé Futuro Onlus: organization of workshops on air pollution and traffic control, environmental education activities in schools.
Merope	A.Ge. Piemonte: support for relationships with families. Ammp Sede Val Sangone: support in case of disabled transport. Echos Communication: support in the communication action.
Parrocchia Cottolengo	Ente Piccola Casa della Divina Provvidenza: to transmit these values to the minors and young people of the SGB Cottolengo oratory, through training meetings with operators and moments of sharing.
Presidio Sanitario San Camillo di Torino	University Hospital City of Health and Science Turin, School of Medicine of the University of Turin-Implementation of medical rehabilitation projects and research activities University of Turin, Department of Psychology: shared research activity.
S.E.A.–Servizio Emergenza Anziani	PRO NATURA Torino-Promotion of events, guided tours of the area. Ass. TELEHELP Turin-Te-rescue and telephone assistance. UNI.VO.C.A. Unione Volontari Culturali Associati: cultural events aimed at the elderly. ASSOCIAZIONE AMICI DELLA SACRA DI SAN MICHELE ONLUS: organization and provision of volunteer guides and witnesses for real and virtual visits.
S.E.A. delle Alpi	/
S.O.G.IT–Opera di Soccorso dell'Ordine di San Giovanni–I Giovanniti	S.O.G.I.TCOMITATO REGIONALE PIEMONTE: activity coordination function.
Società Cooperativa Sociale La Mimosa	/
Sollievo	A.S.D. ANIMA LIBERA: teaching of dance workshops and courses. A.C.F. MUSIC-ALL: singing lessons, show production and assembly.
Specialmente Tu	/
VIVERE–Associazione volontari e famiglie con figli Portatori di Handicap	FONDAZIONE FRIMARIDE ONLUS: communication, awareness of the project. FORUM DEL VOLONTARIATO: search for volunteers and awareness of the project.
Vivi gli animali	Department of the Sustainable City, Municipality of Collegno: educational and awareness-raising programs, organization and promotion of information meetings and recycling laboratories, use of natural resources and recycling. San Donato Social Cooperative, Abele Lavoro Social Consortium: integration of people in socio-economic difficulties, educational and awareness-raising courses Djanet Association: educational and awareness programs. Tricycle Association: educational and awareness paths, use of natural resources and recycling.
Vol.To	Permanent Interregional Forum of Piedmont and Valle d'Aosta Volunteering:contacts with schools, project promotion, teacher meetings, scheduling of meetings, participation in the final event. S.A.ASchool of Business Administration of the University of Turin: provision of a communication expert for training.

## Appendix B

Scope of Action	Type of Activity			
	- Organization and management of workshops for children			
	- Participation in the organization of events for minors and families			
	- Sports activities management			
	- Participation in the preparation of games, activities, sets			
	- Planning of individual activities to be carried out with children and families			
	- Organization of events and playful-recreational activities			
	- Organization and management of proposals for local schools			
	- Management of relations with external bodies			
	- Secretarial activities: management of parental registrations and authorizations, reporting of events			
	- Support for managing and maintaining contacts			
Cultural animation of minors and young people	- Event communication through social media			
)	- Educational, linguistic, and relational support activities and participation in business workshops			
	- Organization and participation in educational and recreational outings			
	- Organization and participation in school camps			
	- Supporting school educators/teachers in carrying out educational activities			
	- Search through the web for initiatives aimed at children			
	- After school support for school staff			
	- Management and coordination of groups during the educational workshop			
	- Support the Association's volunteers in information meetings in schools			
	- Map the public places of aggregation of young people and organize meetings			
	- Process data to improve the service			
	- Social telephony			
	- Accompanying the elderly			
Senior citizens	- Animation and organization of playful activities			
	- Support for socialization			
	- Handling of paperwork with online forms			
	- Listening to family members			
	- Design of ad hoc activities for individual users			
Disabled	- Organization and participation in individual and group outings with clients			
	- Participation in apartment group activities			
	- Management of creative workshops for the disabled (dance, poetry, reading, etc.)			

Scope of Action	Type of Activity			
<b>`</b>	- Search for new food donors and personal contact			
	- Monitoring of partnership requests from catering companies			
	- Drafting of agreements with the charitable structures			
	- Counter activities for collecting information, requests, needs			
	- Reception and listening to people in conditions of hardship			
	- Accompanying users in handling bureaucratic procedures			
	- Assisted medical transport			
	- Emergency services as stretcher bearers			
	<ul> <li>Promotion of the culture of emergency through active participation in training opportunities for schools and citizens</li> </ul>			
Adult discomfort and pathologies	- Supporting hospital staff during targeted interviews with patients and their families in order to understand their needs.			
	- Quarterly update of a report indicating the list of critical issues and patient needs			
	- Research and analysis of the offer on the territory in terms of structures, facilities, associations and services in general that may be of help to the patient who returns to their home			
	<ul> <li>Setting up moments of meeting between patients to encourage socialization and sharing their difficulties</li> </ul>			
	- Organization through the support of other professionals, of recreational activities			
	- Contact discharged patients by telephone having feedback on the perceived well-being linked to the results of the rehabilitation process concluded in the structure.			
	- Report through a specific report of the path developed during the project with the details for each patient taken in care			
Troubled women and mothers	- Anti-trafficking prevention and support for former victims and single women			
	- Support for entire families and very young people			
	- Research farms at Km. 0			
	- Management of the G.A.S.			
	- Planning and realization of laboratories			
	- Accompaniment in external activities			
Promotion of rights	- Realization of courses in schools (relationship with teachers and students)			
0	- Design of survey questionnaires			
	- Planning of interventions against discrimination			
	- Management of a class			
	- Realization and management of multi-ethnic events with students			
	- Mapping of intercultural events in Turin and its province			

Scope of Action	Type of Activity	
	- Monitoring of the territory and mapping of areas at risk of forest fire	
	- Monitoring of the territory and sighting of outbreaks in periods of high risk of forest fires	
	- Forest fire prevention activities with the cleaning of woods and forest areas	
	- Information and awareness-raising activities for citizens and the public administration on the issues of preventing and fighting forest fires	
	- Creation and updating of specific web pages	
	- Implementation of training programs with the schools in the Municipality of Bussoleno	
	- Organization of periodic network exercises between the public administration and the various associations	
Environmental protection and fire prevention	- Promote workshops on new mobility to create opportunities for exchange between the various bodies	
	- Stimulate the participation of citizens to dare new lifestyles and take care of air quality	
	- Promote and disseminate good practices with information material, dossiers and statistical reports	
	- Strengthen information and communication through the site, Facebook and YouTube	
	- Meeting, knowledge and care of farm animals	
	- Management of school visits and any hospitality	
	- Use of natural resources and recycling	
	- Care of the green and common areas	
	- Organization of work groups for people in socio-economic difficulties	
	- Scheduling of meetings in schools	
	- Preparation of materials needed for the lesson, computer programs, exercises, textbooks.	
	- Interviews with the teachers of the students followed alongside an educator	
	- Cognitive meetings with families	
	- Realization and administration of tests for the student	
	- Data analysis	
	- Study support activities	
Tutoring and school support	<ul> <li>Monitoring of the activities carried out, the conduct and performance of students</li> </ul>	
	- Presence to guarantee children with SEN access to work in small groups	
	- Presence in the after school and summer center to guarantee children with difficulties a positive experience in attending various activities	
	<ul> <li>Presence in the canteen to allow all children to experience a situation of well-being during convivial moments with a child/adult ratio below the legal limits</li> </ul>	
	- Collaboration with the management and administration to organize cultural events	

### References

- 1. Secinaro, S.; Corvo, L.; Brescia, V.; Iannaci, D. Hybrid Organizations: A Systematic Review of the Current Literature. *Int. Bus. Res.* **2019**, *12*, 1–21. [CrossRef]
- 2. Dionisio, M.; de Vargas, E.R. Corporate Social Innovation: A Systematic Literature Review. *Int. Bus. Rev.* 2020, 29, 101641. [CrossRef]
- 3. Jäger, U.P.; Schröer, A. Integrated Organizational Identity: A Definition of Hybrid Organizations and a Research Agenda. *VOLUNTAS Int. J. Volunt. Nonprofit Organ.* **2014**, *25*, 1281–1306. [CrossRef]
- 4. Bozeman, B.; Rimes, H.; Youtie, J. The Evolving State-of-the-Art in Technology Transfer Research: Revisiting the Contingent Effectiveness Model. *Res. Policy* **2015**, *44*, 34–49. [CrossRef]
- 5. Greve, C.; Flinders, M.; van Thiel, S. Quangos—What's in a Name? Defining Quangos from a Comparative Perspective. *Governance* **1999**, *12*, 129–146. [CrossRef]
- 6. Røiseland, A. Understanding Local Governance: Institutional Forms of Collaboration. Public Adm. 2011, 89, 879–893. [CrossRef]
- Christensen, T.; Lægreid, P. Complexity and Hybrid Public Administration—Theoretical and Empirical Challenges. *Public Organ. Rev.* 2011, *11*, 407–423. [CrossRef]
- 8. Christensen, T. Post-NPM and Changing Public Governance. *Meiji J. Political Sci. Econ.* 2012, 1, 1–11.
- 9. Ceolin, M. Il Nuovo Codice Del Terzo Settore Italiano (d. Lgs. n. 117/2017)/The New Code of the Third Italian Sector (Legislative Decree No. 117/2017). *Rev. Direito Cid.* 2018, *10*, 2856–2885.
- 10. Iannaci, D.; Aiassa, A. Adeguamento statuti delle ODV. Eur. J. Volunt. Community Based Proj. 2020, 1, 1–26.
- 11. Chiampi, A. Regime fiscale degli enti del terzo settore. Eur. J. Volunt. Community-Based Proj. 2020, 1, 89.
- 12. Billis, D. Towards a Theory of Hybrid Organizations. In *Hybrid Organizations and the Third Sector: Challenges for Practice, Theory and Policy;* MacMillan Education: London, UK, 2010; pp. 46–69.
- 13. Amelio, S.; Orlandini, P. Network di interesse pubblico e new public governance, il ruolo del terzo settore nell'equazione di governo. *Eur. J. Volunt. Community Based Proj.* **2020**, *1*, 23–38.
- 14. Cepiku, D. Performance management in public administrations. In *Handbook of Global Public Policy and Administration;* Klassen, T.R., Cepiku, D., Lah, T.J., Eds.; Routledge: Abingdon, UK, 2015.
- 15. Campra, M.; Esposito, P.; Brescia, V. State of the Art of COVID-19 and Business, Management, and Accounting Sector. A Bibliometrix Analysis. *Int. J. Bus. Manag.* **2021**, *16*, 1–27.
- 16. Esposito, P.; Ricci, P. How to Turn Public (Dis) Value into New Public Value? Evidence from Italy. *Public Money Manag.* **2015**, *35*, 227–231. [CrossRef]
- 17. Haldar, S. Towards a Conceptual Understanding of Sustainability-Driven Entrepreneurship. *Corp. Soc. Responsib. Environ. Manag.* **2019**, *26*, 1157–1170. [CrossRef]
- 18. Secinaro, S.; Brescia, V.; Calandra, D.; Saiti, B. Impact of Climate Change Mitigation Policies on Corporate Financial Performance: Evidence-Based on European Publicly Listed Firms. *Corp. Soc. Responsib. Environ. Manag.* **2020**, *27*, 2491–2501. [CrossRef]
- 19. O'Flynn, J. From New Public Management to Public Value: Paradigmatic Change and Managerial Implications. *Aust. J. Public Adm.* 2007, *66*, 353–366. [CrossRef]
- 20. Stoker, G. Public Value Management: A New Narrative for Networked Governance? *Am. Rev. Public Adm.* 2006, 36, 41–57. [CrossRef]
- 21. Ziruolo, A. Valore Pubblico e Società Partecipate: Tendenze Evolutive Della Performance; Franco Angeli: Milan, Italy, 2016.
- 22. Kelly, G.; Mulgan, G.; Muers, S. Creating Public Value. Lond. Cabinet Off. 2002, 1–35.
- 23. Ruddin, L.P. You Can Generalize Stupid! Social Scientists, Bent Flyvbjerg, and Case Study Methodology. *Qual. Inq.* **2006**, *12*, 797–812. [CrossRef]
- 24. Quélin, B.V.; Kivleniece, I.; Lazzarini, S. Public-Private Collaboration, Hybridity and Social Value: Towards New Theoretical Perspectives. *J. Manag. Stud.* **2017**, *54*, 763–792. [CrossRef]
- 25. Biancone, P.; Secinaro, S. La Valutazione Dell'impatto Sociale. Aspetti Metodologi; Pearson: Milan, Italy, 2020; ISBN 978-88-919-1948-9.
- 26. Becker, H.A. Social Impact Assessment. Eur. J. Oper. Res. 2001, 128, 311-321. [CrossRef]
- 27. Dubois, A.; Gadde, L.-E. Systematic Combining: An Abductive Approach to Case Research. J. Bus. Res. 2002, 55, 553–560. [CrossRef]
- 28. Battilana, J.; Besharov, M.; Mitzinneck, B. On Hybrids and Hybrid Organizing: A Review and Roadmap for Future Research. *SAGE Handb. Organ. Inst.* **2017**, *2*, 133–169.
- 29. Smallbone, D.; Evans, M.; Ekanem, I.; Butters, S. *Researching Social Enterprise*; Great Britain, Small Business Service: New York, NY, USA, 2001.
- 30. Holt, D. Where Are They Now? Tracking the Longitudinal Evolution of Environmental Businesses from the 1990s. *Bus. Strategy Environ.* **2011**, *20*, 238–250. [CrossRef]
- 31. Oxford English Dictionary; Oxford University Press: Oxford, UK, 1989.
- 32. Wilson, F.; Post, J.E. Business Models for People, Planet (& Profits): Exploring the Phenomena of Social Business, a Market-Based Approach to Social Value Creation. *Small Bus. Econ.* **2013**, *40*, 715–737.
- 33. Buccino, G.; Mele, S. The Hybrid Organizations: A Systematic Literature Review; 2019. [CrossRef]
- 34. Boyd, B.; Henning, N.; Reyna, E.; Wang, D.; Welch, M.; Hoffman, A.J. *Hybrid Organizations: New Business Models for Environmental Leadership*; Routledge: New York, NY, USA, 2017.

- 35. Trones, M. Hybrid Organizations: Defining Characteristics and Key Factors for Organizational Sustainability. A Qualitative Study from Latin America. Master's Thesis, Norwegian University of Life Sciences, Ås, Norway, 2015.
- 36. Besharov, M.L.; Smith, W.K. Multiple Logics within Organizations: An Integrative Framework and Model of Organizational Hybridity. *Ithaca NY Cornell Univ. Work. Pap.* **2012**, 1–56.
- Jay, J. Navigating Paradox as a Mechanism of Change and Innovation in Hybrid Organizations. *Acad. Manag. J.* 2013, 56, 137–159. [CrossRef]
- 38. Alter, K. Social Enterprise Typology. Virtue Ventur. LLC 2007, 12, 1–124.
- 39. Santos, F.M. A Positive Theory of Social Entrepreneurship. J. Bus. Ethics 2012, 111, 335–351. [CrossRef]
- 40. Haigh, N.; Hoffman, A.J. The New Heretics: Hybrid Organizations and the Challenges They Present to Corporate Sustainability. *Organ. Environ.* **2014**, *27*, 223–241. [CrossRef]
- 41. Grossi, G.; Reichard, C.; Thomasson, A.; Vakkuri, J. Theme: Performance Measurement of Hybridorganizations: Emerging Issues and Future Researchperspectives. *Public Money Manag.* **2017**, *37*, 379–386. [CrossRef]
- 42. Powell, W.W.; Staw, B.; Cummings, L.L. Neither Market nor Hierarchy; Sage Publications: Los Angeles, CA, USA, 1990.
- 43. Arellano-Gault, D.; Demortain, D.; Rouillard, C.; Thoenig, J.-C. Bringing Public Organization and Organizing Back In. *Organ. Stud.* **2013**, *34*, 145–167. [CrossRef]
- 44. Owen-Smith, J. From Separate Systems to a Hybrid Order: Accumulative Advantage across Public and Private Science at Research One Universities. *Res. Policy* **2003**, *32*, 1081–1104. [CrossRef]
- 45. Dunn, M.B.; Jones, C. Institutional Logics and Institutional Pluralism: The Contestation of Care and Science Logics in Medical Education, 1967–2005. *Adm. Sci. Q.* 2010, *55*, 114–149. [CrossRef]
- Doherty, B.; Haugh, H.; Lyon, F. Social Enterprises as Hybrid Organizations: A Review and Research Agenda. Int. J. Manag. Rev. 2014, 16, 417–436. [CrossRef]
- 47. Alexius, S.; Furusten, S. Enabling Sustainable Transformation: Hybrid Organizations in Early Phases of Path Generation. *J. Bus. Ethics* **2020**, *165*, 547–563. [CrossRef]
- Jolink, A.; Niesten, E. Recent Qualitative Advances on Hybrid Organizations: Taking Stock, Looking Ahead. Scand. J. Manag. 2012, 28, 149–161. [CrossRef]
- 49. Dahl, R.A. Polyarchy: Participation and Opposition; Yale University Press: London, UK, 1973.
- 50. Demone, H.W.; Gibelman, M. Services for Sale: Purchasing Health and Human Services; Rutgers University Press: New York, NY, USA, 1989.
- 51. Gray, B.H. Nonprofit Hospitals and the For-Profit Challenge. Bull. N. Y. Acad. Med. 1990, 66, 366.
- 52. Courpasson, D.; Dany, F. Indifference or Obedience? Business Firms as Democratic Hybrids. *Organ. Stud.* **2003**, *24*, 1231–1260. [CrossRef]
- 53. Koppell, J.G. *The Politics of Quasi-Government: Hybrid Organizations and the Control of Public Policy;* Cambridge University Press: New York, NY, USA, 2003.
- 54. Skelcher, C.; Mathur, N.; Smith, M. The Public Governance of Collaborative Spaces: Discourse, Design and Democracy. *Public Adm.* **2005**, *83*, 573–596. [CrossRef]
- 55. Brandsen, T.; Van de Donk, W.; Putters, K. Griffins or Chameleons? Hybridity as a Permanent and Inevitable Characteristic of the Third Sector. *Intl J. Public Adm.* 2005, *28*, 749–765. [CrossRef]
- 56. Evers, A. Mixed Welfare Systems and Hybrid Organizations: Changes in the Governance and Provision of Social Services. *Intl J. Public Adm.* **2005**, *28*, 737–748. [CrossRef]
- 57. Kirkwood, J.; Walton, S. What Motivates Ecopreneurs to Start Businesses? Int. J. Entrep. Behav. Res. 2010, 16, 204–228. [CrossRef]
- 58. Brunsson, N. Politicization and 'company-Ization'—On Institutional Affiliation and Confusion in the Organizational World. *Manag. Account. Res.* **1994**, *5*, 323–335. [CrossRef]
- 59. Alexius, S.; Grossi, G. Decoupling in the Age of Market-Embedded Morality: Responsible Gambling in a Hybrid Organization. *J. Manag. Gov.* **2018**, *22*, 285–313. [CrossRef]
- 60. Battilana, J.; Lee, M.; Walker, J.; Dorsey, C. In Search of the Hybrid Ideal. In *Stanford Social Innovation Review*; Summer: Stanford, CA, USA, 2012.
- 61. Alexius, S.; Furusten, S. Managing Hybrid Organizations; Springer: Cham, Switzerland, 2019.
- Denis, J.-L.; Ferlie, E.; van Gestel, N. Understanding Hybridity in Public Organizations. *Public Adm.* 2015, 93, 273–289. [CrossRef]
   Christensen, T.; Lægreid, P. Coordination Quality in Central Government–the Case of Norway. *Public Organ. Rev.* 2020, 20, 145–162. [CrossRef]
- 64. Pollitt, C. Bureaucracies Remember, Post-Bureaucratic Organizations Forget? Public Adm. 2009, 87, 198–218. [CrossRef]
- 65. Teisman, G.; van Buuren, A.; Gerrits, L.M. Managing Complex Governance Systems; Routledge: New York, NY, USA, 2009.
- 66. Gamble, E.N.; Parker, S.C.; Moroz, P.W. Measuring the Integration of Social and Environmental Missions in Hybrid Organizations. *J. Bus. Ethics* **2019**, 167, 1–14. [CrossRef]
- 67. Grossi, G.; Thomasson, A. Bridging the Accountability Gap in Hybrid Organizations: The Case of Copenhagen Malmö Port. *Int. Rev. Adm. Sci.* **2015**, *81*, 604–620. [CrossRef]
- 68. Chaddad, F. Advancing the Theory of the Cooperative Organization: The Cooperative as a True Hybrid. *Ann. Public Coop. Econ.* **2012**, *83*, 445–461. [CrossRef]

- 69. Villani, E.; Greco, L.; Phillips, N. Understanding Value Creation in Public-Private Partnerships: A Comparative Case Study. J. Manag. Stud. 2017, 54, 876–905. [CrossRef]
- 70. Wang, H.; Xiong, W.; Wu, G.; Zhu, D. Public–Private Partnership in Public Administration Discipline: A Literature Review. *Public Manag. Rev.* 2018, 20, 293–316. [CrossRef]
- 71. Esposito, P.; Ricci, P. Public (dis) Value: A case study. In *Public Value Management, Measurement and Reporting*; Emerald Group Publishing Limited: Bingley, UK, 2014.
- 72. Wood, C.; Leighton, D. Measuring Social Value: The Gap between Policy and Practice. Undercurrent 2010, 2, 7–9.
- 73. Lyons, M. Third Sector: The Contribution of Non-Profit and Cooperative Enterprise in Australia; Routledge: New York, NY, USA, 2020.
- 74. Enjolras, B.; Sivesind, K.H. The Roles and Impacts of the Third Sector in Europe. In *The Third Sector As A Renewable Resource for Europe*; Palgrave Macmillan: Cham, Switzerland, 2018; pp. 95–124.
- 75. Karré, P.M. 12 Value Creation by Two Types of Hybrid Organisations. In *Hybrid Governance, Organisations and Society: Value Creation Perspectives*; Routledge: New York, NY, USA, 2020; p. 62.
- 76. Esposito, P.; Dicorato, S.L. Sustainable Development, Governance and Performance Measurement in Public Private Partnerships (PPPs): A Methodological Proposal. *Sustainability* **2020**, *12*, 5696. [CrossRef]
- 77. Rhodes, R.A. Understanding Governance: Policy Networks, Governance, Reflexivity and Accountability; Open University: Philadelphia, PA, USA, 1997.
- 78. Osborne, S.P. The New Public Governance? Public Manag. Rev. 2006, 83, 377–387. [CrossRef]
- 79. Aveyard, H. Doing a Literature Review in Health and Social Care: A Practical Guide; McGraw-Hill Education: Berkshire, UK, 2014.
- 80. Borgonovi, E. Principi e Sistemi Aziendali per Le Amministrazioni Pubbliche; Egea: Milan, Italy, 2005.
- 81. Gagliardo, E.D. La Creazione Del Valore Nell'ente Locale. Il Nuovo Modello Di Governo Economico; Giuffrè: Milan, Italy, 2002; Volume 49, ISBN 978-88-14-09781-2.
- 82. Agranoff, R. Collaborative Public Management: New Strategies for Local Governments; Georgetown University Press: Washington, DC, USA, 2004.
- 83. Kickert, W.J.; Klijn, E.-H.; Koppenjan, J.F. Managing Complex Networks: Strategies for the Public Sector; Sage: London, UK, 1997.
- 84. Bryson, J.; Sancino, A.; Benington, J.; Sørensen, E. Towards a Multi-Actor Theory of Public Value Co-Creation. *Public Manag. Rev.* **2017**, *19*, 640–654. [CrossRef]
- 85. Alford, J. A Public Management Road Less Travelled: Clients as Co-Producers of Public Services. *Aust. J. Public Adm.* **1998**, 57, 128–137. [CrossRef]
- 86. Bovaird, T. Beyond Engagement and Participation: User and Community Coproduction of Public Services. *Public Adm. Rev.* 2007, 67, 846–860. [CrossRef]
- 87. Stewart, J. Value conflict and policy change. In Public Policy Values; Springer: London, UK, 2009; pp. 33-46.
- Van der Wal, Z.; de Graaf, G.; Lawton, A. Competing Values in Public Management: Introduction to the Symposium Issue. *Public Manag. Rev.* 2011, 13, 331–341. [CrossRef]
- 89. De Graaf, G.; Huberts, L.; Smulders, R. Coping with Public Value Conflicts. Adm. Soc. 2016, 48, 1101–1127. [CrossRef]
- 90. Cowling, K. Prosperity, Depression and Modern Capitalism. *Kyklos* 2006, 59, 369–381. [CrossRef]
- 91. Searcy, C. Measuring Enterprise Sustainability. Bus. Strategy Environ. 2016, 25, 120–133. [CrossRef]
- 92. Brown, P.J.; Bajada, C. An Economic Model of Circular Supply Network Dynamics: Toward an Understanding of Performance Measurement in the Context of Multiple Stakeholders. *Bus. Strategy Environ.* **2018**, *27*, 643–655. [CrossRef]
- 93. Faulkner, N.; Kaufman, S. Avoiding Theoretical Stagnation: A Systematic Review and Framework for Measuring Public Value. *Aust. J. Public Adm.* **2018**, *77*, 69–86. [CrossRef]
- Martín-de Castro, G.; Amores-Salvadó, J.; Navas-López, J.E. Environmental Management Systems and Firm Performance: Improving Firm Environmental Policy through Stakeholder Engagement. Corp. Soc. Responsib. Environ. Manag. 2016, 23, 243–256. [CrossRef]
- 95. Spano, A. How do we measure public value? From theory to practice. In *Public Value Management, Measurement and Reporting;* Emerald Group Publishing Limited: Bingley, UK, 2014.
- 96. Benington, J. Creating the Public in Order to Create Public Value? Int. J. Public Adm. 2009, 32, 232-249. [CrossRef]
- 97. Benington, J. From Private Choice to Public Value. In *Public Value: Theory and Practice*; Palgrave Macmillan: New York, NY, USA, 2011; pp. 31–51.
- 98. Heeks, R. ICT4D 2.0: The next Phase of Applying ICT for International Development. Computer 2008, 41, 26–33. [CrossRef]
- 99. Karkin, N.; Janssen, M. Evaluating Websites from a Public Value Perspective: A Review of Turkish Local Government Websites. *Int. J. Inf. Manag.* 2014, 34, 351–363. [CrossRef]
- 100. Talbot, C.; Wiggan, J. The Public Value of the National Audit Office. Int. J. Public Sect. Manag. 2010. [CrossRef]
- Meynhardt, T.; Bartholomes, S. (De) Composing Public Value: In Search of Basic Dimensions and Common Ground. Int. Public Manag. J. 2011, 14, 284–308. [CrossRef]
- Brookes, S.; Wiggan, J. Reflecting the Public Value of Sport: A Game of Two Halves? *Public Manag. Rev.* 2009, 11, 401–420. [CrossRef]
- 103. Van Thiel, S.; van Genugten, M.; Voorn, B. The Relationship between Governments and State-Owned Enterprises. In *The Routledge Handbook of State-Owned Enterprises*; Routledge: New York, NY, USA, 2020; p. 322.
- 104. Kearney, R. Public Sector Performance: Management, Motivation, and Measurement; Routledge: New York, NY, USA, 2018.

- 105. Van Dooren, W.; Voets, J.; Winters, S. Autonomy and Reregulation: Explaining Dynamics in the Flemish Social Housing Sector. *Public Adm.* **2015**, *93*, 1068–1083. [CrossRef]
- 106. Brown, T.L.; Potoski, M. Contract–Management Capacity in Municipal and County Governments. *Public Adm. Rev.* 2003, *63*, 153–164. [CrossRef]
- Egeberg, M.; Trondal, J. Political Leadership and Bureaucratic Autonomy: Effects of Agencification. *Governance* 2009, 22, 673–688. [CrossRef]
- 108. Cambini, C.; Filippini, M.; Piacenza, M.; Vannoni, D. Corporatization and Firm Performance: Evidence from Publicly-Provided Local Utilities. *Rev. Law Econ.* 2011, 7, 191–213. [CrossRef]
- 109. Swarts, D.; Warner, M.E. Hybrid Firms and Transit Delivery: The Case of Berlin. *Ann. Public Coop. Econ.* **2014**, *85*, 127–146. [CrossRef]
- 110. Pérez-López, G.; Prior, D.; Zafra-Gómez, J.L. Rethinking New Public Management Delivery Forms and Efficiency: Long-Term Effects in Spanish Local Government. *J. Public Adm. Res. Theory* **2015**, 25, 1157–1183. [CrossRef]
- 111. Girth, A.M. A Closer Look at Contract Accountability: Exploring the Determinants of Sanctions for Unsatisfactory Contract Performance. J. Public Adm. Res. Theory 2014, 24, 317–348. [CrossRef]
- 112. Melkers, J.; Willoughby, K. Models of Performance-Measurement Use in Local Governments: Understanding Budgeting, Communication, and Lasting Effects. *Public Adm. Rev.* 2005, 65, 180–190. [CrossRef]
- 113. Benabou, R.; Tirole, J. Intrinsic and Extrinsic Motivation. Rev. Econ. Stud. 2003, 70, 489–520. [CrossRef]
- 114. Gray, R. Social, Environmental and Sustainability Reporting and Organisational Value Creation? Whose Value? Whose Creation? Account. Audit. Account. J. 2006. [CrossRef]
- 115. Dyllick, T.; Rost, Z. Towards True Product Sustainability. J. Clean. Prod. 2017, 162, 346–360. [CrossRef]
- 116. Mahadi, Z.; Sino, H. Defining Public Needs in Sustainable Development: A Case Study of Sepang, Malaysia. *PERTANIKA J. Soc. Sci. Humanit.* 2013, 21. [CrossRef]
- 117. Presidency of the Italian Council of Ministers Civic Service in Other Countries. Available online: https://www.serviziocivile.gov. it/menusx/servizio-civile-nazionale/sc-allestero/negli-altri-paesi.aspx (accessed on 27 January 2021).
- 118. Sentenza, N. 54 / A/2016 Corte dei Conti, nei giudizi di appello in materia di responsabilità amministrativa iscritti ai nn. 5329/A/Resp, 5322/A/Resp, 5344/Resp, promossi ad istanza. Available online: https://corteconti-iit.almawave.cloud/api/ portal/downloadDocument/sentenze/SEZIONE%20DI%20APPELLO%20PER%20LA%20SICILIA/SENTENZA/54/2016 (accessed on 6 April 2021).
- 119. Serofilli, M. Promuovere La Progettualità Del Volontariato. In *Riflessioni sulla Progettazione Sociale dei Centri di Servizio per il Volontariato in Emilia Romagna;* Franco Angeli: Milan, Italy, 2001.
- Aleksandrov, E.; Bourmistrov, A.; Grossi, G. Participatory Budgeting as a Form of Dialogic Accounting in Russia: Actors' Institutional Work and Reflexivity Trap. Account. Audit. Account. J. 2018, 31, 1098–1123. [CrossRef]
- 121. Secinaro, S.; Brescia, V.; Iannaci, D.; Jonathan, G.M. Does Citizen Involvement Feed on Digital Platforms? *Int. J. Public Adm.* 2021. [CrossRef]
- 122. Hervieux, C.; Voltan, A. Toward a Systems Approach to Social Impact Assessment. Soc. Enterp. J. 2019, 15, 264–286. [CrossRef]
- 123. Branch, K. Guide to Social Impact Assessment: A Framework for Assessing Social Change; Routledge: New York, NY, USA, 2019.
- 124. Gallo-Rivera, M.T.; Mancha-Navarro, T.; Garrido-Yserte, R. Application of the Counterfactual Method to Assess of the Local Economic Impact of a Nuclear Power Station. *Energy Policy* **2013**, *62*, 1481–1492. [CrossRef]
- 125. Il Margine. Storytelling Le Testimonianze VOL.TO Versione 2. 2020. Available online: https://www.youtube.com/watch?v= DJgN1LcP0ks (accessed on 10 April 2020).
- 126. Caria, M. Revenue and Expenses Civil Service Vol. To 2019 2020. Available online: http://doi.org/10.5281/zenodo.4629402 (accessed on 10 April 2020).
- 127. Brescia, V.; Caria, M.; Mapelli, E. The Assessment of the Social Impact of the Civil Service Vol. To in the Year 2019 Project Work. 2020. Available online: http://doi.org/10.5281/zenodo.4629441 (accessed on 10 April 2020).
- 128. Torino Social Impact Corso Universitario sulla Valutazione di Impatto Sociale. Available online: https://www.torinosocialimpact. it/attivita/corso-universitario-sulla-valutazione-di-impatto/ (accessed on 23 March 2021).
- 129. CEPAS ACCREDIA Chief Value Officer and Impact Assessor Process. Available online: https://www.cepas.it/settori-dicertificazione/chief-value-officer-e-valutatore-dimpatto/ (accessed on 23 March 2021).
- 130. Lever UP. Available online: http://www.leverproject.eu/ (accessed on 31 January 2021).
- 131. Manual on the Measurement of Volunteer Work; International Labour Organization: Geneva, Swizerland, 2012; ISBN 978-92-2-125070-8.





# Article The Human Capital for Value Creation and Social Impact: The Interpretation of the IR's HC Definition

Maurizio Cisi \* D and Francesca Alice Centrone

Department of Management, University of Torino, 10134 Turin, Italy; francescaalice.centrone@unito.it \* Correspondence: Maurizio.cisi@unito.it

Abstract: The paper aimed at assessing and identifying in a comparative way the potentiality and suitability of the Integrated Reporting (IR)'s definition of human capital (HC) within a selection of companies and social cooperatives. Employing in-depth interviews, the qualitative study analyzed the points of view of a selection of human resources managers to firstly check and test the connection between human capital, value creation, and social impact. The contribution of human capital to value creation is not easily recognized, especially by smaller-sized companies. The results suggested that the HC definition of the IR in the for-profit sector seems to be more applicable to the top management than to the whole workforce, while it appears as "fitting" for the managers of social cooperatives because of its explicit focus on ethical values, loyalty, and motivation. This allowed opening possible channels of dialogue between the profit and not-for-profit sectors. The paper proposed practical recommendations to operationalize the IR's HC definition.

Keywords: human capital; integrated reporting; value creation; social impact

**Citation:** Cisi, M.; Centrone, F.A. The Human Capital for Value Creation and Social Impact: The Interpretation of the IR's HC Definition. *Sustainability* **2021**, *13*, 6989. https://doi.org/10.3390/su13136989

Academic Editor: Andrea Pérez

Received: 3 May 2021 Accepted: 15 June 2021 Published: 22 June 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

#### 1. Introduction

The paper focused on human capital (HC) and value creation dynamics in a selection of Italian conventional firms and social cooperatives. The paper intended to clarify whether the Integrated Reporting (IR) definition of the HC defines a model that could be applicable in order to better manage the value creation process and indirectly influence the potential social impact. To achieve such a goal, researchers interviewed a purposive selection of HR managers from conventional firms and social cooperatives active in the Northwest of Italy.

The IR framework foresees six capitals: financial, manufactured, intellectual, human, social and relationship, and natural. Such capitals are defined as 'stock of value that are increased, decreased, or transformed through the activities and outputs of the organization. This represents one of the main key concepts of the IR and also one of the most criticized' [1,2], mainly because of the difficulties in comprehension and adoption by firms.

One of the main theoretical references followed in the research is the critical accounting approach [3,4]. The critical accounting approach [5] insists on the role of accounting as a catalyst between organizations and companies to make accountability a wider, inclusive, and democratic process focused on their social impact. The paper contributed to the debate on the limits of the conventional accounting/non-financial reporting and accountability models [6,7], especially if applied to the social entrepreneurship sector [8]. This difficulty of adaptation and application of traditional models increases exponentially in measuring, evaluating, and reporting the most intangible elements of organizations, such as HC [9]. The same standard-setters highlight how the HC is one of the most difficult capitals to be measured and disclosed.

The literature [10,11] has shown that, up to date, the implementation of the IR model is still limited; however, the IR is recognized as a tool for transparency and accountability. Research based on a sample of IR prepared by a sample of 18 European State-owned enterprises between 2013 and 2017 shows an increasing level of disclosure during the observed period [12].

Some emerging studies [13,14] are highlighting the potentials and limits [15] of the IR model. This paper can be included in this recent branch of research. In this sense, the purposive sample of Italian companies and social cooperatives can constitute an interesting and challenging field of analysis not sufficiently explored. Additionally, the study intends to practically contribute to the call launched by several authors [10,16,17] to critically stress and explore benefits and constraints related to the IR application and directly contributing to the active debate promoted by the IIRC.

The definition provided by the IIRC states that the HC is related to 'people's competencies, capabilities and experience, and their motivations to innovate, including their:

- alignment with and support for an organization's governance framework, risk management approach, and ethical values  $\rightarrow$
- ability to understand, develop and implement an organization's strategy  $\rightarrow$
- loyalties and motivations for improving processes, goods and services, including their ability to lead, manage and collaborate'. This definition is closely linked to value creation [18], goes beyond economic performance, and is an extensive and wide concept involving different scales also external to the organizations.

The dynamics between HC and value creation represent a quite challenging topic to be in-depth studied both from the academics and the practitioners' point of view. The IIRC itself highlighted the difficulties to adequately explore the connections between HC and value the creation process, in which reporting can play a crucial role for communication and measurement issues. Despite the fact that human capital is increasingly recognized as a relevant driver of corporate competitiveness and sustainability, the limits of financial measurement systems, which cannot fully evaluate intangible resources and are inadequate to deal with the difficulties inherent to managing the development of new emerging assets, are clear [19]. As stated by Beretta et al. [20], intellectual capital disclosure, in fact, is mainly with a backward-looking orientation and with a limited focus on human capital. In order to try to overcome this issue, few studies proposed frameworks for integrating tangible and intangible aspects of HC to comprehensively measure the overall level of human capital, as that by Mubarik et al. [21].

The peculiarity of the present paper is related to its effort to include the HR manager's point of view in the whole analysis, jointly with a more financial-oriented perspective. The same IIRC highlighted that the HR managers were one of the least involved categories in the IR redaction phases. Starting from such challenging aspects, the paper focused on the analysis of the human capital and value creation process within two different types of organizations: conventional firms and social cooperatives. Conventional firms represent a quite well explored scope of analysis for research around IR issues, on the other side, the social cooperatives have been less analyzed. It has been noted that the debate in the social sector around the measurement of social innovation outcomes and impacts mainly reflects the perspective of private social finance that increases the demand for consistent measures to account for the intended social impacts of capital allocation. In response to these demands, a number of organizations have developed methodologies for measuring the success and impact of innovation, attempting to incorporate social and environmental 'bottom lines' [22]. This area of research constitutes an interesting and challenging perspective for a series of reasons. First, the IR appears to be an appropriate framework for combining both qualitative and quantitative elements and is adaptable to structures with entrepreneurial characteristics used to manage multiple capitals and seems to be compatible with multiple objectives and stakeholders' claims as social cooperatives [23,24]. Second, the IR approach appears consistent with nature strongly oriented on mission and commitment typical of some types of social cooperatives focused on disadvantaged people, quite low trained/educated and normally kept outside from the "traditional" job market. Third, the social cooperatives, despite being "quite" formally ignorant of the IR model, are already applying such an approach in practice more than conventional firms with a less holistic vision.

Starting from the assessment of the HC definition provided by the IIRC, the paper analyzed the links between HC and Intellectual Capital (IC), compared HC and value creation dynamics between conventional and social entrepreneurship, and built some possible bridges and exchange of best practices between them. In particular, the paper addressed the following research questions:

- RQ1: Is the IR's HC definition effective and applicable to both conventional companies (including SMEs) and social cooperatives?
- RQ2: Are the components of the IR's HC definition useful to articulate a clear picture of the value generated by the HC management of conventional companies (including SMEs) and social cooperatives?
- RQ3: Could the IR be a possible and useful driving force to improve the HC valorization within both conventional companies (including SMEs) and social cooperatives towards the goal of improving social impact?

The paper is organized as follows: first, we introduce the theoretical framework; then we detail the methodology applied in the research; afterward, we explore and discuss the main findings; finally, we expose the final conclusions of the study.

#### 2. Materials and Methods

The search for greater accountability, the need for valuable and context-oriented HC information (not merely quantitative), and the establishment of proper and specific non-financial reporting models constitute some key issues for the whole social economy sector [25,26]. Previous research as that by Tejedo-Romero and Araujo (2021) showed that companies are adapting to new regulations and voluntarily disclosing information on human capital, a trend that signals their commitment to responsible attitudes towards employees and stakeholders [27]. For such reasons, the paper can be inserted within the stakeholder theory framework, which analyses the value creation at the firm level and considers the implications of stakeholders' expectations on the information disclosure process 6/16/21 7:46:00 PMp.m. At the same time, the behavioral decision-making theory, more focused on the individual level and on the firm attributes and stakeholder perceptions, also represents a valuable theoretical framework for the analysis [28] mainly because of its greater attention to the increased IR quality and not only on its implementation process.

While the greater part of the academic literature on IR is mostly focused on the studies of the Intellectual Capital (IC) [16,29–34], looking at the HC as a subcomponent of the IC together with the relational and organizational capitals, only a few works directly addressed HC in a stand-alone perspective [35,36].

In the last decades, many definitions and interpretations of HC have been proposed related to the human resources field and around the IC perspective [37–39]. Despite HC and HR (human resources) could be often interchangeably used, HC includes a whole range of factors such as people's knowledge, competencies, experiences, and the full contribution of these to the firm's performance [40]. According to a managerial point of view, HC is composed by staff know-how, staff formal qualifications or aptitude, union relationships with management, and staff relations [41]. Human capital is captured of 'know-how, education, vocational qualification, work-related knowledge, work-related competencies, entrepreneurial spirit' [20].

At a general level, findings from a cross-country analysis emphasized the role of human capital in enhancing the awareness of and compliance with environmental regulations, thus leading to an improvement in environmental performance [42].

However, the measurement and the impact of the specific human capital is a challenging activity. An example is that of Ortega-Lapiedra et al. [19] that defines a human capital specific index (HCSI) to analyze the influence of firms' human capital in their activities devoted to environmental improvements through innovations to produce new ecological products or to introduce eco-innovative processes based on more sustainable resources and the reduction of waste. The study evidenced empirically that human capital is responsible for creating a sustainable competitive advantage in the eco-innovative activities of the business.

Another study has evidenced that both the human capital development and employees' commitment were found to have a partial mediation in the strategic HR management practices and sustainable competitive advantage [43].

These intangible resources are embedded in the context in which they work and can be presented "in action", showing their contribution to value creation [44]. In order to cover the lack of information in regard to the long-term value creation process [45], the IR framework-based disclosures are characterized by a greater variety of items and more emphasis on human capital [46]. Moreover, as stated by Passetti and Cinquini [47], developing an integrated approach to reporting could provide a more complete overview of the companies' characteristics and performance, with the aim to reduce information asymmetry.

Despite its several criticisms and challenges [15,18,48,49], some emerging research has shown that the IR could be a suitable model to be applied also to hybrid and not-for-profit organizations [12].

The general definition given is that hybrid organizations are organizations created in order to address public needs and to produce services that are public in character, at the same time resembling private corporations in the way they are organized and managed [50].

It has been shown that the combination of public and private characteristics in one organization generates ambiguity that manifests itself in the strategic objectives as well as in the structure of the organization. As stated by Cornforth [51], organizational governance is important in helping to manage competing pressures in order not to degenerate into being no different from capitalist forms of businesses. In fact, one of the challenges facing hybrid organizations, such as social enterprises, is how to combine different logics: commercial logic necessary for operating in the market with a 'charitable' logic of pursuing a social mission [52,53].

The ambiguity imposes specific abilities of the management especially, in driving the human capital [50]: research suggested that organizational membership decisions and socialization policies contribute to imbuing members with organizational values and associated working practices so that they may act based on moral reasoning and commitment to organizational goals [52]. It was found that human capital affects social capital [54] and that experience and cognitive ability influence personal relations and complicity. In other terms, the organizational performance also in terms of social impact is strongly influenced by human capital.

One of the starting points of the research lay in the assumption of Gamerschlag and Moeller around the positive cause-effect link between HC reporting and disclosure and the skills [55], motivations, and commitment of the workforce. Producing human "capital" disclosure on the one side allows companies to determine and appreciate the contribution that human resources provide to the overall economy of the company and on the other, to understand how to improve it, through both a qualitative and quantitative assessment done ex ante or ex post as well [56].

The model was based on a relationship between a company's human capital and its financial performance, taking three factors into consideration (see Figure 1):

- workforce-related factors (human capital), such as the workforce's capabilities, its motivation, and commitment;
- (2) company-internal factors (structural capital) include a company's operational performance, its innovation ability, as well as its corporate culture;
- (3) company-external factors (relational capital), parameters outside the company that are relevant to a company's success.



Figure 1. The cause–effect link model.

Our analysis responded to the call launched by Gamerschlag and Moeller to empirically test the veracity of their assumption, focusing in particular on the first factor (the workforce-related or human capital) [55], since it appeared to be the most coherent both with the HC definition of the IR and the core of the paper as well, through semi-structured qualitative interviews addressed to the human resources managers (HR), of ten selected Italian companies and social cooperatives.

The logical process applied in the research was as follows. Starting from the HC definition of the IR, the focus was on three joint levels/points of views of value creation processes: (1) the value creation for/within the HC concept; (2) the value creation for/within the company/cooperative; (3) the value creation towards the external environment (i.e., the social impact).

In order to explore such value creation processes, the following key issues/drivers were checked: (1) how the HC definition of the IR was adapted to the "work scheme" of the selected organizations; (2) how the single HC definition components effectively contributed to the value creation; (3) whether the social cooperatives expressed a greater level of acceptance/sharing of the HC definition; (4) how such potential greater level of acceptance might imply a better HC management and, thus, a greater internal and external value creation.

The main research statement lay around the hypothesis that whether an organization (mainly a hybrid form as the social cooperative) effectively interpreted and applied all components of the HC definition of the IR, might consequently improve its performance and management of HC and, thus, enhance its social impact also outside.

The research approach was qualitative and interventionist [57]. The data collection was based on multiple sources, both primary and secondary data. The primary data were collected from semi-structured interviews addressed to a purposive sample of HR managers coming from companies and social cooperatives active in the Northwest of Italy, while the secondary data (when available) were collected through the content analysis of the NFR (included the companies-integrated reports when available), of their website's sections related to sustainability and CRS issues, and through the reading of other documents (as the social reporting of the cooperatives when available). The combined use of primary data and secondary data made the research design more convincing and consistent [58] allowing to validate the information collected and to identify further insights [59].

The semi-structured interview method (see Appendix A) allowed a certain level of reciprocity between interviewer and participant, enabling to improvise follow-up questions based on the participants' responses [60]. The choice of in-depth and semi-structured

interviews was in line with the localist perspective in accounting [61] to make it more inclusive and increase its role and social effects. Furthermore, semi-structured interviews allowed understanding 'the way interviewees perceive the social world under study' [57].

The interviews were carried out between February and November 2020. The interviews were addressed to 5 HR managers from 5 conventional firms (5 senior) and 4 HRs (3 senior and 1 junior) from 3 social cooperatives. The characteristics of the respondents are disclosed in Table 1. Regarding the saturation issues, 9 interviewees have been considered enough by Francis et al. [62] if considered as an initial analysis sample. Additionally, samples in qualitative research tend to be small in order to support the depth of the case-oriented analysis specific to this methodology [63]. Additionally, purposive qualitative samples provide richly-textured information on the investigated phenomenon [64].

All interviews were recorded, transcribed, and analyzed using qualitative content and discourse analysis techniques [65].

The "twin approach" towards sustainability reporting identified by Stubbs et al. [66] shaped the necessity to include the HR perspectives. According to this author, involving both internal and external stakeholders allows achieving an actual Integrated Reporting, with many organizational units active in collecting information, communicating, and reporting as stakeholders contributing to the value creation process [66], improving the whole quality, scope, and usability of the reporting. Accordingly, purposive sampling was used to select key informants (HR managers) with acknowledged expertise by their experiences in leading HC [67].

The 5 HR managers of the conventional firms were selected through the geographical proximity criteria, the very different average number of employees, and different sectors, while the 4 HR of the social cooperatives work for some of the biggest social cooperatives from the Northwest of Italy active in the field-of-work integration of disadvantaged workers.

The comparative lens between conventional firms and social cooperative, despite already applied by other branches of the economy (see, for instance, the works on performance and productivity of Thompson [68] or that on wage differentials of Bailly et al. [69]), represents in our knowledge a quite emerging and challenging approach in the literature on sustainability and integrated reporting and human capital issues [70,71].

The choice of very different types of respondents (see Table 1) has been influenced by the will to achieve a pluralist and broad point of view (in line with the dialogic polylogic accounting approach—see Dillard and Roslender [72]) and to study the perspectives of organizations with different levels of added value in terms of human capital intensity. Additionally, a peculiar aspect regarding the social cooperatives respondents was that both the organizations expressly asked to involve two different persons (1 senior and 1 junior from the same department) in the research interviews. This request allowed a different and double point of view on the research topics and also to partially deal with the limited number of involved social cooperatives.

An additional important asset of the selected companies regarded the involvement in the study of 2 companies whose core business was respectively HC management (R3) and IC management (R4). This specialization allowed one to achieve a privileged point of view, mainly in the section of the analysis focused on the connection between HC and IC.

Strictly connected with the willingness to provide a multiple and pluralistic point of view was also the fact that the selected companies provided different types of non-financial reporting. The small and medium conventional firms did not provide any report, 2 social cooperatives provided a social report (which is mandatory for them according to the Italian law), the biggest company of the study was the only which prepared an Integrated Reporting.

ID RESPON- DENT	GENDER (M/F)	SENIORITY (Years in the Company)	INDUSTRY/SECTOR	CONVENTIONA FIRM OR SOCIAL COOPERATIVE	I N. OF EMPLOYEES	TYPE OF NFR REPORTING PROVIDED (If Any)
R1	М	8	Pharmaceutical	Conventional firm	560	n.a.
R2	F		Taps and fitting	Conventional firm	87	n.a.
R3	F		Human resources	Conventional firm	2293	IR
R4	М		Intellectual property consultancy	Conventional firm	100	n.a.
R5	М		Textile	Conventional firm	13	n.a.
R6	F	30	Welfare, socio-health and educational services	Social cooperative	743	SOCIAL REPORT
R7	F	34	Cleaning and Environmental Services	Social cooperative	459	SOCIAL REPORT
R8	F	5	Cleaning and Environmental Services	Social cooperative	459	SOCIAL REPORT
R9	М	22	Circular economy/waste collection and disposal	Social cooperative	250	n.a.

#### Table 1. Overview of the main respondents' characteristics.

#### 3. Results

The findings from the interviews were divided into the following sections:

- 1. Knowledge and use of the human capital concept,
- 2. HC and social value creation,
- 3. Value creation through HC between conventional firms and social cooperatives.

#### 3.1. Knowledge and Use of the Human Capital Concept

Concerning the level of sharing and operability with respect to the IR's definition of HC, respondents were asked to respond by using a 5-point Likert scale (1 = completely disagree, 5 = completely agree). Almost all respondents answered 3 or 4. Thus, the level of sharing to the contents of the HC definition of the IIRC appears quite high, even if at the same time such definition has still been criticized for its excessive technicality and the too theoretical and less operational approach. At the same time, the representatives of the social cooperatives showed a double attitude about the efficacy of applying such HC definition to the basis of the cooperative, to the less-skilled workers/working members. On the one hand, the HC definition can be more easily applied to the top management level ('firstly, the motivations to innovate and the ability to understand, develop and implement an organization's strategy', R9), whereas some of the HC definition components ('as the alignment with and support for the organization's governance and the loyalties and motivations towards the final results', R8) could be more efficiently expressed by "simple" working members. Indeed, for them, the support and the loyalty towards the social cooperative itself was very important, representing an actual 'way to take back their destiny' (R9).

Another point highlighted by some respondents was the possible lack of the HC definition provided by the IR model. Among such identified lacks, one of the most significant ones was the 'self-awareness of people, of workers to be such, to be part of something very specific' (R7).

The empirical study was aimed at verifying whether and how widespread the use of the term HC was within the selected companies since previous research had also highlighted a general preference towards other terms such as people, employees, staff [73]. It could be observed that a slight majority of companies did not use this term in their corporate communication, while most of them usually employ it within their own organization. The company size does not seem to be a determining factor regarding the use of the term HC within companies. One of the most senior respondents, HR of social cooperatives, clearly stated: 'I do not like the use of the word human capital, because (it is) too much connected to the financial capital concept' (R8), while the younger professionals appeared quite open towards the importance of considering the human and financial capital at the same level within the organization.

It was much more interesting to analyze the personal definitions of HC provided by the HR of the companies. Such definitions could be divided into two main categories:

- on the one hand, those who see the HC as a physical resource of the company, a "fuel that makes it go", from a perspective mainly oriented to the interest and the point of view of the company;
- (2) then, some look primarily at the more "intangible" components of the HC, those related to the skills inherent or developed by employees primarily on a personal and individual level, and not only and solely in the workplace or the formative framework.

This interpretation tends to conceive a closer connection between the company and its staff that can allow the opening of a quite unusual perspective on the delicate dynamics of human capital and the creation of value. On the one hand, the person who leaves the company represents a loss for both parties, from the point of view of the resource, the commitment, and the skills acquired or provided but also about the social and relational capital developed in the company. On the other hand, people (with wrong or right skills and characteristics) placed in the wrong positions can constitute an inefficient investment both in terms of performance and commitment to the company mission as well.

An aspect emerging from the interviews was a sort of division made by some respondents—mainly from the conventional firms—between the 'more skilled and R&D employees for which the academic titles count more, and the human capital represented by the less skilled and more operational job positions for which motivation is a greater asset to be considered' (R1). If the technical competencies of people composing the IC are normally clearly recognized by HR, a controversial issue regards the difficulty of the IC "elite" to relate with the HC people entailing quite rare occasions of sharing and cooperation.

The analysis about the level of operationality and possible measurement of the IR's HC definition was performed, streamlining and facilitating the responses through the aggregation of the HC indicators into different categories as follows:

- (1) employee involvement, staff satisfaction rate, motivation, commitment;
- (2) corruption, business ethics, respect for human rights;
- (3) diversity of staff, gender equality;
- (4) experience, training, education;
- (5) average age/professional seniority/qualifications, education;
- (6) productivity;
- (7) turnover, staff loyalty/commitment rate.

Then the respondents were asked to order such indicators according to a Likert scale (1 most significant; 7 less significant).

The HR managers confirmed the difficulty of measuring and evaluating the HC, which is usually applied in a limited way by research companies and, in any case, requires large investments of time and resources generally not accessible by smaller companies.

It emerged that the HRs included multiple indicators that hold together employee involvement/staff satisfaction rate/motivation/commitment in the first place. The parameters of average age/professional seniority/qualifications/education were placed in the second position. On the other hand, in third place, there were some fairly different variables: productivity, staff turnover/retention rate, and corruption/business ethics/respect

for human rights. Finally, in the "last" position, there was the parameter diversity of the staff/gender equality, a theme that, although it is currently the page for academics and professionals (think also of the growing attention to the issue of diversity and inclusion as an engine for the enhancement of personnel and contribution to the creation of value [74]), from the answers collected it appeared relatively marginal compared to other more "conventional" indicators and macro-themes.

#### 3.2. HC and Social Value Creation

In the light of the interviewees' insights, we tried to understand the usefulness of the components of the IR's HC definition to articulate a clear picture of the value generated by the HC management. We Considered, in fact, that the IR's definition of HC is based not only on people's competencies, capabilities, and experience and on motivations to innovate, but it also relies on specific components (such as alignment with and support for an organization's governance framework; risk management approach; ethical values; ability to understand, develop, and implement an organization's strategy; loyalties and motivations for improving processes, goods, and services; ability to lead, manage, and collaborate) that can be directly or indirectly linked with the company's social impact or social value created.

In this second perspective, the HC definition of the IIRC expresses a concept of human capital that is closely linked to the creation of value, and that goes beyond the economic/performance aspects, implying a more extensive concept involving wider areas and scales (as the social, environmental, cultural, etc.) also outside the company [75,76].

Therefore, the research hypothesized that the creation of value (as that described by the IR model) outside the selected organizations might in some way give rise to effects similar to those of the social impact.

The social impact of a company or organization can include social and environmental changes resulting from activities and investments. The impacts on society include issues related to the issues of equality, livelihoods, health, nutrition, poverty, security, and justice [77]. Investments that create social impacts can take various forms and consist of time, skills, material resources, network connections, reputation, and other valuable resources.

Based on the feedback of the HR's responses, we could identify two main dynamics/attitudes.

Firstly, on the one hand, the realities of SMEs explicitly declared that they do not contribute to either of these two effects/phenomena. If this answer could be partly explained by perhaps a limited knowledge and/or mastery of these concepts, on the other hand, an interesting aspect emerged. That is the fact that, although these companies denied contributing in a positive way to the creation of value and/or social impact, in reality, they did so in practice. This contribution could pass primarily from the economic and social benefits that come directly to the families of their employees, to have shown attention to the times of corporate, social, and environmental sustainability since before it became a sort of moral obligation for all business realities, but also to the fact of using a network of expressly local suppliers and sub-suppliers, thus, helping to keep the reference production district active both in terms of employment and relationships of trust.

Secondly, on the other hand, the larger companies (with more than 100 employees) showed that they know these two concepts well and recognize the contribution of their companies without doubts/hesitation. This different attitude could certainly be attributable to greater marketing and communication skills but also certainly to attentive and well-organized corporate welfare policies that see their contribution in terms of social impact as a mix between the transmission/acquisition of skills, attitudes and positive values, stable employability levels over time, and a combination of different corporate welfare tools. Namely, positive policies that place the 'social impact generated as the starting point and ultimate goal of these business realities, although sometimes it can mean making more efforts and having slower reactions in terms of performance/reaction on the market' (R5).

A controversial and unexpected point that emerged during the research is that of how we cannot always talk about HC and the "incremental" creation of value, but how sometimes the opposite can happen: that is, a decrease in value '*in terms of loss of so-cial/ethical/skills* ... *impoverishment of the HC due to the company* ... *there may be business contexts in which this occurs*' (R3). What emerges is that this negative dynamic/vision, which associates a strong moral/social connotation with the concept of value, that is the cause of this impoverishment and should be sought more from the managerial point of view than from that one of the people who are part of it.

#### 3.3. Value Creation through HC between Conventional Firms and Social Cooperatives

Partially connected with the topic of the loss of human capital was the specific point of view expressed by the implicit nature of the social cooperatives: '*if you count 1 when you are part of the cooperative when you live it you don't count 0.5 or other, you count 0*' (R6).

To compare with the for-profit organizations, the interviewed social cooperatives more easily recognized the social impact of their organization. An exhaustive example was offered by R9, who described a quite paradoxical situation where 'people entering in the buildings to collect waste paper were maybe those who before committed crimes inside the same buildings [ ... ]. The same happened with some supermarkets where our former laborers were then employed'. This is to say that quite often, the people/human resources appeared as the first direct witnesses and proof of the social impact generated by the social cooperatives, first, in terms of self-determination, employment, and welfare opportunities. Such employees also expressed a large and shared corporate spirit made of positive and virtuous behaviors to avoid negative effects on the whole organization to which they belong.

Obviously, it is not possible to make a significant comparison between the social impact generated by a social cooperative and a more conventional firm because of their different nature, vision, and mission. However, such findings allow to undermine a series of common key points between these two types of organizations:

- a quite hidden acknowledgment and awareness of the generated social impact;
- a quite general lack of shared and effective quantitative metrics to measure and assess the related social impact;
- a general impression that the external social impact of the organization may, in some cases, appear more easily catchable/caught than the internal social impact.

Some respondents expressed several doubts regarding the efficacy of the social impact concept, which appeared as a quite vague and trend model, rich in quantitative metrics (e.g., increased number of beneficiaries; the scope of the carried-out activities; etc.), not well adapted to be used to assess the actual management of human people and to represent the value creation process inside and outside the organizations.

#### 4. Discussion

The first aspect that clearly emerged from the research is the limited mutual knowledge between these two sectors, which, therefore, entails visions strongly conditioned by the backgrounds or personal/social networks of the interviewees. In some cases, the opposite vision between profit and not-for-profit emerges more: '*I can say that we are the polar opposite of the not-for-profit, we belong to an investment fund and we have to soak the return in coffee and milk every morning*' (R1).

One of the priority aspects that emerged from the analysis is the vision of a not-forprofit/social enterprise sector more attentive to the ethical dimension of work, in which people are valued more for their level of sharing the mission and values of the organization rather than their skills and performance. About the skill topics, indeed an interesting aspect to be co-opted by the conventional firms should be the skills enhancement process, sometimes "sacrificed" by the social cooperative because 'once you enter there, you adapt yourself, [...] we don't make treasure of those who are innovative' (R8). Another aspect that could be by the for-profit sector is a more organized and structured process of staff selection. Social cooperatives indeed would tend to privilege the empathic concept than the performance or the attitude of people. Conversely, an interesting insight that social cooperatives could bring is the ability and experience related with fragile and marginalized people, those, which normally would be excluded by the conventional labor market.

Closely linked to the theme of the performance was how, among the points in the profit sector that could constitute a virtuous example to export/apply to the non-profit sector, there was a better company organization, respect for deadlines, and deadlines. While, on the one hand, this interpretation could be partially confirmed by the literature on the subject, which describes the difficulties of the not-for-profit sector to approach the issue of the corporate organization more efficiently, on the other, it reflected a vision of the non-profit sector as a sector almost aside; it must not work willy-nilly within the market dynamics of large companies.

Secondly, another relevant aspect of the interviews was the importance of the company size. That is, if at a qualitative level the social impact of the non-profit sector is often seen as greater, at a quantitative level, according to some respondents, larger profit companies would seem to be able to count on a greater reach in terms of social impact. From certain points of view, this vision could find partial confirmation in the past industrial reality of the Italian north-west (see for example how some large historic companies such as FIAT, Olivetti, etc. have in the past largely influenced the human, social, and entire cities through an intense loyalty-building action of its employees, through ad hoc services, and benefits), on the other hand, two possible objections could be outlined. First of all, it was necessary to be careful not to confuse the social impact with the positive effects linked to economies of scale and production districts; furthermore, it could be observed how these large industrial realities have radically changed over time and currently have a territorial presence and an economic (and social) impact, not comparable to the past. The attention that more and more SMEs and social enterprises are showing towards the need to be accountable to the more social and environmental aspects of their activities, despite not representing a legal obligation for them, made us understand how much human capital and value creation, even outside companies, is becoming an increasingly transversal aspect, regardless of company size. Another contact point between the SME and the social cooperative was the existence of less hierarchical relations to compare with the large conventional firms, for instance, 'the HRs know all (employees) by name and surname, ... we are as a big family' (R8).

Despite their different approaches towards human capital and value creation dynamics, both conventional firms and social cooperatives expressed similar points of view about a series of common topics (Figure 2).

In another perspective, the results showed HC and IC as two different but complementary concepts (in line with the IR vision of capitals) and both difficult to be measured and assessed. Strongly connected to the link between HC and IC was also the common topic of the intergenerationality, namely the challenge of balancing innovation and creativity of the younger with the expertise and the usual and "traditional" organizational way of the most senior human resources. In this sense, it may appear significant that both the social cooperatives included in the study decided to involve also some younger colleagues in the interviews, in a general view of skills and duties handover. In Addition, the importance of transferring the knowledge to the new generation is a core value of social enterprises in general.

In this context, the contribution to value creation (whose definition largely varies among the different respondents) and social impact was not easily recognized, especially by the smaller-sized conventional firms. It was quite largely recognized as the mission of the organizations to provide 'thought the job, opportunities of growth, and development of capabilities within a company that partially can be yours' (R6). Conversely, the social cooperatives appeared more conscious of the internal and external value created by its activities because 'the social cooperative must respond to the needs of the local community in terms of fragility and social disease, but also the need for job opportunities' (R6).

OUTCOMES/INSIGHTS	MOTIVATIONS/IMPLICATIONS	CONVENTIONAL	SOCIAL
		FIRMS	COOPERATIVES
Difficulty to measure, assess, and communicate the HC implications, effects		х	xx
Limited knowledge of the IR Framework	Not surprising considering that most of the selected organisations are not requested by law to provide any type of NFR	xx	XXX
Unclarity and unknowledge of the social impact concept	Due to different reasons i.e. because the organisation is not completely conscious of its social impact; because it appears a quite "fashionable" model not applicable in the realty, etc.	XXX	x
Presence of reciprocal bias and unknowledge/misunderstanding each other	May impede an effective exchange of possible best practices around HC and value creation	xx	xx
General good sharing level towards the HC definition provided by the IR model and quite shared vision on the need to include also the emotional component of people/employees in such definition		xx	XXX
Positive social effect for all categories of employees (first the most disadvantaged and the less skilled) generated by the actions and by the organisational model of the selected organisations	Despite such effect is not well reported though specific metrics or standards and despite the several criticisms shown by the HRs on a well-recognised and shared model (as that of the social impact)	XX	XXX
Close link between HC commitment and value creation dynamics		x	XXX

**Figure 2.** Similarities between conventional firms and social cooperatives. Legend: x = not very evident; xx = evident; xxx = very evident.

Regarding the truthfulness of the paper's initial assumptions, which state a similar effect between the HC and value creation process and the social impact, at this stage of the research, partial veracity could be observed. Indeed, for the social cooperatives, this assumption appears more easily confirmed since they highlighted a mission strictly connected to dealing with people's diseases, increasing at the same time their wellness; for the conventional firms, such a link diverged according to the type of sector, organization, etc. In this sense, the most conscious of their social impacts ('we generate value because people live experiences together, exchange, grow together, confront each other', R3) were the companies that, by nature, employ the most qualified people (i.e., as the intellectual capital property consultancy or the human resources management company). Conversely, the most "traditional" and manufactured companies showed more limited awareness and knowledge about the concept and the scope of their social impact ('we cannot speak of a contribution to the creation of value, [here] there is too much gratuitous malice ... everyone looks only at their own reality', R2). A separated discussion should be done with the unique family and SME company (see R5), where the limited number of employees could be translated into an in-depth knowledge and attention both to people's needs, characteristics, and to the local value chain and the environmental and socio-economic milieu as well.

Both the categories (conventional firms and social cooperatives) showed quite diffused HC biases, firstly related to the different levels of commitment and engagement and the

organizational management and performance. The IR appeared a quite suitable option of NFR also for the social cooperatives, firstly because of its "flexibility". The main challenge relied on how to equilibrate value creation and social effects granting more comprehensive accountability, including both private and social entrepreneurship needs and perspectives. One possible solution could be to overcome the traditional distinction between social and conventional enterprises, focusing mainly on 'good enterprises which means good and accurate management, leaving to the sons a healthy enterprise, valuing its human capital and future generation as well' (R6).

Concerning the initial research questions on the basis of the main findings, we can say that the IR definition of HC can be applicable to both the organizations (RQ1), despite many respondents highlighting its better application to the management level and less to the basis of the workforce, while about RQ2 we can state that the social cooperatives appeared more conscious of the social value creation process generated through their HC management. The same happened in relation to the bigger conventional firms, which presented a more confident attitude, knowledge, and awareness around such topics to compare with the SMEs.

Findings showed that the issue of integrated thinking is still a controversial and quite complex topic to be clearly and effectively explored (in line with Dumay and Day [16]). In some ways, the social cooperatives stressed more the strict connection between financial performance and attention to the environment and the social aspects of their organizations, first, the employee's welfare and commitment. The same happened in relation to their quite strong attention to the relationship with the main stakeholders, suppliers of the territory. Conversely, the bigger conventional firms are more used to operate in an integrated way because of their complex and interrelated structure.

Finally, it could be observed that the IR model (and its definition of HC) has undoubtedly some potentialities for both types of organizations (RQ3), firstly, because of its flexibility 'that in practice seems designed to fit all organizations into the same framework' [78]. Nonetheless, the IR model still presents some limits of application and diffusion.

#### 5. Conclusions

The qualitative study analyzed the points of view of a selection of human resources managers to test the connection between human capital, value creation, and social impact.

It seemed that the IR definition of HC is able to define a link between human capital, value creation, and social impact. It is not only based on people's competencies, capabilities, and experience. It is focused on motivation to innovate and on an explicit focus on ethical values, loyalty, and motivation. Even if the contribution of human capital to value creation is not easily recognized, especially by smaller-sized companies, the results suggested that the HC definition of the IR in the for-profit sector seems to be more applicable to the top management than to the whole workforce, while it appears as "fitting" for the managers of social cooperatives.

The research contributed to the theoretical debate on value creation [79,80] and to the more technical discussion on the difficulty of valuing the intangible aspects of organizations [81]. The value creation, its social effects, and its link with HC and IC [82] allow opening possible channels of dialogue between profit and not-for-profit organizations. In this sense, the research responded to the call launched by Girella and Dameri to apply the IR model to the third sector, social enterprises [83]. The research produced practical recommendations to make the IR's HC definition more operational and understandable and has made it possible to outline some steps that can be used to set up a sequential and circular model based on:

- recognizing that HC is an asset for the company (i.e., all respondents speak about HC but no one looks at it as capital, preferring the use of other concepts such as value, skill, etc.);
- (2) evaluating the stock of HC based on personal skills, education-training-attitudes (a phase that normally occurs during the staff selection and hiring);

- (3) acting in order to implement policies aimed at HC stock growth in order to improve the company's performance (in practice, it seems that some larger companies make this operational effort more evident/concrete than others);
- (4) assessing the results/impacts of the company actions and policies in terms of increasing the individual sub-components of the HC definition. This is a phase very rarely carried out by organizations, primarily for reasons of excessive complexity and difficulty in measuring intangibles or partly also due to a mismatch between costs and benefits. Further investigations are needed to provide a more complete picture of such critical points.

Considering that most of the analyses have mainly considered the IR preparers' point of view [84], this analysis proposed the inclusion of the non-experts' opinions (like HR managers) around the IR. The positive effects of involving external staff (not normally included in the sustainability and non-financial reporting tasks) to enhance the institutionalization of IR and the integrated thinking in practice has been stressed by scholars (as Guthrie et al. [81]; Dumay et al. [75]), who highlighted the limited research related to the IR practice (instead mostly focused on the normative level) and the lack of the practitioners' contribution to the IR debate. The paper intended to contribute to filling such gaps.

An interesting insight offered by the research was the attempt to propose some categories of KPIs for measuring/evaluating the link between HC and value creation. The orientation of most of the HR managers towards the importance of commitment and the level of involvement of their HC was shown to be consistent with the definition of HC of the IIRC, which revolves around the key concepts of sharing and support of the governance, loyalty, and commitment. The results of the semi-structured interviews showed that the definition of HC provided by the IIRC represents a good level of completeness and recognition by the HC professionals. However, from an improvement perspective, such a definition should also include the most personal and emotional components of people and be integrated to become less abstract and complex.

The main final aim was indeed to make the non-financial reporting process more transparent, shared, and accessible (and consistent with a logic of dialogic polylogic accounting, see De Villiers et al. [18]; Brown and Dillard [82]), and get it out of the usual circuits of experts and large companies accustomed to dealing with the issue. On the other hand, the attention that SMEs are recently showing towards the choice of the IR model justifies the inclusion of SMEs within the research reference target [13,14].

A limitation of the research was represented by the non-inclusion in the respondents of representatives of SMEs, a category of companies which, despite being increasingly attentive to the issue of enhancing the HC and narrating its sustainability, is less used to dealing with the metrics of non-financial reporting.

In this context, an interesting starting point for future investigations concerns the inclusion of specific parameters on HC health issues within the KPIs proposed for measuring/evaluating the link between value creation and human capital. An innovative contribution, which could enhance the current IR definition of HC, mostly focused on the "traditional" aspects of human capital (i.e., skills, sharing of values, experience, etc.).

The next step of the research could be to analyze how much the management and protection of people during a period of global emergency can make a difference, both to make companies more performing but also more accountable and appreciated by customers, investors, and lenders. In this sense, it seems necessary that the internal debate within the IIRC also considers the need to include a reference to the healthier aspects and the protection of workers' health, along the lines of what has already been done by the 403 (Occupational Health and Safety) standard of the Global Reporting Initiative (GRI).

The research conducted lay the foundations for investigating, in more detail, the role that marginal actors can play in the debate on the degree of accountability and affordability of Integrated Reporting in companies such as SMEs and social enterprises, both in terms of the definition of contents and metrics that are closer to reality and more inclusive of their potential as possible users of this non-financial reporting tool. **Author Contributions:** Conceptualization, M.C. and F.A.C.; methodology, F.A.C.; software, F.A.C.; validation, M.C. and F.A.C.; formal analysis, M.C.; investigation, F.A.C.; resources, M.C. and F.A.C.; data curation, F.A.C.; writing—original draft preparation, M.C. and F.A.C.; writing—review and editing, M.C.; visualization, M.C. and F.A.C.; supervision, M.C.; project administration, M.C.; funding acquisition, M.C. and F.A.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

### Appendix A

#### TRACK FOR INTERVIEWS—ENTERPRISES AND SOCIAL COOPERATIVES

- a. BRIEF INTRO WITH OBJECTIVES AND PRACTICAL IMPLICATIONS OF THE RESEARCH: Focus on the non-monetary value of "human capital" in order to reach a common/shared model but also adapted to the needs of social cooperatives to improve personnel management, decrease turnover, enhance the role of conventional enterprises and social cooperatives in the creation of value for society, and social impact through the enhancement of the HC.
- b. PRELIMINARY INFO
  - Company/cooperative name
  - Typology
  - Position held
  - Years of seniority in the company
  - Number of employees
- c. HC AND INTEGRATED REPORT
  - 1. Do you use the term "human capital" (HC) in your organization?
    - yes (1)
    - no (2)
    - in part (3)
  - 2. If you don't use it, why?
  - 3. Alternatively, which term do you prefer to use?
  - 4. Based on your experience, could you give us a personal definition of "human capital" (HC)?
  - 5. For your work, have you ever heard of the Global Reporting Initiative (GRI) or Integrated Reporting (IR)?
    - yes (1)
    - no (2)
    - in part (3)
  - 6. The IR defines HC as "People's competencies, capabilities and experience, and their motivations to innovate, including their:
    - Alignment with and support for an organization's governance framework, risk management approach, and ethical values
    - Ability to understand, develop and implement an organization's strategy
    - Loyalties and motivations for improving processes, goods and services, including their ability to lead, manage and collaborate ".

What do you think of this definition? Do you find it suitable for your organization?

- 7. Based on this definition of HC, in your opinion what are the best indicators to measure/evaluate the impact of HC on the creation of value for your organization (in order: 1 = most important, 7 = least important)?
  - \_\_\_\_\_ employee involvement/staff satisfaction rate/motivation/commitment
  - \_\_\_\_\_ corruption/business ethics/respect for human rights
  - \_\_\_\_\_ staff diversity/gender equality
  - \_\_\_\_\_ average age/seniority/qualifications/education
  - \_\_\_\_\_ productivity
    - \_\_\_\_\_ experience/training/education
    - \_\_\_\_\_ staff turnover/retention rate
- 8. In your company/cooperative, the social report or another form of a non-financial report:
  - is required
  - you prepare it even if it is not mandatory
  - if you do, which standard do you use?
- 9. In your opinion, why is it (or not) important to prepare a non-financial report?
- 10. If you prepare it, what is the most difficult aspect to measure/value/tell when you talk about your HC?
  - Measure/evaluate
  - Valuing/telling
- d. HC AND SOCIAL IMPACT
  - 11. If you refer to the "social capital" of your organization, what comes to your mind?
  - 12. How does the human capital of your company/cooperative influence its social impact?
  - 13. How could the connection between HC and the social impact of your company/cooperative be measured?
- e. HC BETWEEN NO PROFIT AND PROFIT
  - 14. In your opinion, what are the main differences of HC between profit and non-profit?
  - 15. Please indicate a positive aspect of the profit sector that should be co-opted for the management and enhancement of HC in the not-for-profit/world of social cooperatives.
  - 16. Indicate a negative aspect of the profit sector in your opinion that should be avoided or in which the not-for-profit can instead set a good example for imitation.
  - 17. In your opinion, is the social impact generated by the not-for-profit sector greater/lesser/equal to that of the profit sector, or it is not possible to answer this?
  - 18. How can the world of research/university help you to improve your work of valorization, measurement, description of the HC?

#### References

- 1. Cheng, M.; Green, W.; Conradie, P.; Konishi, N.; Romi, A. The International Integrated Reporting Framework: Key Issues and Future Research Opportunities. *J. Int. Financ. Manag. Account.* **2014**, *25*, 90–119. [CrossRef]
- 2. Dumay, J.; Dai, T. Integrated thinking as a cultural control? *Meditari Account. Res.* 2017, 25, 574–604. [CrossRef]
- 3. Bebbington, J.; Brown, J.; Frame, B.; Thomson, I. Theorizing engagement: The potential of a critical dialogic approach. *Account. Audit. Account. J.* **2007**, *20*, 356–381. [CrossRef]
- 4. Modell, S. Making institutional accounting research critical: Dead end or new beginning? *Account. Audit. Account. J.* 2015, 28, 773–808. [CrossRef]
- 5. Gray, R. Taking a Long View on What We Now Know About Social and Environmental Accountability and Reporting. *Issues Soc. Environ. Account.* 2007, *1*, 169. [CrossRef]

- 6. Gray, R.; Gray, S. Accountability and human rights: A tentative exploration and a commentary. *Crit. Perspect. Account.* **2011**, 22, 781–789. [CrossRef]
- 7. Gray, R.; Bebbington, J.; Collison, D. NGOs, civil society and accountability: Making the people accountable to capital. *Account. Audit. Account. J.* **2006**, *19*, 319–348. [CrossRef]
- 8. Andreaus, M.; Costa, E. Toward an Integrated Accountability Model for Nonprofit Organizations. In *Advances in Public Interest Accounting*; Costa, E., Parker, L.D., Andreaus, M., Eds.; Emerald Group Publishing Limited: Bingley, UK, 2014; pp. 153–176.
- 9. Molloy, J.C.; Chadwick, C.; Ployhart, R.E.; Golden, S.J. Making Intangibles "Tangible" in Tests of Resource-Based Theory: A Multidisciplinary Construct Validation Approach. *J. Manag.* **2011**, *37*, 1496–1518. [CrossRef]
- 10. Vitolla, F.; Raimo, N. Adoption of Integrated Reporting: Reasons and Benefits—A Case Study Analysis. *Int. J. Bus. Manag.* 2018, 13, 244. [CrossRef]
- 11. Camodeca, R.; Almici, A.; Sagliaschi, U. Strategic information disclosure, integrated reporting and the role of intellectual capital. *J. Intellect. Cap.* **2019**, *20*, 125–143. [CrossRef]
- 12. Manes-Rossi, F.; Nicolò, G.; Tudor, A.T.; Zanellato, G. Drivers of integrated reporting by state-owned enterprises in Europe: A longitudinal analysis. *Meditari Account. Res.* 2020. [CrossRef]
- 13. Del Baldo, M. The implementation of integrating reporting in SMEs: Insights from a pioneering experience in Italy. *Meditari Account. Res.* **2017**, *25*, 505–532. [CrossRef]
- 14. Del Baldo, M. Is It Time for Integrated Reporting in Small and Medium-Sized Enterprises? Reflections on an Italian Experience. In *Corporate Social Responsibility and Governance*; Idowu, S.O., Frederiksen, C.S., Mermod, A.Y., Nielsen, M.E.J., Eds.; CSR, Sustainability, Ethics & Governance; Springer International Publishing: Cham, Switzerland, 2015; pp. 183–209. [CrossRef]
- 15. Eccles, R.G.; Krzus, M.P.; Ribot, S. Meaning and Momentum in the Integrated Reporting Movement. *J. Appl. Corp. Financ.* 2015, 27, 8–17. [CrossRef]
- 16. Rinaldi, L.; Unerman, J.; de Villiers, C. Evaluating the integrated reporting journey: Insights, gaps and agendas for future research. *Account. Audit. Account. J.* **2018**, *31*, 1294–1318. [CrossRef]
- 17. Paternostro, S. Integrated Reporting and Social Disclosure: True Love or Forced Marriage? A Multidimensional Analysis of a Contested Concept. In *Studies in Managerial and Financial Accounting*; Songini, L., Pistoni, A., Baret, P., Kunc, M.H., Eds.; Emerald Publishing Limited: Bingley, UK, 2020; pp. 107–146.
- 18. de Villiers, C.; Hsiao, P.-C.K.; Maroun, W. Developing a conceptual model of influences around integrated reporting, new insights and directions for future research. *Meditari Account. Res.* **2017**, *25*, 450–460. [CrossRef]
- 19. Ortega-Lapiedra, R.; Marco-Fondevila, M.; Scarpellini, S.; Llena-Macarulla, F. Measurement of the Human Capital Applied to the Business Eco-Innovation. *Sustainability* **2019**, *11*, 3263. [CrossRef]
- 20. Beretta, V.; Demartini, C.; Trucco, S. Does environmental, social and governance performance influence intellectual capital disclosure tone in integrated reporting? *J. Intellect. Cap.* **2019**, *20*, 100–124. [CrossRef]
- 21. Mubarik, M.S.; Chandran, V.G.R.; Devadason, E.S. Measuring Human Capital in Small and Medium Manufacturing Enterprises: What Matters? *Soc. Indic. Res.* 2018, 137, 605–623. [CrossRef]
- 22. Antadze, N.; Westley, F.R. Impact Metrics for Social Innovation: Barriers or Bridges to Radical Change? J. Soc. Entrep. 2012, 3, 133–150. [CrossRef]
- 23. Costa, E.; Ramus, T.; Andreaus, M. Accountability as a Managerial Tool in Non-Profit Organizations: Evidence from Italian CSVs. *Volunt. Int. J. Volunt. Nonprofit Organ.* **2011**, *22*, 470–493. [CrossRef]
- 24. Brown, J.; Dillard, J. Integrated reporting: On the need for broadening out and opening up. *Account. Audit. Account. J.* **2014**, 27, 1120–1156. [CrossRef]
- 25. Crawford, L.; Morgan, G.G.; Cordery, C.J. Accountability and not-for-profit organisations: Implications for developing international financial reporting standards. *Financ. Account. Manag.* **2018**, *34*, 181–205. [CrossRef]
- 26. Jiao, L. Multifaceted not-for-profit accountability: Its measurement, cultural context, and impact on perceived social performance. *Financ. Account. Manag.* **2020**. [CrossRef]
- 27. Tejedo-Romero, F.; Araujo, J.F.F.E. The influence of corporate governance characteristics on human capital disclosure: The moderating role of managerial ownership. *J. Intellect. Cap.* **2021**. [CrossRef]
- 28. Secundo, G.; Elena Perez, S.; Martinaitis, Ž.; Leitner, K.H. An Intellectual Capital framework to measure universities' third mission activities. *Technol. Forecast. Soc. Chang.* **2017**, *123*, 229–239. [CrossRef]
- 29. Dumay, J.; Bernardi, C.; Guthrie, J.; Demartini, P. Integrated reporting: A structured literature review. *Account. Forum* **2016**, *40*, 166–185. [CrossRef]
- 30. Cuozzo, B.; Dumay, J.; Palmaccio, M.; Lombardi, R. Intellectual capital disclosure: A structured literature review. *J. Intellect. Cap.* **2017**, *18*, 9–28. [CrossRef]
- 31. Dumay, J. A critical reflection on the future of intellectual capital: From reporting to disclosure. *J. Intellect. Cap.* **2016**, *17*, 168–184. [CrossRef]
- 32. Zambon, S. Ten years after: The past, the present and the future of scholarly investigation on intangibles and intellectual capital (IC). *J. Intellect. Cap.* **2016**, *17*. [CrossRef]
- 33. Nicolò, G.; Aversano, N.; Sannino, G.; Tartaglia Polcini, P. ICD corporate communication and its determinants: Evidence from Italian listed companies' websites. *Meditari Account. Res.* **2017**. [CrossRef]

- 34. Guthrie, J.; Ricceri, F.; Dumay, J. Reflections and projections: A decade of Intellectual Capital Accounting Research. *Br. Account. Rev.* **2012**, *44*, 68–82. [CrossRef]
- 35. Beattie, V.; Smith, S.J. Human capital, value creation and disclosure. J. Hum. Resour. Costing Account. 2010, 14, 262–285. [CrossRef]
- 36. Sveiby, K.E. *The New Organizational Wealth: Managing & Measuring Knowledge-Based Assets*, 1st ed.; Berrett-Koehler Publishers: San Francisco, CA, USA, 1997.
- 37. Ployhart, R.E.; Nyberg, A.J.; Reilly, G.; Maltarich, M.A. Human Capital Is Dead; Long Live Human Capital Resources! *J. Manag.* **2014**, *40*, 371–398. [CrossRef]
- 38. Abeysekera, I.; Guthrie, J. An empirical investigation of annual reporting trends of intellectual capital in Sri Lanka. *Crit. Perspect. Account.* **2005**, *16*, 151–163. [CrossRef]
- 39. Kim, D.; Go, S. Human Capital and Environmental Sustainability. Sustainability 2020, 12, 4736. [CrossRef]
- Hamadamin, H.H.; Atan, T. The Impact of Strategic Human Resource Management Practices on Competitive Advantage Sustainability: The Mediation of Human Capital Development and Employee Commitment. Sustainability 2019, 11, 5782. [CrossRef]
- 41. Bini, L.; Dainelli, F.; Giunta, F. Business model disclosure in the Strategic Report: Entangling intellectual capital in value creation process. *J. Intellect. Cap.* **2016**, *17*, 83–102. [CrossRef]
- 42. Dilling, P.F.A.; Caykoylu, S. Determinants of Companies that Disclose High-Quality Integrated Reports. *Sustainability* **2019**, *11*, 3744. [CrossRef]
- 43. Terblanche, W.; De Villiers, C. The influence of integrated reporting and internationalisation on intellectual capital disclosures. *J. Intellect. Cap.* **2019**, 20, 40–59. [CrossRef]
- 44. Passetti, E.; Cinquini, L. A Comparative Analysis of Human Capital Disclosure in Annual Reports and Sustainability Reports. In *Value Creation, Reporting, and Signaling for Human Capital and Human Assets*; Russ, M., Ed.; Palgrave Macmillan US: New York, NY, USA, 2014; pp. 213–241. [CrossRef]
- 45. Flower, J. The International Integrated Reporting Council: A story of failure. Crit. Perspect. Account. 2015, 27, 1–17. [CrossRef]
- 46. Dumay, J.; Bernardi, C.; Guthrie, J.; La Torre, M. Barriers to implementing the International Integrated Reporting Framework: A contemporary academic perspective. *Meditari Account. Res.* **2017**, 25, 461–480. [CrossRef]
- 47. Thomasson, A. Exploring the ambiguity of hybrid organisations: A stakeholder approach. *Financ. Account. Manag.* **2009**, *25*, 353–366. [CrossRef]
- 48. Cornforth, C. Chapter 13: The Governance of Hybrid Organisations. Handbook on Hybrid Organisations; Edward Elgar Publishing: Cheltenham, UK, 2020; pp. 220–236.
- 49. Ebrahim, A.; Battilana, J.; Mair, J. The governance of social enterprises: Mission drift and accountability challenges in hybrid organizations. *Res. Organ. Behav.* **2014**, *34*, 81–100. [CrossRef]
- 50. Johanson, J.-E.; Vakkuri, J. Governing Hybrid Organisations; Routledge: Abingdon, UK; New York, NY, USA, 2018.
- 51. Augusto Felício, J.; Couto, E.; Caiado, J. Human capital, social capital and organizational performance. *Manag. Decis.* **2014**, *52*, 350–364. [CrossRef]
- 52. Gamerschlag, R.; Moeller, K. The Positive Effects of Human Capital Reporting. Corp. Reput. Rev. 2011, 14, 145–155. [CrossRef]
- 53. Zanda, G.; Lacchini, M.; Oricchio, G. La Valutazione del Capitale Umano Nell'impresa: Modelli Qualitativi e Quantitativi di Logica Economico-Aziendale; G. Giappichelli: Torino, Italy, 1993.
- 54. Qu, S.Q.; Dumay, J. The qualitative research interview. Qual. Res. Account. Manag. 2011, 8, 238–264. [CrossRef]
- 55. Yin, R.K. *Case Study Research: Design and Methods*, 3rd ed.; Applied social research methods series; Sage Publications: Thousand Oaks, CA, USA, 2003.
- 56. Ryan, B.; Scapens, R.W.; Theobald, M. *Research Method and Methodology in Finance and Accounting*, 2nd ed.; South-Western, Cengage Learning: Andover, UK, 2002.
- 57. Galletta, A.; Cross, W.E. Mastering the Semi-Structured Interview and Beyond: From Research Design to Analysis and Publication; NYU Press: New York, NY, USA, 2013.
- 58. Kenno, S.A.; McCracken, S.A.; Salterio, S.E. Financial Reporting Interview-Based Research: A Field Research Primer with an Illustrative Example. *Behav. Res. Account.* **2017**, *29*, 77–102. [CrossRef]
- 59. Francis, J.J.; Johnston, M.; Robertson, C.; Glidewell, L.; Entwistle, V.; Eccles, M.P.; Grimshaw, J.M. What is an adequate sample size? Operationalising data saturation for theory-based interview studies. *Psychol. Health* **2010**, *25*, 1229–1245. [CrossRef]
- 60. Luborsky, M.R.; Rubinstein, R.L. Sampling in Qualitative Research: Rationale, Issues, and Methods. *Res. Aging* **1995**, *17*, 89–113. [CrossRef] [PubMed]
- 61. Marshall, M.N. Sampling for qualitative research. Fam. Pract. 1996, 13, 522–526. [CrossRef] [PubMed]
- 62. Benevene, P.; Cortini, M. Interaction between structural capital and human capital in Italian NPOs: Leadership, organizational culture and human resource management. *J. Intellect. Cap.* **2010**, *11*, 123–139. [CrossRef]
- 63. Stubbs, W.; Higgins, C. Integrated Reporting and internal mechanisms of change. *Account. Audit. Account. J.* **2014**, 27, 1068–1089. [CrossRef]
- 64. Bogner, A.; Littig, B.; Menz, W. Introduction: Expert Interviews—An Introduction to a New Methodological Debate. In *Interviewing Experts*; Bogner, A., Littig, B., Menz, W., Eds.; Palgrave Macmillan UK: London, UK, 2009; pp. 1–13. [CrossRef]
- 65. Thompson, S. Towards a social theory of the firm: Worker cooperatives reconsidered. *J. Coop. Organ. Manag.* **2015**, *3*, 3–13. [CrossRef]

- 66. Bailly, F.; Chapelle, K.; Prouteau, L. Wage differentials between conventional firms and non-worker cooperatives: Analysis of evidence from France. *Compet. Chang.* 2017, *21*, 321–341. [CrossRef]
- 67. Maas, K.; Schaltegger, S.; Crutzen, N. Advancing the integration of corporate sustainability measurement, management and reporting. J. Clean. Prod. 2016, 133, 859–862. [CrossRef]
- Costa, E.; Parker, L.D.; Andreaus, M. The Rise of Social and Non-Profit Organizations and their Relevance for Social Accounting Studies. In *Advances in Public Interest Accounting*; Costa, E., Parker, L.D., Andreaus, M., Eds.; Emerald Group Publishing Limited: Bingley, UK, 2014; pp. 3–21.
- 69. Dillard, J.; Roslender, R. Taking pluralism seriously: Embedded moralities in management accounting and control systems. *Crit. Perspect. Account.* **2011**, 22, 135–147. [CrossRef]
- 70. Cisi, M.; Alice Centrone, F.; Corazza, L. Does the Integrated Reporting's definition of human capital fit with the HR manager's perspective? *Financ. Rep.* 2020, 5–32. [CrossRef]
- 71. Hunt, V.; Prince, S.; Dixon-Fyle, S.; Yee, L. Delivering through Diversity; McKinsey & Company: Chicago, IL, USA, 2018; p. 42.
- 72. Nicholls, A. A General Theory of Social Impact Accounting: Materiality, Uncertainty and Empowerment. *J. Soc. Entrep.* **2018**, *9*, 132–153. [CrossRef]
- 73. Kroeger, A.; Weber, C. Developing a Conceptual Framework for Comparing Social Value Creation. *Acad. Manag. Rev.* 2014, *39*, 513–540. [CrossRef]
- 74. Epstein, M.J.; Yuthas, K. Measuring and Improving Social Impacts: A Guide for Nonprofits, Companies, and Impact Investors; Greenleaf Publishing: Sheffield, UK, 2014.
- 75. Dumay, J.; Guthrie, J.; Farneti, F. Gri Sustainability Reporting Guidelines For Public And Third Sector Organizations: A critical review. *Public Manag. Rev.* 2010, 12, 531–548. [CrossRef]
- 76. Costa, E.; Pesci, C.; Andreaus, M.; Taufer, E. Empathy, closeness, and distance in non-profit accountability. *Account. Audit. Account. J.* **2018**, *32*, 224–254. [CrossRef]
- Quarter, J.; Richmond, B.J. Accounting for Social Value in Nonprofits and For-Profits. Nonprofit Manag. Leadersh. 2001, 12, 75–85. [CrossRef]
- 78. Ryan, C.; Mack, J.; Tooley, S.; Irvine, H. Do Not-For-Profits Need Their Own Conceptual Framework?: A NFP Conceptual Framework? *Financ. Account. Manag.* **2014**, *30*, 383–402. [CrossRef]
- 79. Hussinki, H.; Kianto, A.; Vanhala, M.; Ritala, P. Happy Employees Make Happy Customers: The Role of Intellectual Capital in Supporting Sustainable Value Creation in Organizations. In *Intellectual Capital Management as a Driver of Sustainability*; Matos, F., Vairinhos, V., Selig, P.M., Edvinsson, L., Eds.; Springer International Publishing: Cham, Switzerland, 2019; pp. 101–117. [CrossRef]
- 80. McNally, M.-A.; Cerbone, D.; Maroun, W. Exploring the challenges of preparing an integrated report. *Meditari Account. Res.* 2017, 25, 481–504. [CrossRef]
- 81. Guthrie, J.; Manes-Rossi, F.; Orelli, R.L. Integrated reporting and integrated thinking in Italian public sector organisations. *Meditari Account. Res.* **2017**, *25*, 553–573. [CrossRef]
- 82. Brown, J.; Dillard, J. Critical accounting and communicative action: On the limits of consensual deliberation. *Crit. Perspect. Account.* **2013**, *24*, 176–190. [CrossRef]
- 83. Girella, L.; Dameri, P. Putting Integrated Reporting Where It Was Not: The Case of the Not-for-Profit Sector. *Financ. Rep.* **2019**, 111–140. [CrossRef]
- 84. Girella, L.; Zambon, S.; Rossi, P. Reporting on sustainable development: A comparison of three Italian small and medium-sized enterprises. *Corp. Soc. Responsib. Environ. Manag.* 2019, 26, 981–996. [CrossRef]





# Article Social Finance and Banking Research as a Driver for Sustainable Development: A Bibliometric Analysis

Silvana Secinaro<sup>1</sup>, Davide Calandra<sup>1,\*</sup>, Denisa Petricean<sup>2</sup> and Federico Chmet<sup>1</sup>

- Department of Management, University of Turin, 10134 Turin, Italy; silvana.secinaro@unito.it (S.S.); federico.chmet@unito.it (F.C.)
- <sup>2</sup> Brunel University London, London UB8 3PH, UK; denisa.petricean@brunel.ac.uk
- \* Correspondence: davide.calandra@unito.it

Abstract: Social finance and banking with an embedded social purpose have been on the rise in recent decades. Social entrepreneurs have repeatedly stressed the critical need for financial support from social banks. This study aims to provide a bibliometric analysis of the status of the field in social finance and banking, recognising main topics from existing research and establishing future re-search challenges. Our study used science mapping workflow and multiple research questions to investigate the broad literature about social banking and finance. With in-depth bibliometric analysis, authors examined qualitative and quantitative variables as primary research infor-mation, relevant sources, subject areas, authors data, social, thematic and intellectual structure. The data was retrieved from Web of Science (WOS) and then analysed using Bibliometrix R-package. The analysis was based on a sample of 270 articles and demonstrates a multidisciplinary vision of the research flow investigated. Our results show several insights regarding journals, authors and geographical interest of this research stream. Specifically, the literature, although dwelling on social finance and banking, includes five theoretical and practical clusters as (1) people's well-being, combined with technological innovation, (2) governance, (3) ethical investment and sustainable development, (4) corporate social responsibility (CSR), and (5) transparency. The authors also note a line of research that observes technological solutions for the response to social and environmental problems. These results may be useful for researchers, practitioners, and policymakers to foster social finance and financial system tools.

Keywords: social finance; social banking; sustainable development; bibliometric analysis

#### 1. Introduction

Investment of capital for social and financial creation has ancient roots. Cooperatives, religious institutions, credit unions and mutual societies have been managing capitals for the social project for many years back. However, during recent times, new ways of managing finance for achieving positive social impacts have emerged, which have been classified under the term "Social Finance" (SF).

Past literature reviews have defined SF as an alternative financing tool for projects and ventures with financial returns that are also able to generate a positive impact on society, the environment and sustainable development [1–4].

On the same line is the definition of Périlleux [5] who explains SF as the sum of financial institutions, products and practices in which not only the maximisation of profit is sought but also social, ecological or ethical results. Additionally, SF includes initiatives such as savings groups, microfinance and collaborative finance (such as social crowdfunding).

More generally, the term SF indicates different models and approaches, such as crowdfunding, alternative currencies, social investment, microfinance, socially responsible investments, social impact bonds and venture philanthropy [5–8]. Therefore, as suggested by the latest results of Rizzi et al. (2018) [9], two forms of SF have emerged, including primarily social impact investment and ethical banking.

Citation: Secinaro, S.; Calandra, D.; Petricean, D.; Chmet, F. Social Finance and Banking Research as a Driver for Sustainable Development: A Bibliometric Analysis. *Sustainability* 2021, *13*, 330. https://doi.org/ 10.3390/su13010330

Received: 6 December 2020 Accepted: 29 December 2020 Published: 31 December 2020

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/licenses/by/4.0/). Moreover, when talking about SF, it is important to mention "Social Banking" (SB) as well. As is the case for SF, there is not only one correct definition of SB. For many people, SB could sound like an oxymoron, which is to combine two words that do not belong together. To others, SB refers to these institutions directed to charitable clients.

SB is also associated with governmental institutions, in addition to microfinance. Thus, Frans de Clerck, co-founder of Triodos Belgium and former head of the supervisory board of the Institute for Social Banking, concludes that there is no clear and unique interpretation of SB, as social, alternative, and ethical banks are definitions which refer to methods of giving money. According to the authors, therefore, there is no precise definition of SB as it belongs to different financial traditions [10].

Although De Clerck's definition remains interesting, for this paper, we adopt a more technical explanation. As defined by James Niven, from the Global Alliance for Banking on Values it is possible to actively engage in wide-ranging discussions about the future of the financial sector in order to have a direct impact on organisations which invests in social, environmental, and cultural issues. Hence, in line with Niven's argument, we define SB as banking where the main purpose is to have a positive impact on people and communities [11].

All these premises give the idea of two parallel research models that overlap in practice but demonstrate the inability of the literature to provide an analysis of the research flow conducted so far. This research gap has recently been highlighted by Rizzi et al. (2018) [9], which in its conclusions calls for the convergence of the two models in order to advance the development of the sector concerning commercial finance. Additionally, also the recent publications highlight the importance of sustainable financing tools for social purposes [12–14].

Therefore, this article proposes a broader investigation that includes the study of the primary bibliometric data [15–20] on peer-reviewed papers listed on the Web of Science (WOS).

The main results denote a multidisciplinary vision of the field of research studied. The literature includes an interesting discussion of the tools of social finance, sustainable development, social responsibility, and human rights challenges. Also, we note a research strand that observes technological solutions for the response to social and environmental problems. Our analysis is innovative for the topic of interest as it applies a methodology not yet used. Moreover, and more importantly, it affirmed that the social and right approach is not only related to the person but also the environment.

Focusing on SF and SB, our paper aims to map, discuss, and critique the research discussion on these topics answering the following questions: (1) what are the main features of these research streams considering authors, citations and geographical interest? (2) What are the most frequent issues and topics of this literature? (3) What are the possible implications for future research in this field?

The rest of the paper is structured as follows. Section 2 defines the current literature and debates why a bibliometric analysis using open coding methods may be useful to reach our research aims. Section 3 establishes the methodology flow. Section 4 shows the results of our research. Section 5 provides an in-depth data interpretation, commenting on and critiquing the main findings. In conclusion, Section 6 concludes this paper considering present implications and suggesting future research paths.

#### 2. Background

During the past decades, SF and SB have gained importance and popularity, growing starting to grow significantly and attracting the interest of different stakeholders ranging from governmental agencies to mainstream financial markets. Although the accessible data on SF is limited due to the fragmentation of sectors and investment approaches, the market potential has been estimated to be extremely prosperous. For instance, microfinance, which can be considered an earlier example of SF, is estimated to contain over over USD \$50 billion of loans to over 100 million micro-entrepreneurs [21]. Another study conducted by

the European Investment Forum outlined that the European SF market showed around USD \$21 billion in 2013 in France, Italy, Germany, the Netherlands and the United Kingdom (UK), representing the leading markets [22].

SF in Europe is developing in different ways and forms, driven by other actors. In the beginning, SF was created to assist social ventures with their financial needs. Even though SF can create both social and financial value, social ventures sometimes face difficulties accessing different financial markets due to their nature and high risk [23–27]. Secondly, most European governments have recognised the potential and possibile of growth of SF. In this case, governments have been sustaining the development of SF through direct and indirect investment or through developing particular policies to facilitate SF markets [28]. Thirdly, SF provides the perfect scenario for the growth and demand of corporate social responsibility (CSR), firms in the financing and banking sector [29]. Lastly, SF attracts socially-minded investors, such as philanthropic organisations, or socially motivated people, who are continuously researching new methods of allocating financial resources to socially motivated firms, to not only returns but also keep their main priority, which is the "bettering of the society" [3,30].

Nevertheless, accessing enough financial resources is highly challenging for social enterprises [31]. As conventional funding in the social sector is scarce and limited to specific projects, they do not allow investment in overheads or product and service development. However, recently social enterprises have been gaining interest from venture capitalists because of their fast growth in terms of generating income [32]. Financial instruments considered in the discussion are mostly private equity, unsecured debt and their hybrids [28]. Notwithstanding, investors do not solely value just financial returns. The concept of impact investing [33,34] refers to proactively perusing social and ecological features together with financial goals. In this investing, the creation of measurable social impact is the primary goal, and financial returns are considered as additional benefits. During the last few years, social impact investing has gained traction in the USA and UK with an increased number of private impact investing funds spread across the globe [35].

The growing interest in the literature and the intersection of SF and SB theories, however, are not accompanied by specific bibliometric studies. For example, recent studies have focused on vertical research flows such as social entrepreneurship [36], the social aspects of Islamic finance [37] and sustainable banking investments [38], leaving aside a relevant research gap in addition to what Rizzi et al. (2018) [9] have pointed out, as well as the social impact outlined in Sustainability journal [12].

#### 3. Methodology

To generate a dataset of relevant sources, the authors started with keyword selection as the giant step after research questions were assessed [39]. Our analysis considered "So-cial finance" and "Social Banking" as a research strategy. This strategy allowed us to fo-cus the research inclusively and explore the two research flows [40–42]. Keyword selection was outlined by Chen & Xiao (2016) [43] in a paper which suggested an analysis from the top-down level, considering macro-field analysis for the identification of broad research links. Additionally, our dataset creation benefits from the study of past relevant mile-stones [17,37,44,45] which allowed keyword analysis to consider qualitative and more re-strictive criteria. Therefore, the ABS list (2018) with its 1, 2, 3, 4 and 4\* ABS star classification was chosen as it has been widely adopted as a policy tool and researchers commonly use it for the identification of quality scientific articles [46–49].

Besides, the authors use more research's filter.

Our analysis is based on an emerging and continually evolving field of research, therefore we have focused on all sources in the database including not only peer-review articles but also conferences as a source of knowledge [50].

Moreover, the articles are selected from the business, management, and accounting fields due to the research questions strictly related to these three studied fields [51].

Finally, to increase the understandability and quality of publications, the authors considered only English language and peer-reviewed articles [17,20]. Therefore, all non-scientific sources such as non-peer-reviewed books, white papers and popular articles were not considered in our analysis.

Applying this previous restrictive research criteria, the authors selected 270 articles for further analysis. After the definition of the field of study considering the literature gap, the selection of the proper database, and the identification of the more restrictive research criteria, the authors began the analysis using Bibliometrix a statistical package available on R-Studio [52]. The software allowed us to retrieve relevant bibliometric qualitative and quantitative information such as authors, citations, country of production, or keywords. Figure 1 shows all the methodological flow followed by the authors.



Figure 1. Characteristics of the bibliometric study. Source: Author's elaboration.

#### 4. Results

As previously mentioned, this bibliometric analysis started by describing the dataset characteristics (and related implications) of the 270 records. The scientific production re-search of a specific field is relevant for understanding the evolution of the literature, de-tecting research trends and organizing past research to suggest future research lines. Therefore, in the next paragraphs, we present primary research flow information, relevant sources, subject areas, authors data, social, thematic and intellectual structure.

#### 4.1. Main Information

The identified articles covered a period ranging from 1986 to 2019. It is only from 2015 that publications started to become systematic; the number of publications also be-gan to be consistent and steadily grew (Figure 2).

The reason for this research intensification is related to the fact that in 2015 all United Nations Member States adopted the 2030 Agenda for Sustainable Development, with the 17 Sustainable Development Goals (SDGs) at its heart. Furthermore, since that year in a more pronounced way, states and the scientific world have focused more attention on sustainability in all its forms. This can also explain the rise to the peak of publications on sustainability.





The data suggests that the topic has ancient origins. As can be seen, the quotations did not follow precisely the same trend as the publications—the citations up to 2010 were on the rise and reached over a thousand, after which there was a decrease. These findings suggest poor productivity in the early years. Table 1 indicates that researchers are aiming to solve this issue with a developing number of collaborations. In comparison, the number of publications and countries reporting on this issue has been gradually growing.

Variables	Explanation	Results
Timespan	Year of publication	1986–2019
Sources (journals, books, etc.)	The frequency distribution of sources as journals	19
Documents (articles)	Total number of documents	270
Average years from publication	-	5.96
Average citations per document	The average number of citations in each document	19.19
Average citations per year per document	The average number of citations in each article	2411
References	Total number of references	13.964
	DOCUMENT CONTENTS	
Keywords plus (ID)	Total number of phrases that frequently appear in the title and on the article's references	832
Author's keywords (DE)	Author's keywords (DE) Total number of keywords	
-	AUTHORS	
Authors	Total number of authors	741
Author appearances	The authors' frequency distribution	774
Authors of single-authored documents	The number of single authors per article	43
Authors of multi-authored documents The number of authors of multi-authored articles		698
	AUTHORS COLLABORATION	
Single-authored documents	-	44
Documents per author	-	0.364
Authors per document	-	2.74
Co-Authors per documents	-	2.87
Collaboration index	-	3.09

Table 1. Main information.

Source: Authors' elaboration.
## 4.2. Sources

The top ten sources corresponded to more than half of the scientific production. Table 2 indicates a cumulative number of papers, with 40 articles and 649 citations, in the Journal of Cleaner Development. Furthermore, 23 articles and 618 quotes were included in the Journal of Business Ethics, listed second in the list. In turn, three journals with 10, 8 and 5 articles respectively (see Table 3), for sustainable entrepreneurship, were among the most quoted. They are Energy Policy, Ecological Economics and Tourism Management.

#### Table 2. Journal analysis.

Journal	Number of Articles	H-Index	Citations
Journal of Cleaner Production	40	14	649
Journal of Business Ethics	23	13	618
Physica A: Statistical Mechanics and its Applications	17	7	189
Journal of Business Research	13	6	230
Journal of Social Entrepreneurship	11	4	35
Social Enterprise Journal	11	3	37
Technological Forecasting and Social Change	11	6	143
Energy Policy	10	6	371
Human Organization	10	7	135
Computers in Human Behavior	9	6	157

Source: Authors' elaboration.

Authors	Α	С	C/A	1st A	Last A	H-Index	Country	Affiliation
Aseem Kaul	3	28	9.33	2018	2019	11	USA	University of Minnesota
Jiao Luo	3	28	9.33	2018	2019	9	USA	University of Minnesota
Jarrod Ormiston	3	14	4.67	2015	2019	4	Netherlands	Maastricht University
Jing Shao	3	26	8.67	2016	2016	8	China	Northwestern Polytechnical University
Frank Vanclay	3	46	15.33	2017	2018	57	Netherlands	University of Groningen
Irene Bengo	2	18	9.00	2016	2018	7	Italy	Politecnico di Milano
Mario Calderini	2	18	9.00	2016	2018	17	Italy	Politecnico di Milano
Jana Dlouhà	2	36	18.00	2013	2018	13	Czech Republic	Charles University
Bob Doherty	2	2	1.00	2019	2019	29	UK	The University of York
Daniel M. Franks	2	39	19.50	2012	2014	28	Australia	University of Queensland

**Table 3.** The top 10 most productive authors.

A: Number of total articles; C: Number of citations for all papers; C/A: Average citation per article; 1st A: Year of first published article; Last A: Year of last published article. Source: Authors' elaboration.

### 4.3. Subject Areas

Figure 3 indicates the key topic areas for the online database on social finance and banking for science output. The figure demonstrates the multidisciplinary of the social finance and banking theme. The most relevant subject areas were Business with 16%, followed by Environmental Sciences with 15% and Engineering Environmental with 8%.

The other subject category refers to Physics Multidisciplinary, Anthropology, Social Sciences Interdisciplinary, Energy, Fuels, Psychology Experimental, Psychology Multidisciplinary, Ecology, Geography, Development Studies, Hospitality Leisure Sport Tourism, Rehabilitation, Business Finance, Health Care Sciences Services, and Information Science Library Science.



Figure 3. Subject areas that stand out in social finance and banking; Source: authors' elaboration based on Web of Science.

# 4.4. Social Structure (Countries, Institutions, and Relevant Authors)

As shown in Figure 4, several countries have published articles on social finance. However, most of the published articles were concentrated in a few countries. As expected, English-speaking countries (such as USA, UK and Australia) were the most represented. There was also a substantial presence of European countries. This may be due to the presence of a very developed financial/bank market [53,54].



Figure 4. Collaboration WorldMap. Source: author's elaboration with Bibliometrix R-Package.

The results of the collaboration were also reflected in the country of production; the first three countries confirmed to be those that natively speak English, the first in the

USA with 79 papers, the UK with 65 papers and Australia with 56 papers. Then followed European countries, such as Spain with 44 papers and Italy with 35 papers. The results were also reflected in the affiliations, the most relevant with seven articles published were Texas A&M University (however the citations amounted to only 11) and the University of Groningen, whose scientific production was cited 76 times. The third university in terms of production was the University of Sydney which published six papers, with the scientific output mentioned 99 times. As for the most cited affiliation, this was from the University of Washington, which published only two articles, but the paper Dyer et al. (2007) was the most mentioned of the sample analysed with 242 citations.

Considering the most productive authors, some useful insights have emerged from the bibliometric analysis. Although it has been verified that the theme had its roots already in the 1980s, there is no referenced author. Table 3 shows that the maximum number of articles written by these authors was five and three; also, five authors who wrote the articles: Aseem Kaul, Jiao Luo, Jarrod Ormiston, Jing Shao and Frank Vanclay. It turned out that the first two authors Aseem Kaul and Jiao Luo, collaborated on the same articles (which are Kaul & Luo, 2018; Luo et al., 2018; and Luo & Kaul, 2019). The same can also be seen for Irene Bengo and Mario Calderini who collaborated in writing the papers "Unlocking finance for social tech start-ups: Is there a new opportunity space?" and "New development: Are social impact bonds (SIBs) viable in Italy? A new roadmap" (Arena et al., 2018; Bengo & Calderini, 2016). Among the authors analysed in this paper, only 23 had published two articles, while the remaining 713 had only published one article each. It appears that the first ten authors had published their scientific production only in the last eight years, so we cannot identify an author who has contributed to the development of the topic over the years.

#### 4.5. Conceptual Structure (Trends and Thematic Analysis)

The conceptual and functional structure to analyse the research field and to identify and visualise its conceptual subdomains (i.e., topics/themes or general thematic areas) is also achieved by analyzing its thematic evolution [55]. These topics can be interpreted as textual knowledge conglomerates or as semantic/conceptual classes with various themes discussed in the field of study [56]. Therefore, this section includes an analysis of trend topics thematically related in this field and the topic dendrogram.

Figure 5 defines the trend topics, based on the Keywords Plus, which are words or phrases that often appeared in the titles of article references, but not in the title of the arti-cle itself. Based on an algorithm developed by Clarivate Analytics, Keywords Plus in-creases the search power of cited references by searching in all disciplines for all articles that have cited references in common. Since Keywords Plus terms are derived from the ti-tles of articles cited by the author of the article to be indexed, articles without references and articles whose references are not linked to the source articles will not have Keywords Plus. Keywords Plus may also be present for articles that do not have the author's key-words or may include important terms not listed among the author's keywords.

Garfield claimed that Keywords Plus terms could capture an article's content with greater accuracy and variety [57]. Keywords Plus is as powerful as Author Keywords in the bibliometric study of the information structure in science areas but is less detailed in the representation of the substance of the paper [58]. Table 4 highlights the whole number of Keywords Plus in the top 20 positions.

Figure 5 highlights that, in the analysed period, there was a change in the trend of the issues; in 2012 researchers focused their studies on strategy and quality of life, from 2014 to 2016 they focused on social impact and, recently, they focused on sustainability, entre-preneurship and the performance of the third sector. As can be seen from the figure un-derlying the study of the social impact, i.e., the study of the effect that the actions of a fi-nancial organization have on the community of reference [59], scholars began to develop this consistently from 2014 onwards.



Figure 5. Trend topics. Source: Authors' elaboration.

**Table 4.** Journal analysis.

Keywords Plus (Top 20)	Articles		
Entrepreneurship	23		
Framework	21		
Management	19		
Performance	18		
Sustainability	16		
Innovation	15		
Impact	14		
Social impact	14		
Behavior	12		
Governance	10		
Model	10		
Perceptions	10		
Challenges	9		
Consumption	9		
Impacts	9		
Responsibility	9		
Business	8		
Care	8		
Community	8		
Energy	8		

Source: Authors' elaboration.

Figure 6 explains the composition of the academic sector at the conclusion of the study. The thematic map shows each keyword in the two-dimensional space explained by the axes [60]. The scale of the points is proportional to the cumulative input of each key-word. Through applying a clustering algorithm to a keyword network, it is possible to highlight the various themes of a given domain. Each cluster may be depicted on a special plot known as a strategic or thematic map. For example, centrality can be read as the im-portance of the theme in the whole field of science, and density can be read as a measure of the growth of the theme. Therefore, considering the figure, in this research flow as high-ly developed and isolated themes, authors found the risk and the perception among stakeholders of social banking and finance. As necessary and transversal themes, authors



also found that performance of social finance was a highly repetitive theme. Moreover, considering the figure, the issue of sustainability is also acting as a "deus ex machina" towards this sector. Finally, as a declining theme, the following assistance was reported.

Centrality

Figure 6. Thematic map in social finance and banking. Source: Authors' elaboration.

Authors often recognise problems that hold hybrid positions, i.e., they comprise two quadrants. For instance, the social impact theme occupied a position between emerging or declining themes and highly developed and isolated themes, which reflects the theoretical links about this topic. The need for a common framework occupying a position between emerging or declining themes and basic and transversal themes, and finally, entrepreneurship and management, occupied a position between motor themes and basic and transversal themes.

Figure 7 shows a dendrogram to observe the hierarchical relationship between keywords produced by hierarchical clustering [61]. The diagram shows two lines of analysis relating to the colours blue and red, respectively. It is used to assign objects to clusters by calculating the height of the various objects that are connected in the divisions. Humans are originally depicted in the dendrogram diagram by the researchers. The blue strand can be divided into four clusters, while the red strand consists of only one group. Each subdivision represents relevant themes and keywords. The results found that the flow of literature on social finance and banking included (1), peoples' well-being combined with technological innovation, (2) governance, (3) ethical investment and (4) corporate social responsibility (CSR). Finally, common to each of the above issues was the transparency (5) that the subject required. The combination of multi-disciplinary matters shows that social finance is an instrument aimed at improving peoples' well-being through entrepreneurial actions and projects. Moreover, the flow of re-search on social banking brings tools such as ethical investments and CSR vision to the previous literature. Also, the two themes at the theoretical approach level require greater transparency and consequently, governance of processes than the traditional financial system.



Figure 7. Topic dendrogram. Source: Authors' elaboration.

## 4.6. Conceptual Structure (Trends and Thematic Analysis)

The following paragraph aims outlines the results of co-citation analysis including historiographic parameters [44]. The author's co-citation analysis offers a variety of valuable perspectives to consider the effect of these scholars on the academic framework of the social finance and banking industry.

Figure 8 shows the social finance and banking impact co-citation analysis and the intellectual structure. The size of each cluster indicates the degree of co-citation of the authors, and the width of the lines shows the strength of the co-citation relations. The proximity between writers is seen by connections, not by spatial proximity to each other. Network edges can have varying meanings based on the citation type (co-citation or direct citation); citation analysis is the most common bibliometric analysis in the context of co-citation between authors or records. Co-citation research, as analysed over time, tends to discern a change in paradigms in schools of thought [62].

It is possible to classify three clusters from the following diagram.

Subgroups were identified in six teams of different colours: blue, purple, green, red, and orange. The brown sub-cluster was an outlier. The third orange cluster, composed of only two bi-univocal co-citations, was also defined as an outlier, for the distance from the centre of the co-citations. Above the centre of gravity, the first cluster was marked in blue, the theme of which can be described as theory and social structure. The purple cluster at the core of the study consisted of unique research on the effects of social finance and banking, comprised of specific studies. The green theme that formed around the purple theme and around the centre of gravity, was defined as the source of the focus of the study.

Moving on to historiographic analysis, it is possible to see chronological tables which highlight the most-cited works in and outside the collection. This is used to help scholars quickly identify the most significant work on this topic and trace its year-by-year historical development [63]. Historical path identifies a research topic and its core authors/documents, also, each node represents a document (included in the analysed collection) cited by other papers [64]. Each edge represents a direct citation. Nodes and edges were plotted on an oriented graph where the horizontal axis represents the publication years. Considering Figure 9, it is possible to identify three research paths, which were circled to make them easily identifiable.



Figure 8. Co-citation analysis. Source: Authors' elaboration.



Figure 9. Historiographic parameters. Source: Authors' elaboration.

The first path is around the authors Cetindamar & Ozkazanc-Pan (2017), Ormiston et al. (2015), and Tullberg (2012) [65–67]. The first path was identifiable with the knowledge of the structure, and the publications that traced the main foundations of the theme of finance and banking impact, investment impact and the development of this sector. The second path is around the authors Bacq & Eddleston (2018), Herrera (2016), Maas & Grieco (2017) and Smith et al. (2016) [68–71]), and is about social entrepreneurship and its role and impact on social challenges. And the third path is around the authors Agyekum et al. (2017), Neugebauer et al. (2017), Prasara-A & Gheewala (2018) and Subramanian & Yung (2018) [72–75], with publications about future social finance guidelines.

The examined scientific production focused on the social issues of the community of reference concerning the standard of living and well-being of the same and the environmental impact assessment.

# 5. Discussion

Our bibliometric analysis focused on SF and SB research streams. The study aimed to investigate qualitative and quantitative variables such as authors, citations, journals and topics. Among them, some interesting insights emerged. For instance, Aseem Kau and Jiao Luo were the authors with the highest number of publications. Their contributions focused on corporate philanthropy as a form of reputation assurance; additionally, their work aimed to develop new theories for dealing with social issues. Other colleagues, such as Jarrod Ormiston, focused more on social impact investment as a social finance tool. Jing Shao focused mostly on sustainable development behaviour, seeing the link between social finance and the environment. Frank Vanclay recognised and addressed problems in implementing organizational human rights responsibility. Irene Bengo and Mario Calderini focused their scientific production on social impact bonds (SIBs) in Italy and financial tools that could be used to finance social innovation, with a particular emphasis on social technology start-ups that create and deliver technology-driven innovations to meet social challenges in a financially viable manner.

Additionally, our analysis shows that the journals with the highest number of articles on the topics were the Journal of Cleaner Production with 40 publications and the Journal of Business Ethics with 23 publications. Concerning the Journal of Cleaner Production, the publications referred to the link among sustainability, social impact and impact finance. At the same time, the Journal of Business Ethics considered ethical banking and social finance.

From the analysis conducted it was found that there was a consistent scientific production on ethical banking, finance impact and social impact, but only one paper was found to analyse what the social impact generated by the bank. Following the theories and historical processes of the bank's development in parallel with finance [4,76], it is undoubtedly of central interest to study the social impact it creates in ethical banking.

Keywords analysis on SF and SB revealed some important insights.

Cluster one related to peoples' well-being combined with technological innovation. The research strand confirmed the trend in the literature of collaboration between research centres to address and solve complex problems observed in the real world [45,77,78]. For instance, the paper of McCrea et al. [79] discussed community spirit and social investment in managing the development of new gas technology networks. Another topic was the analysis of the social impact of new fair wage models in the agricultural sector, showing how social impact assessment could be an essential decision-making tool for companies [73].

The second cluster related to governance, mainly concerning social impact. For example, as Luo e Kaul [80] discussed, efficient organisational forms are increasingly need-ed to address issues of social concern. The authors advocated the use of various governance instruments such as public-private partnerships as well as social enterprises. Ale-do-Tur and Domínguez-Gómez [81] also discussed this issue in order to stress how social impact assessment needs many tools from different fields, including governance.

The third cluster analysed referred to ethical investments.

Although at first glance, it could only refer to SB, the research area included a broad discussion. For example, the results of Schinckus [82] demonstrated how financial innovation in ethical investment (e.g., social bonds [83]) meets social impact by favouring funding for social projects such as rehabilitating prisoners rather than helping the homeless people. As authors, we believe that this paper is the academic and practical synthesis of how SF and SB can quickly meet. Finally, we also consider Revelli contribution [84] used a theoretical approach; it demonstrating how ethical and socially responsible investments require constant measurement of impact and extra-financial performance.

The fourth cluster concerned corporate social responsibility (CSR). Although it is a vast territory, the links between CSR, financial performance and social impact are many.

For example, several authors showed that higher CSR had a positive value on companies' financial performance, on the social impact created towards stakeholders, and on the environmental impact of companies [85–87].

Finally, the last cluster included all the previous ones. Several authors emphasised that SF and SB research topics require greater transparency of information, especially about projects with a social and environmental purpose [32,82,88].

#### 6. Conclusions

To conclude, our study used science mapping workflow and multiple research questions to investigate the broad literature about social banking and finance yet unexplored. For accurately answering these research questions, we conducted a bibliometric analysis by using Bibliometrix R-package. Starting from the gap of the literature which saw two parallel literatures among the topics investigated [9], this paper provides an in-depth qualitative and quantitative analysis on bibliometric variables. Some useful theoretical in-sights can be made.

Firstly, scientific production in this area concerns social impact, social and sustainable finance and ethical banking. Therefore, as shown by Figure 6, banking and financial systems should engage and create useful tools for sustainable development, such as social and environmental projects, respectively [12].

Secondly, from a theoretical point of view, the analysis did not find parallel and competing issues. The research shows that social finance and social banking were not on different levels but instead that the banking approach provides practical tools to achieve more significant social impact for entrepreneurial choices. Therefore, we observed a union of the two pieces of literature and not a mere thematic division.

Thirdly, the keyword "social" does not only coincide with the aspect of caring for and fa-vouring others through social initiatives. The social element is linked to environmental good. For this reason, the topic has been widely studied by international journals such as Sustainability, which creates a tangible link considering sustainable development [89].

Moreover, our analysis also contained some practical insights.

Firstly, entrepreneurs and policymakers should aspire to and demand more transparency, as suggested by cluster five discovered in our article. The theoretical link between the topics discussed, has practical insights in terms of reporting. The social approach requires more transparency, and the social finance and banking sector needs to provide more answers in terms of writing. Therefore, more practical accountability may be required to include both social and environmental impact assessment methodologies created for funded activities.

Secondly, the contributions and critical words in this research flow call for multistakeholder and multidisciplinary governance that encourages social enterprises to convey their role as protagonists. Therefore, policymakers will be able to benefit from the analysis by noting how collaboration is valued at a scientific level and could enhance social entrepreneurship initiatives for the good of the community and the environment.

Our article opens exciting scenarios for future research.

The analysis shows the beginning of research in new technologies. Future studies could be conducted on blockchain technology and the ability to track investments within social finance. Furthermore, case studies could investigate the role of trust in sustainable investments, transparency, and decentralisation. Moreover, an interesting point for future research could be the way to deal with cryptocurrency transactions, which will most likely lead to a higher trust in the social finance sector activities attracting a higher number of donors or investors and benefiting the under-privileged communities.

Further studies should also be conducted to understand whether social and environmental impact assessments could be a tool to stimulate social finance initiatives. More in-depth studies could also be achieved in terms of the social impact created by the banking system. Researchers could verify the level of responsibility of financial institutions in solving social, ethical and environmental problems. Furthermore, the research should focus on comparing academic perspectives of this research flow from our contribution, assessing the demands of professionals.

Finally, like all research, our study has limitations.

Firstly, to ensure a rigorous methodology, we have limited the articles within the sample. This may have caused us to lose interesting publications and proceedings for research. Secondly, considering the analysis of bibliometric variables, future studies could dwell on broader samples of literature also finalising investigations with open codes for better identification of the scientific topics discussed.

**Author Contributions:** Conceptualization, S.S.; methodology, F.C. and D.P.; software, F.C. and D.P.; validation, D.C.; formal analysis, S.S. and D.C.; investigation, D.C.; data curation, F.C. and D.P.; writing—original draft preparation, F.C. and D.P.; writing—review and editing, D.C.; visualization, D.C., D.P., F.C.; supervision, S.S. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** The data are available on request; however, it is possible to extract the bibliometric source directly from the databased used by the authors.

**Conflicts of Interest:** The authors declare no conflict of interest.

### References

- 1. Bishop, M.; Green, M. Philanthrocapitalism: How Giving Can Save the World; Bloomsbury Publishing: New York, NY, USA, 2010.
- 2. Emerson, J.; Spitzer, J. From Fragmentation to Function: Critical Concepts and Writing on Social Capital Markets' Structure, Operation and Innovation; University of Oxford: Oxford, UK, 2007.
- Nicholls, A. The Legitimacy of Social Entrepreneurship: Reflexive Isomorphism in a Pre-Paradigmatic Field. *Entrep. Theory Pract.* 2010, 34, 611–633. [CrossRef]
- 4. Weber, O.; Duan, Y. Social Finance and Banking. In *Socially Responsible Finance and Investing: Financial Institutions, Corporations, Investors, and Activists;* John Wiley & Sons: Hoboken, NJ, USA, 2012; Volume 160, p. 180.
- 5. Périlleux, A. When Social Enterprises Engage in Finance: Agents of Change in Lending Relationships, a Belgian Typology. *Strateg. Chang.* **2015**, *24*, 285–300. [CrossRef]
- Allison, T.H.; Davis, B.C.; Short, J.C.; Webb, J.W. Crowdfunding in a Prosocial Microlending Environment: Examining the Role of Intrinsic versus Extrinsic Cues. *Entrep. Theory Pract.* 2015, 39, 53–73. [CrossRef]
- 7. Howard, E. Challenges and Opportunities in Social Finance in the UK.; Cicero Group: Washington, DC, USA, 2012.
- Martínez-Gómez, C.; Jiménez-Jiménez, F.; Alba-Fernández, M.V. Determinants of Overfunding in Equity Crowdfunding: An Empirical Study in the UK and Spain. Sustainability 2020, 12, 10054. [CrossRef]
- Rizzi, F.; Pellegrini, C.; Battaglia, M. The Structuring of Social Finance: Emerging Approaches for Supporting Environmentally and Socially Impactful Projects. J. Clean. Prod. 2018, 170, 805–817. [CrossRef]
- 10. De Clerck, F. Ethical banking. In Ethical Prospects; Springer: Berlin/Heidelberg, Germany, 2009; pp. 209–227.
- 11. Weber, O.; Remer, S. Social Banks and the Future of Sustainable Finance; Taylor & Francis: Oxfordshire, UK, 2011; Volume 64.
- 12. Baraibar-Diez, E.; Luna, M.; Odriozola, M.D.; Llorente, I. Mapping Social Impact: A Bibliometric Analysis. *Sustainability* **2020**, *12*, 9389. [CrossRef]
- 13. Höhnke, N. Doing Good or Avoiding Evil? An Explorative Study of Depositors' Reasons for Choosing Social Banks in the Pre and Post Crisis Eras. *Sustainability* **2020**, *12*, 10082. [CrossRef]
- 14. Rizzello, A.; Kabli, A. Sustainable Financial Partnerships for the SDGs: The Case of Social Impact Bonds. *Sustainability* **2020**, *12*, 5362. [CrossRef]
- 15. Dal Mas, F.; Massaro, M.; Lombardi, R.; Garlatti, A. From Output to Outcome Measures in the Public Sector: A Structured Literature Review. *Int. J. Organ. Anal.* **2019**, *27*, 1631–1656. [CrossRef]
- 16. Dumay, J.; Cai, L. A Review and Critique of Content Analysis as a Methodology for Inquiring into IC Disclosure. *J. Intellect. Cap.* **2014**, *15*, 264–290. [CrossRef]
- 17. Massaro, M.; Dumay, J.; Guthrie, J. On the Shoulders of Giants: Undertaking a Structured Literature Review in Accounting. *Account. Audit. Account. J.* 2016, 29, 767–801. [CrossRef]
- 18. Massaro, M.; Dumay, J.; Garlatti, A.; Dal Mas, F. Practitioners' Views on Intellectual Capital and Sustainability: From a Performance-Based to a Worth-Based Perspective. *J. Intellect. Cap.* **2018**, *19*, 367–386. [CrossRef]
- Secinaro, S.; Brescia, V.; Calandra, D.; Biancone, P. Employing Bibliometric Analysis to Identify Suitable Business Models for Electric Cars. J. Clean. Prod. 2020, 264, 121503. [CrossRef]

- 20. Secinaro, S.; Calandra, D. Halal Food: Structured Literature Review and Research Agenda. *Br. Food J.* 2020. ahead-of-print. [CrossRef]
- 21. Rangan, V.K.; Appleby, S.; Moon, L. *The Promise of Impact Investing*; Background Note; Harvard Business School: Boston, MA, USA, 2011.
- 22. Eurosif Impact Investing in Europe: Extract from European SRI Study 2014. European Sustainable Investment Forum, Brussel. Available online: https://informaconnect.com/sustainability-and-impact-investor-forum/?vip\_code=FKN274 2GOOGLE&gclid=Cj0KCQjw7qn1BRDqARIsAKMbHDaP7nZqySPzBzxA\_RJDzCAt4JJ8vTBCODDpIH\_IL20WG\_0AN6sv1 CoaAiZEEALw\_wcB (accessed on 30 April 2020).
- 23. Baumli, K.; Jamasb, T. Assessing Private Investment in African Renewable Energy Infrastructure: A Multi-Criteria Decision Analysis Approach. *Sustainability* **2020**, *12*, 9425. [CrossRef]
- 24. Di Domenico, M.; Haugh, H.; Tracey, P. Social Bricolage: Theorizing Social Value Creation in Social Enterprises. *Entrep. Theory Pract.* **2010**, *34*, 681–703. [CrossRef]
- 25. Gundry, L.K.; Kickul, J.R.; Griffiths, M.D.; Bacq, S.C. Creating Social Change out of Nothing: The Role of Entrepreneurial Bricolage in Social Entrepreneurs' Catalytic Innovations. *Adv. Entrep. Firm Emerg. Growth* **2011**, *13*, 1–24.
- 26. Huybrechts, B.; Nicholls, A. Social entrepreneurship: Definitions, drivers and challenges. In *Social Entrepreneurship and Social Business*; Springer: Berlin/Heidelberg, Germany, 2012; pp. 31–48.
- 27. Urmanaviciene, A.; Arachchi, U.S. The Effective Methods and Practices for Accelerating Social Entrepreneurship through Corporate Social Responsibility. *Eur. J. Soc. Impact Circ. Econ.* **2020**, *1*, 27–47. [CrossRef]
- 28. Martin, M.; Impact Economy. Status of the Social Impact Investing Market: A Primer. 2013. Available online: https://www.impacteconomy.com/papers/IE\_PRIMER\_JUNE2013\_EN (accessed on 10 November 2020).
- 29. Barigozzi, F.; Tedeschi, P. Credit Markets with Ethical Banks and Motivated Borrowers. *Rev. Financ.* 2015, 19, 1281–1313. [CrossRef]
- 30. Nicholls, A. The Institutionalization of Social Investment: The Interplay of Investment Logics and Investor Rationalities. *J. Soc. Entrep.* **2010**, *1*, 70–100. [CrossRef]
- 31. Secinaro, S.; Corvo, L.; Brescia, V.; Iannaci, D. Hybrid Organizations: A Systematic Review of the Current Literature. *Int. Bus. Res.* **2019**, *12*, p1. [CrossRef]
- 32. Iannaci, D. Reporting Tools for Social Enterprises: Between Impact Measurement and Stakeholder Needs. *Eur. J. Soc. Impact Circ. Econ.* **2020**, *1*, 1–18. [CrossRef]
- 33. Bugg-Levine, A.; Emerson, J. Impact Investing: Transforming How We Make Money While Making a Difference. *Innov. Technol. Gov. Glob.* **2011**, *6*, 9–18. [CrossRef]
- 34. Hebb, T. Impact Investing and Responsible Investing: What Does It Mean? Taylor & Francis: Oxfordshire, UK, 2013.
- 35. Glänzel, G.; Scheuerle, T. Social Impact Investing in Germany: Current Impediments from Investors' and Social Entrepreneurs' Perspectives. *Volunt. Int. J. Volunt. Nonprofit Organ.* **2016**, 27, 1638–1668. [CrossRef]
- 36. Dionisio, M. The Evolution of Social Entrepreneurship Research: A Bibliometric Analysis. *Soc. Enterp. J.* **2019**, *15*, 22–45. [CrossRef]
- 37. Biancone, P.P.; Saiti, B.; Petricean, D.; Chmet, F. The Bibliometric Analysis of Islamic Banking and Finance. *J. Islamic Account. Bus. Res.* **2020**. ahead-of-print. [CrossRef]
- 38. Fabregat-Aibar, L.; Barberà-Mariné, M.G.; Terceño, A.; Pié, L. A Bibliometric and Visualization Analysis of Socially Responsible Funds. *Sustainability* **2019**, *11*, 2526. [CrossRef]
- 39. Okoli, C.; Schabram, K. A Guide to Conducting a Systematic Literature Review of Information Systems Research. *SSRN Electron. J.* **2010**. [CrossRef]
- 40. Neely, A. The Evolution of Performance Measurement Research. Int. J. Oper. Prod. Manag. 2005, 25, 1264–1277. [CrossRef]
- 41. Riva, P.; Comoli, M.; Bavagnoli, F.; Gelmini, L. Performance Measurement: From Internal Management to External Disclosure. *Corp. Ownersh. Control* **2015**, *13*, 907–926. [CrossRef]
- 42. Taticchi, P.; Tonelli, F.; Cagnazzo, L. Performance Measurement and Management: A Literature Review and a Research Agenda. *Meas. Bus. Excell.* **2010**, *14*, 4–8. [CrossRef]
- 43. Chen, G.; Xiao, L. Selecting Publication Keywords for Domain Analysis in Bibliometrics: A Comparison of Three Methods. *J. Informetr.* **2016**, *10*, 212–223. [CrossRef]
- 44. Dal Mas, F.; Garcia-Perez, A.; José Sousa, M.; Lopes da Costa, R.; Cobianchi, L. Knowledge Translation in the Healthcare Sector, A Structured Literature Review. *Electron. J. Knowl. Manag.* **2020**, *18*.
- 45. Massaro, M.; Secinaro, S.; Mas, F.D.; Brescia, V.; Calandra, D. Industry 4.0 and Circular Economy: An Exploratory Analysis of Academic and Practitioners' Perspectives. *Bus. Strategy Environ.* **2020**. [CrossRef]
- 46. Li, J.; Wu, D.; Li, J.; Li, M. A Comparison of 17 Article-Level Bibliometric Indicators of Institutional Research Productivity: Evidence from the Information Management Literature of China. *Inf. Process. Manag.* **2017**, *53*, 1156–1170. [CrossRef]
- 47. Mingers, J.; Willmott, H. Taylorizing Business School Research: On the 'One Best Way' Performative Effects of Journal Ranking Lists. *Hum. Relat.* **2013**, *66*, 1051–1073. [CrossRef]
- 48. Tüselmann, H.; Sinkovics, R.R.; Pishchulov, G. Revisiting the Standing of International Business Journals in the Competitive Landscape. *J. World Bus.* **2016**, *51*, 487–498. [CrossRef]

- 49. Xu, X.; Chen, X.; Jia, F.; Brown, S.; Gong, Y.; Xu, Y. Supply Chain Finance: A Systematic Literature Review and Bibliometric Analysis. *Int. J. Prod. Econ.* **2018**, 204, 160–173. [CrossRef]
- 50. Easterby-Smith, M.; Thorpe, R.; Jackson, P.; Lowe, A. Management Research, 4th ed.; SAGE: London, UK, 2012.
- 51. Levy, Y.; Ellis, T.J. A Systems Approach to Conduct an Effective Literature Review in Support of Information Systems Research. *Inf. Sci. Int. J. Emerg. Transdiscipl.* **2006**, *9*, 181–212. [CrossRef]
- 52. Aria, M.; Cuccurullo, C. Bibliometrix: An R-Tool for Comprehensive Science Mapping Analysis. J. Informetr. 2017, 11, 959–975. [CrossRef]
- 53. Fry, M.J. Money and Capital or Financial Deepening in Economic Developments. In *Money and Monetary Policy in Less Developed Countries*; Elsevier: Amsterdam, The Netherlands, 1980; pp. 107–113.
- 54. Galbis, V. Financial Intermediation and Economic Growth in Less-Developed Countries: A Theoretical Approach. *J. Dev. Stud.* **1977**, *13*, 58–72. [CrossRef]
- 55. Cobo, M.J.; López-Herrera, A.G.; Herrera-Viedma, E.; Herrera, F. An Approach for Detecting, Quantifying, and Visualizing the Evolution of a Research Field: A Practical Application to the Fuzzy Sets Theory Field. *J. Informetr.* **2011**, *5*, 146–166. [CrossRef]
- 56. Aparicio, G.; Iturralde, T.; Maseda, A. Conceptual Structure and Perspectives on Entrepreneurship Education Research: A Bibliometric Review. *Eur. Res. Manag. Bus. Econ.* **2019**, *25*, 105–113. [CrossRef]
- 57. Garfield, E.; Sher, I.H. Key Words plus [TM]-Algorithmic Derivative Indexing. J. Am. Soc. Inf. Sci. 1993, 44, 298–299. [CrossRef]
- 58. Zhang, J.; Yu, Q.; Zheng, F.; Long, C.; Lu, Z.; Duan, Z. Comparing Keywords plus of WOS and Author Keywords: A Case Study of Patient Adherence Research. J. Assoc. Inf. Sci. Technol. 2016, 67, 967–972. [CrossRef]
- 59. Latané, B. The Psychology of Social Impact. Am. Psychol. 1981, 36, 343. [CrossRef]
- 60. Noyons, E. Bibliometric Mapping of Science in a Policy Context. *Scientometrics* 2004, *50*, 83–98. [CrossRef]
- 61. Forina, M.; Armanino, C.; Raggio, V. Clustering with Dendrograms on Interpretation Variables. *Anal. Chim. Acta* 2002, 454, 13–19. [CrossRef]
- 62. Small, H. Co-Citation in the Scientific Literature: A New Measure of the Relationship between Two Documents. J. Am. Soc. Inf. Sci. 1973, 24, 265–269. [CrossRef]
- 63. Vogel, B.; Reichard, R.J.; Batistič, S.; Černe, M. A Bibliometric Review of the Leadership Development Field: How We Got Here, Where We Are, and Where We Are Headed. *Leadersh. Q.* **2020**, 101381. [CrossRef]
- 64. Garfield, E. Historiographic Mapping of Knowledge Domains Literature. J. Inf. Sci. 2004, 30, 119–145. [CrossRef]
- 65. Cetindamar, D.; Ozkazanc-Pan, B. Assessing Mission Drift at Venture Capital Impact Investors. *Bus. Ethics A Eur. Rev.* 2017, 26, 257–270. [CrossRef]
- 66. Ormiston, J.; Charlton, K.; Donald, M.S.; Seymour, R.G. Overcoming the Challenges of Impact Investing: Insights from Leading Investors. *J. Soc. Entrep.* **2015**, *6*, 352–378. [CrossRef]
- 67. Tullberg, J. Triple Bottom Line—A Vaulting Ambition? Bus. Ethics A Eur. Rev. 2012, 21, 310–324. [CrossRef]
- 68. Bacq, S.; Eddleston, K.A. A Resource-Based View of Social Entrepreneurship: How Stewardship Culture Benefits Scale of Social Impact. *J. Bus. Ethics* **2018**, *152*, 589–611. [CrossRef]
- 69. Herrera, M.E.B. Innovation for Impact: Business Innovation for Inclusive Growth. J. Bus. Res. 2016, 69, 1725–1730. [CrossRef]
- 70. Maas, K.; Grieco, C. Distinguishing Game Changers from Boastful Charlatans: Which Social Enterprises Measure Their Impact? J. Soc. Entrep. 2017, 8, 110–128. [CrossRef]
- 71. Smith, B.R.; Kistruck, G.M.; Cannatelli, B. The Impact of Moral Intensity and Desire for Control on Scaling Decisions in Social Entrepreneurship. *J. Bus. Ethics* **2016**, *133*, 677–689. [CrossRef]
- 72. Agyekum, E.O.; Fortuin, K.K.; van der Harst, E. Environmental and Social Life Cycle Assessment of Bamboo Bicycle Frames Made in Ghana. *J. Clean. Prod.* 2017, 143, 1069–1080. [CrossRef]
- 73. Neugebauer, S.; Emara, Y.; Hellerström, C.; Finkbeiner, M. Calculation of Fair Wage Potentials along Products' Life Cycle-Introduction of a New Midpoint Impact Category for Social Life Cycle Assessment. *J. Clean. Prod.* **2017**, *143*, 1221–1232. [CrossRef]
- 74. Prasara-A, J.; Gheewala, S.H. Applying Social Life Cycle Assessment in the Thai Sugar Industry: Challenges from the Field. *J. Clean. Prod.* **2018**, *172*, 335–346. [CrossRef]
- 75. Subramanian, K.; Yung, W.K. Modeling Social Life Cycle Assessment Framework for an Electronic Screen Product—A Case Study of an Integrated Desktop Computer. *J. Clean. Prod.* **2018**, *197*, 417–434. [CrossRef]
- 76. Jeucken, M. Sustainable Finance and Banking: The Financial Sector and the Future of the Planet; Earthscan: London, UK, 2010.
- 77. Zuo, Z.; Zhao, K. The More Multidisciplinary the Better—The Prevalence and Interdisciplinarity of Research Collaborations in Multidisciplinary Institutions. *J. Informetr.* **2018**, *12*, 736–756. [CrossRef]
- Klemeš, J.J.; Varbanov, P.S.; Huisingh, D. Recent Cleaner Production Advances in Process Monitoring and Optimisation. J. Clean. Prod. 2012, 34, 1–8. [CrossRef]
- 79. McCrea, R.; Walton, A.; Leonard, R. Rural Communities and Unconventional Gas Development: What's Important for Maintaining Subjective Community Wellbeing and Resilience over Time? *J. Rural Stud.* **2019**, *68*, 87–99. [CrossRef]
- 80. Luo, J.; Kaul, A. Private Action in Public Interest: The Comparative Governance of Social Issues. *Strateg. Manag. J.* **2019**, 40, 476–502. [CrossRef]
- 81. Aledo-Tur, A.; Domínguez-Gómez, J.A. Social Impact Assessment (SIA) from a Multidimensional Paradigmatic Perspective: Challenges and Opportunities. *J. Environ. Manag.* 2017, 195, 56–61. [CrossRef]

- 82. Schinckus, C. Financial Innovation as a Potential Force for a Positive Social Change: The Challenging Future of Social Impact Bonds. *Res. Int. Bus. Financ.* 2017, *39*, 727–736. [CrossRef]
- 83. Arena, M.; Bengo, I.; Calderini, M.; Chiodo, V. Social Impact Bonds: Blockbuster or Flash in a Pan? *Int. J. Public Adm.* 2016, 39, 927–939. [CrossRef]
- 84. Revelli, C. Re-Embedding Financial Stakes within Ethical and Social Values in Socially Responsible Investing (SRI). *Res. Int. Bus. Financ.* **2016**, *38*, 1–5. [CrossRef]
- 85. Shen, C.-H.; Chang, Y. Ambition Versus Conscience, Does Corporate Social Responsibility Pay off? The Application of Matching Methods. *J. Bus. Ethics* **2009**, *88*, 133–153. [CrossRef]
- Campra, M.; Esposito, P.; Lombardi, R. The Engagement of Stakeholders in Nonfinancial Reporting: New Information-Pressure, Stimuli, Inertia, under Short-Termism in the Banking Industry. *Corp. Soc. Responsib. Environ. Manag.* 2020, 27, 1436–1444. [CrossRef]
- 87. Davila, A.; Rodriguez-Lluesma, C.; Elvira, M.M. Engaging Stakeholders in Emerging Economies: The Case of Multilatinas. *J. Bus. Ethics* **2018**, 152, 949–964. [CrossRef]
- 88. Haggerty, J.; McBride, K. Does Local Monitoring Empower Fracking Host Communities? A Case Study from the Gas Fields of Wyoming. *J. Rural Stud.* 2016, 43, 235–247. [CrossRef]
- 89. Muñoz de Prat, J.; Escriva-Beltran, M.; Gómez-Calvet, R. Joint Ventures and Sustainable Development. A Bibliometric Analysis. *Sustainability* **2020**, *12*, 10176. [CrossRef]

MDPI St. Alban-Anlage 66 4052 Basel Switzerland Tel. +41 61 683 77 34 Fax +41 61 302 89 18 www.mdpi.com

Sustainability Editorial Office E-mail: sustainability@mdpi.com www.mdpi.com/journal/sustainability



MDPI St. Alban-Anlage 66 4052 Basel Switzerland

Tel: +41 61 683 77 34 Fax: +41 61 302 89 18

www.mdpi.com



ISBN 978-3-0365-3525-8